

CAHIER 9613

STOCHASTIC VOLATILITY

Eric Ghysels¹, Andrew Harvey² and Eric Renault³

- Département de sciences économiques and Centre de recherche et développement en économique (C.R.D.E.), Université de Montréal, and CIRANO
- London School of Economics
- 3 GREMAQ and IDEI, Université des Sciences Sociales, Toulouse and Institut Universitaire de France

July 1996

The authors benefitted from helpful comments from Torben Andersen, David Bates, Frank Diebold, René Garcia, Eric Jacquier and Nell Shephard on preliminary drafts of the paper. The first author would like to acknowledge the financial support of FCAR (Québec), SSHRC (Canada) as well as the hospitality and support of CORE (Louvain-ta-Neuve, Belgium). The second author wishes to thank the ESRC for financial support. The third author would like to thank the institut Universitaire de France, the Fédération Française des Sociétés d'Assurance as well as CIRANO and C.R.D.E. for financial support.

Ce cahier a également été publié au Centre de recherche et développement en économique (C.R.D.E.) (publication no 1796). Dépôt légal - 1996 Bibliothèque nationale du Québec Bibliothèque nationale du Canada ISSN 0709-9231

RÉSUMÉ

Cet article, préparé pour le Handbook of Statistics (vol. 14 : "Statistical Methods in Finance"), passe en revue les modèles de volatilité stochastique. On traite les sujets suivants : volatilité des actifs financiers (volatilité instantanée des rendements d'actifs, volatilités implicites dans les prix d'option et régularités empiriques), modélisation statistique en temps discret et continu et, enfin, inférence statistique (méthodes de moments, pseudo-maximum de vraisemblance, méthodes bayésiennes et autres fondées sur la vraisemblance, inférence indirecte).

Mots clés : rendements d'actifs, hétéroscédasticité conditionnelle, prix d'option, modèles état-espace, processus de diffusion

ABSTRACT

This paper, prepared for the *Handbook of Statistics* (Vol. 14: "Statistical Methods in Finance"), surveys the subject of stochastic volatility. The following subjects are covered: volatility in financial markets (instantaneous volatility of asset returns, implied volatilities in option prices and related stylized facts), statistical modelling in discrete and continuous time and, finally, statistical inference (methods of moments, quasi-maximum likelihood, likelihood-based and bayesian methods and indirect inference).

Key words: asset returns, conditional heteroskedasticity, option prices, state-space models, diffusion process

1. Introduction

The class of stochastic volatility (SV) models has its roots both in mathematical finance and financial econometrics. In fact, several variations of SV models originated from research looking at very different issues. Clark (1973), for instance, suggested to model asset returns as a function of a random process of information arrival. This so-called time deformation approach yielded a time-varying volatility model of asset returns. Later Tauchen and Pitts (1983) refined this work proposing a mixture of distributions model of asset returns with temporal dependence in information arrivals. Hull and White (1987) were not directly concerned with linking asset returns to information arrival but rather were interesting in pricing European options assuming continuous time SV models for the underlying asset. They suggested a diffusion for asset prices with volatility following a positive diffusion process. Yet another approach emerged from the work of Taylor (1986) who formulated a discrete time SV model as an alternative to Autoregressive Conditional Heteroskedasticity (ARCH) models. Until recently estimating Taylor's model, or any other SV model, remained almost infeasible. Recent advances in econometric theory have made estimation of SV models such as ARCH.

Contributions to the literature on SV models can be found both in mathematical finance and econometrics. Hence, we face quite a diverse set of topics. We say very little about ARCH models because several excellent surveys on the subject have appeared recently, including those by Bera and Higgins (1995), Bollerslev, Chou and Kroner (1992), Bollerslev, Engle and Nelson (1994) and Diebold and Lopez (1995). Furthermore, since this chapter is written for the Handbook of Statistics, we keep the coverage of the mathematical finance literature to a minimum. Nevertheless, the subject of option pricing figures prominently out of necessity. Indeed, section 2, which deals with definitions of volatility has extensive coverage of Black-Scholes implied volatilities. It also summarizes empirical stylized facts and concludes with statistical modelling of volatility. The reader with a greater interest in statistical concepts may want to sldp the first three subsections of section 2 which are more finance oriented and start with section 2.4. Section 3 discusses discrete time models, while section 4 reviews continuous time models. Statistical inference of SV models is the subject of section 5. Section 6 concludes.

2. Volatility in Financial Markets

Volatility plays a central role in the pricing of derivative securities. The Black-Scholes model for the pricing of an European option is by far the most widely used formula even when the underlying assumptions are known to be violated. Section 2.1 will therefore take the Black-Scholes model as a reference point from which to discuss several notions of volatility. A discussion of stylized facts regarding volatility and option prices will appear next in section 2.2. Both sections set the scene for a formal framework defining stochastic volatility which is treated in section 2.3. Finally, section 2.4 introduces the *statistical* models of stochastic volatility.

2.1. The Black-Scholes Model and Implied Volatilities

More than half a century after the seminal work of Louis Bachelier (1900), continuous time stochastic processes have become a standard tool to describe the behavior of asset prices. The work of Black and Scholes (1973) and Merton (1990) has been extremely influential in that regard. In section 2.1.1 we review some of the assumptions that are made when modelling asset prices by diffusions, in particular to present the concept of instantaneous volatility. In section 2.1.2 we turn to option pricing models and the various concepts of implied volatility.

2.1.1. An Instantaneous Volatility Concept

We consider a financial asset, say a stock, with today's (time t) market price denoted by S_t .² Let the information available at time t be described by I_t and consider the conditional distribution of the return S_{t+h}/S_t of holding the asset over the period [t,t+h] given I_t .³ A maintained assumption throughout this chapter will be that asset returns have finite conditional expectation given I_t or:

$$E_t(S_{t+h}/S_t) = S_t^{-1} E_t S_{t+h} < +\infty$$
 (2.1.1)

and likewise finite conditional variance given I_L , namely

$$V_t(S_{t+h}/S_t) = S_t^{-2} V_t S_{t+h} < +\infty$$
 (2.1.2)

The continuously compounded expected rate of return will be characterized by $h^{-1} \log E_t(S_{t+h}/S_t)$. Then a first assumption can be stated as follows:

Assumption 2.1.1.4: The continuously compounded expected rate of return converges almost surely towards a finite value $\mu_S(I_4)$ when h > 0 goes to zero.

From this assumption one has $E_t S_{t+h} - S_t \sim h\mu_S(I_t) S_t$ or in terms of its differential representation:

$$\frac{d}{d\tau}E_{t}\left(S_{\tau}\right)\Big|_{\tau=t}=\mu_{S}\left(I_{t}\right)S_{t}$$
 almost surely (2.1.3)

where the derivatives are taken from the right. Equation (2.1.3) is sometimes loosely defined as: $E_t(dS_t) = \mu_S(I_t) S_t dt$. The next assumption pertains to the conditional variance and can be stated as:

Assumption 2.1.1.B: The conditional variance of the return $h^{-1}V_t(S_{t+h}/S_t)$ converges almost surely towards a finite value $\sigma_S^2(I_t)$ when h > 0 goes to zero.

Again, in terms of its differential representation this amounts to :

$$\frac{d}{d\tau} Var_t(S_\tau)\Big|_{\tau=t} = \sigma_S^2(I_t) S_t^2 \text{ almost surely}$$
(2.1.4)

and one loosely associates with the expression $V_t(dS_t) = \sigma_S^2(I_t) S_t^2 dt$.

²Here and in the remainder of the paper we will focus on options written on stocks or exchange rates. The large literature on the term structure of interest rates and related derivative securities will not be covered.

³Section 2.3 will provide a more rigorous discussion of information sets. It should also be noted that we will indifferently be using conditional distributions of asset prices S_{t+h} and of returns S_{t+h}/S_t since S_t belongs to I_t .

Both assumptions 2.1.1.A and B lead to a representation of the asset price dynamics by an equation of the following form:

$$dS_t = \mu_S(I_t)S_t dt + \sigma_S(I_t)S_t dW_t$$
 (2.1.5)

where W_t is a standard Brownian Motion. Hence, every time a diffusion equation is written for an asset price process we have automatically defined the so-called instantaneous volatility process $\sigma_S(I_t)$ which from the above representation can also be written as:

$$\sigma_{S}(I_{t}) = \left[\lim_{h \downarrow o} h^{-1}V_{t}(S_{t+h}/S_{t}) \right]^{\frac{1}{2}}$$
(2.1.6)

Before turning to the next section we would like to provide a brief discussion of some of the foundations for the Assumptions 2.1.1.A and B. It was noted that Bachelier (1900) proposed Brownian Motion process as a model of stock price movements. In modern terminology this amounts to the random walk theory of asset pricing which claims that asset returns ought not to be predictable because of the informational efficiency of financial markets. Hence, it assumes that returns on consecutive regularly sampled periods $\{t+k, t+k+1\}, k=0,2,...,h-1$ are independently (identically) distributed. With such a benchmark in mind, it is natural to view the expectation and the variance of the continuously compounded rate of return $\log (S_{t+h}/S_t)$ as proportional to the maturity h of the investment.

Obviously we no longer use Brownian Motions as a process for asset prices but it is nevertheless worth noting that Assumptions 2.1.1.A and B also imply that the expected rate of return and the associated squared risk (in terms of variance of the rate of return) of an investment over an infinitely-short interval [t, t+h] is proportional to h. Sims (1984) provided some rationale for both assumptions through the concept of "local unpredictability".

To conclude, let us briefly discuss a particular special case of (2.1.5) predominantly used in theoretical developments and also highlight an implicit restriction we made. When $\mu_S(I_t) = \mu_S$ and $\sigma_S(I_t) = \sigma_S$ are constants for all t the asset price is a Geometric Brownian Motion. This process was used by Black and Scholes (1973) to derive their well-known pricing formula for European options. Obviously, since $\sigma_S(I_t)$ is a constant we no longer have an instantaneous volatility process but rather a single parameter σ_S - a situation which undoubtedly greatly simplifies many things including the pricing of options. A second point which needs to be stressed is that Assumptions 2.1.1.A and B allow for the possibility of discrete jumps in the asset price process. Such jumps are typically represented by a Poisson process and have been prominent in the option pricing literature since the work of Merton (1976). Yet, while the assumptions allow in principle for jumps, they do not appear in (2.1.5). Indeed, throughout this chapter we will maintain the assumption of sample path continuity and exclude the possibility of jumps as we focus exclusively on SV models.

2.1.2. Option Prices and Implied Volatilities

It was noted in the introduction that SV models originated in part from the literature on the pricing of options. We have witnessed over the past two decades a spectacular growth in options and other derivative security markets. Such markets are sometimes characterized as places where "volatilities are traded". In this section we will provide the rationale for such statements and study the relationship between so-called options implied volatilities and the concepts of instantaneous and averaged volatilities of the underlying asset return process.

The Black-Scholes option pricing model is based on a Log-Normal or Geometric Brownian Motion model for the underlying asset price:

$$dS_t = \mu_S S_t dt + \sigma_S S_t dW_t \tag{2.1.7}$$

where μ_S and σ_S are fixed parameters. A European call option with strike price K and maturity t+h has a payoff:

 $[S_{t+h} - K]^{+} = \begin{cases} S_{t+h} - K & \text{if } S_{t+h} \ge K \\ 0 & \text{otherwise} \end{cases}$ (2.1.8)

Since the seminal Black and Scholes (1973) paper, there is now a well established literature proposing various ways to derive the pricing formula of such a contract. Obviously, it is beyond the scope of this paper to cover this literature in detail.⁴ Instead, the bare minimum will be presented here allowing us to discuss the concepts of interest regarding volatility.

With continuous costless trading assumed to be feasible, it is possible to form in the Black-Scholes economy a portfolio using one call and a short-sale strategy for the underlying stock to eliminate all risk. This is why the option price can be characterized without ambiguity, using only arbitrage arguments, by equating the market rate of return of the riskless portfolio containing the call option with the risk-free rate. Moreover, such arbitrage-based option pricing does not depend on individual preferences.

This is the reason why the easiest way to derive the Black-Scholes option pricing formula is via a "risk-neutral world", where asset price processes are specified through a modified probability measure, referred to as the risk neutral probability measure denoted Q (as discussed more explicitly in section 4.2). This fictitious world where probabilities in general do not coincide with the Data Generating Process (DGP), is only used to derive the option price which remains valid in the objective probability setup. In the risk neutral world we have:

$$dS_t/S_t = r_t dt + \sigma_S dW_t \tag{2.1.9}$$

$$C_t = C(S_t, K, h, t) = B(t, t + h)E_t^Q(S_{t+h} - K)^+$$
(2.1.10)

where E_t^Q is the expectation under Q, B(t, t+h) is the price at time t of a pure discount bond with payoff one unit at time t+h and

$$r_{t} = -\lim_{h \to 0} \frac{1}{h} Log \ B(t, t + h)$$
 (2.1.11)

is the riskless instantaneous interest rate. We have implicitly assumed that in this market interest rates are nonstochastic (W_t is the only source of risk) so that:

$$B(t,t+h) = exp\left[-\int_{t}^{t+h} r_{\tau} d_{\tau}\right].$$
 (2.1.12)

⁴See however Jarrow and Rudd (1983), Cox and Rubiasteia (1985), Duffie (1989), Duffie (1992), Hull (1993) or Hull (1995) among others for more elaborate coverage of options and other derivative securities.

⁸This is sometimes referred to as preference free option pricing. This terminology may somewhat be misleading since individual preferences are implicitly taken into account in the market price of the stock and of the riskless bond. However, the option price only depends on individual preferences through the stock and bond market prices.

⁴For notational convenience we denote by the same symbol W_i a Brownian Motion under P (in 2.1.7) and under Q (in 2.1.9). Indeed, Girsanov's theorem establishes the link between these two processes (see e.g. Duffie (1992) and section 4.2.1).

By definition, there are no risk premia in a risk neutral context. Therefore r_i coincides with the instantaneous expected rate of return of the stock and hence the call option price C_t is the discounted value of its terminal payoff $(S_{t+h} - K)^+$ as stated in (2.1.10).

The log-normality of S_{t+h} given S_t allows one to compute the expectation in (2.1.10) yielding the call price formula at time t:

$$C_t = S_t \phi(d_t) - KB(t, t+h)\phi(d_t - \sigma_S \sqrt{h})$$
 (2.1.13)

where ϕ is the cumulative standard normal distribution function while d_t will be defined shortly. Formula (2.1.13) is the so-called Black-Scholes option pricing formula. Thus, the option price C_t depends on the stock price S_t , the strike price K and the discount factor B(t, t + h). Let us now define :

$$x_t = Log S_t/KB(t, t+h)$$
 (2.1.14)

Then we have:

$$C_t/S_t = \phi(d_t) - e^{-x_t}\phi(d_t - \sigma_S\sqrt{h}) \qquad (2.1.15)$$

with $d_t = (x_t/\sigma_s\sqrt{h}) + \sigma_s\sqrt{h}/2$. It is easy to see the critical role played by the quantity x_t , called the moneyness of the option.

- If $x_t = 0$, the current stock price S_t coincides with the present value of the strike price K. In other words, the contract may appear to be fair to somebody who would not take into account the stochastic changes of the stock price between t and t + h. We shall say that we have in this case an at the money option.
- If $x_t > 0$ (respectively $x_t < 0$) we shall say that the option is in the money (respectively out the money).7

It was noted before that the Black-Scholes formula is widely used among practitioners, even when its assumptions are known to be violated. In particular the assumption of a constant volatility σ_S is unrealistic (see section 2.2 for empirical evidence). This motivated Hull and White (1987) to introduce an option pricing model with stochastic volatility assuming that the volatility itself is a state variable independent of $W_{\epsilon}:^{\mathbf{B}}$

$$\begin{cases} dS_t/S_t = r_t dt + \sigma_{S_t} dW_t \\ (\sigma_{S_t})_{t \in [0,T]}, (W_t)_{t \in [0,T]} \text{ independent Markovian} \end{cases}$$
(2.1.16)

It should be noted that (2..1.16) is still written in a risk neutral context since r_t coincides with the instantaneous expected return of the stock. On the other hand the exogenous volatility risk is not directly traded, which prevents us from defining unambiguously a risk neutral probability measure, as discussed in more detail in section 4.2. Nevertheless, the option pricing formula (2.1.10) remains valid provided the expectation is computed with respect to the joint probability distribution of the Markovian process (S, σ_S) , given (S_t, σ_{St}) . We can then rewrite (2.1.10) as follows:

We use here a slightly modified terminology with respect to the usual one. Indeed, it is more common to call at the money in the money out of the money options, when $S_i = K/S_i > K/S_i < K$ respectively. From an economic point of view, it is more appealing to compare S_t with the present value of the strike price K.

Other stochastic volstility models similar to Hull and White (1987) appear in Johnson and Shanno (1987), Scott (1987), Wiggins (1987), Chesney and Scott (1969), Stein and Stein (1991) and Heston (1993) among others.

We implicitly assume here that the available information I_t contains the past values $(S_{\tau}, \sigma_{\tau})_{\tau \leq t}$. This assumption will

be discussed in section 4.2.

$$C_{t} = B(t, t+h) E_{t}(S_{t+h} - K)^{+} = B(t, t+h) E_{t} \left\{ E\left[(S_{t+h} - K)^{+} | (\sigma_{ST})_{t \leq t \leq t+h} \right] \right\}$$
(2.1.17)

where the expectation inside the brackets is taken with respect to the conditional probability distribution of S_{t+h} given I_t and a volatility path $\sigma_{S\tau}$, $t \leq \tau \leq t+h$. However, since the volatility process $\sigma_{S\tau}$ is independent from W_t , we obtain using (2.1.15) that:

$$B(t, t+h) E_t \left[(S_{t+h} - K)^+ \left| (\sigma_{S_7})_{t \le \tau \le t+h} \right| = S_t E_t \left[\phi(d_{1t}) - e^{-\alpha_t} \phi(d_{2t}) \right]$$
 (2.1.18)

where d_{1t} and d_{2t} are defined as follows:

$$\begin{cases} d_{1t} = \left(x_t/\gamma\left(t, t+h\right)\sqrt{h}\right) + \gamma\left(t, t+h\right)\sqrt{h}/2 \\ d_{2t} = d_1 - \gamma\left(t, t+h\right)\sqrt{h} \end{cases}$$

where $\gamma(t, t+h) > 0$ and :

$$\gamma^{2}(t, t + h) = \frac{1}{h} \int_{t}^{t+h} \sigma_{S\tau}^{2} d\tau.$$
 (2.1.19)

This yields the so-called Hull and White option pricing formula:

$$C_{t} = S_{t}E_{t} \left[\phi \left(d_{1t} \right) - e^{-x_{t}} \phi \left(d_{2t} \right) \right], \qquad (2.1.20)$$

where the expectation is taken with respect to the conditional probability distribution (for the risk neutral probability measure) of $\gamma(t,t+h)$ given σ_{St} .

In the remainder of this section we will assume that observed option prices obey Hull and White's formula (2.1.20). Then option prices would yield two types of implied volatility concepts: (1) an instantaneous implied volatility and (2) an averaged implied volatility. To make this more precise, let us assume that the risk neutral probability distribution belongs to a parametric family, P_{θ} , $\theta \in \Theta$. Then, the Hull and White option pricing formula yields an expression for the option price as a function:

$$C_t = S_t F\left[\sigma_{St}, x_t, \theta_o\right] \tag{2.1.21}$$

where θ_o is the true unknown value of the parameters. Formula (2.1.21) reveals why it is often claimed that "option markets can be thought of as markets trading volatility" (see e.g. Stein (1989)). As a matter of fact, if for any given (x_t, θ) , $F(\cdot, x_t, \theta)$ is one-to-one, then equation (2.1.21) can be inverted to yield an implied instantaneous volatility:¹¹

$$\sigma_{\star}^{imp}(\theta) = G[S_t, C_t, x_t, \theta] \tag{2.1.22}$$

Bajeux and Rochet (1992), by showing that this one-to-one relationship between option prices and instantaneous volatility holds, in fact formalize the use of option markets as an appropriate instrument to hedge volatility risk. Obviously implied instantaneous volatilities (2.1.22) could only be useful in practice for pricing or hedging derivative instruments when we know the true unknown value θ_o or , at least, are able to compute a sufficiently accurate estimate of it.

¹⁰The conditioning is with respect to σ_t since it summarises the relevant information taken from I_t (the process σ is essumed to be Markovian and independent from W).

¹¹The fact that $F(\cdot, x_t, \theta)$ is one-to-one is shown to be the case for any diffusion model on σ_{St} under certain regularity conditions, see Bajeux and Rochet (1992).

However, the difficulties involved in estimating SV models has for long prevented their wide spread use in empirical applications. This is the reason why practitioners often prefer another concept of implied volatility, namely the so-called Black-Scholes implied volatility introduced by Latane and Rendleman (1976). It is a process $\omega^{imp}(t, t + h)$ defined by:

$$\begin{cases} C_{1} = S_{1} \left[\phi \left(d_{11} \right) - e^{-x_{1}} \phi \left(d_{21} \right) \right] \\ d_{1t} = \left(x_{t} / \omega^{imp} \left(t, t + h \right) \sqrt{h} \right) + \omega^{imp} \left(t, t + h \right) \sqrt{h} / 2 \\ d_{2t} = d_{1t} - \omega^{imp} \left(t, t + h \right) \sqrt{h} \end{cases}$$
(2.1.23)

where C_i is the observed option price. 12

The Hull and White option pricing model can indeed be seen as a theoretical foundation for this practice; the comparison between (2.1.23) and (2.1.20) allows us to interpret the Black Scholes implied volatility $\omega^{imp}(t,t+h)$ as an implied averaged volatility since $\omega^{imp}(t,t+h)$ is something like a conditional expectation of $\gamma(t,t+h)$ (assuming observed option prices coincide with the Hull and White pricing formula). To be more precise, let us consider the simplest case of at the money options (the general case will be studied in section 4.2). Since $x_t = 0$ it follows that $d_{2t} = -d_{1t}$ and therefore: $\phi(d_{1t}) - e^{-x_t}\phi(d_{2t}) = 2\phi(d_{1t}) - 1$. Hence, $\omega_a^{imp}(t,t+h)$ (the index o is added to underline that we consider at the money options) is defined by:

$$\phi\left(\frac{\omega_{\phi}^{imp}\left(t,t+h\right)\sqrt{h}}{2}\right) = E_{i}\phi\left(\frac{\gamma\left(t,t+h\right)\sqrt{h}}{2}\right) \tag{2.1.24}$$

Since the cumulative standard normal distribution function is roughly linear in the neighborhood of zero, if follows that (for small maturities h):

$$\omega_o^{imp}(t,t+h) \approx E_t \gamma(t,t+h)$$

This yields an interpretation of the Black-Scholes implied volatility $\omega_a^{imp}\left(t,t+h\right)$ as an implied average volatility :

$$\omega_o^{imp}(t, t+h) \approx E_t \left[\frac{1}{h} \int_t^{t+h} \sigma_{S\tau}^2 d\tau \right]^{\frac{1}{2}}$$
 (2.1.25)

2.2. Some Stylized Facts

The search for model specification and selection is always guided by empirical stylized facts. A model's ability to reproduce such stylized facts is a desirable feature and failure to do so is most often a criterion to dismiss a specification although one typically does not try to fit or explain all possible empirical regularities at once with a single model. Stylized facts about volatility have been well documented in the ARCH literature, see for instance Bollerslev, Engle and Nelson (1994). Empirical regularities regarding derivative securities and implied volatilities are also well covered for instance by Bates (1995a). In this section we will summarize empirical stylized facts, complementing and updating some of the material covered in the aforementioned references.

¹²We do not explicitly study here the dependence between $\omega^{imp}(t, t + h)$ and the various related processes: C_t , S_t , x_t . This is the reason why, for sake of simplicity, this dependence is not apparent in the notation $\omega^{imp}(t, t + h)$.

(a) Thick tails

Since the early sixties it was observed, notably by Mandelbrot (1963), Fama (1963, 1965), among others that asset returns have leptokurtic distributions. As a result, numerous papers have proposed to model asset returns as i.i.d. draws from fat-tailed distributions such as Paretian or Lévy.

(b) Volatility clustering

Any casual observations of financial time series reveals bunching of high and low volatility episodes. In fact, volatility clustering and thick tails of asset returns are intimately related. Indeed, the latter is a static explanation whereas a key insight provided by ARCH models is a formal link between dynamic (conditional) volatility behavior and (unconditional) heavy tails. ARCH models, introduced by Engle (1982) and the numerous extensions thereafter as well as SV models are essentially built to mimic volatility clustering. It is also widely documented that ARCH effects disappear with temporal aggregation, see e.g. Diebold (1988) and Drost and Nijman (1993).

(c) Leverage effects

A phenomenon coined by Black (1976) as the leverage effect suggests that stock price movements are negatively correlated with volatility. Because falling stock prices imply an increased leverage of firms it is believed that this entails more uncertainty and hence volatility. Empirical evidence reported by Black (1976), Christie (1982) and Schwert (1989) suggests, however, that leverage alone is too small to explain the empirical asymmetries one observes in stock prices. Others reporting empirical evidence regarding leverage effects include Nelson (1991), Gallant, Rossi and Tauchen (1992, 1993), Campbell and Kyle (1993) and Engle and Ng (1993).

(d) Information arrivals

Asset returns are typically measured and modeled with observations sampled at fixed frequencies such as daily, weekly or monthly observations. Several authors, including Mandelbrot and Taylor (1967) and Clark (1973) suggested to link asset returns explicitly to the flow of information arrivals. In fact it was already noted that Clark proposed one of the early examples of SV models. Information arrivals are non-termination through time and quite often not directly observable. Conceptually, one can think of asset price movements as the realization of a process $Y_i = Y_{Z_i}$ where Z_i is a so-called directing process. This positive nondecreasing stochastic process Z_t can be thought of as being related to the arrival of information. This idea of time deformation or subordinated stochastic processes was used by Mandelbrot and Taylor (1967) to explain fat tailed returns, by Clark (1973) to explain volatility and was recently refined and further explored by Ghysels, Gouriéroux and Jasiak (1995a). Moreover, Easley and O'Hara (1992) provide a microstructure model involving time deformation. In practice, it suggests a direct link between market volatility and (1) trading volume, (2) quote arrivals, (3) forecastable events such as dividend announcements or macroeconomic data releases, (4) market closures, among many other phenomena linked to information arrivals.

Regarding trading volume and volatility there are several papers documenting stylized facts notably linking high trading volume with market volatility, see for example Karpoff (1987) or Gallant, Rossi and Tauchen (1992).¹³ The intraday patterns of volatility and market activity measured for instance by quote arrivals is also well-known and documented. Wood, McInish and Ord (1985) and Harris (1986) studied this phenomenon for securities markets and found a U-shaped pattern with volatility typically

¹³There are numerous models, theoretical and empirical, linking trading volume and seest returns which we cannot discuss in detail. A partial list includes Foster and Viswanathan (1993a,b), Ghysels and Jasiak (1994a,b), Hausman and Lo (1991), Huffman (1987), Lamoureux and Lastrapes (1990, 1993), Wang (1993) and Andersen (1995).

high at the open and close of the market. The around the clock trading in foreign exchange markets also yields a distinct volatility pattern which is tied with the intensity of market activity and produces strong seasonal patterns. The intradaily patterns for FX markets are analyzed for instance by Müller et al. (1990), Baillie and Bollerslev (1991), Harvey and Huang (1991), Dacorogna et al. (1993), Bollerslev and Ghysels (1994), Andersen and Bollerslev (1995), Ghysels, Gouriéroux and Jasiak (1995b) among others. Another related empirical stylized fact is that of overnight and weekend market closures and their effect on volatility. Fama (1965) and French and Roll (1986) have found that information accumulates more slowly when the NYSE and AMEX are closed resulting in higher volatility on those markets after weekends and holidays. Similar evidence for FX markets has been reported by Baillie and Bollerslev (1989). Finally, numerous papers documented increased volatility of financial markets around dividend announcements (Cornell (1978), Patell and Wolfson (1979,1981)) and macroeconomic data releases (Harvey and Huang (1991, 1992), Ederington and Lee (1993)).

(e) Long memory and persistence

Generally speaking volatility is highly persistent. Particularly for high frequency data one finds evidence of near unit root behavior of the conditional variance process. In the ARCH literature numerous estimates of GARCH models for stock market, commodities, foreign exchange and other asset price series are consistent with an IGARCH specification. Likewise, estimation of stochastic volatility models show similar patterns of persistence (see for instance Jacquier, Polson and Rossi (1994)). These findings have led to a debate regarding modelling persistence in the conditional variance process either via a unit root or a long memory process. The latter approach has been suggested both for ARCH and SV models, see Baillie, Bollerslev and Mikkelsen (1993), Breidt et al. (1993), Harvey (1993) and Comte and Renault (1995). Ding, Granger and Engle (1993) studied the serial correlations of $|r(t, t+1)|^c$ for positive values of c where r(t, t+1) is a one-period return on a speculative asset. They found $|r(t, t+1)|^c$ to have quite high autocorrelations for long lags while the strongest temporal dependence was for c close to one. This result initially found for daily S&P500 return series was also shown to hold for other stock market indices, commodity markets and foreign exchange series (see Granger and Ding (1994)).

(f) Volatility comovements

There is an extensive literature on international comovements of speculative markets. Concerns whether globalization of equity markets increase price volatility and correlations of stock returns has been the subject of many recent studies including, von Fustenberg and Jean (1989), Hamao, Masulis and Ng (1990), King, Sentana and Wadhwani (1994), Harvey, Ruiz and Sentana (1992), Lin, Engle and Ito (1994). Typically one uses factor models to model the commonality of international volatility, as in Diebold and Nerlove (1989), Harvey, Ruiz and Sentana (1992), Harvey, Ruiz and Shephard (1994) or explores so-called common features, see e.g. Engle and Kozicki (1993) and common trends as studied by Bollerslev and Engle (1993).

(g) Implied volatility correlations

Stylized facts are typically reported as model-free empirical observations.¹⁴ Implied volatilities are obviously model-based as they are calculated from a pricing equation of a specific model, namely the Black and Scholes model as noted in section 2.1.3. Since they are computed on a daily basis there is obviously an internal inconsistency since the model presumes constant volatility. Yet, since many option prices are in fact quoted through their implied volatilities it is natural to study the time series behavior of the latter. Often one computes a composite measure since synchronous option prices with different strike prices and

¹⁴This is in some part fictitious even for macroeconomic data for instance when they are detrended or seasonally adjusted. Both detrending and seasonal adjustment are model-based. For the potentially severe impact of detrending on stylized facts see Canova (1992) and Harvey and Jeeger (1993) and for the effect of seasonal adjustment on empirical regularities see Chysche et al. (1993).

maturities for the same underlying asset yield different implied volatilities. The composite measure is usually obtained from a weighting scheme putting more weight on the near-the-money options which are the most heavily traded on organized markets.¹⁵

The time series properties of implied volatilities obtained from stock, stock index and currency options are quite similar. They appear stationary and are well described by a first order autoregressive model (see Merville and Pieptea (1989) and Sheikh (1993) for stock options, Poterba and Summers (1986), Stein (1989), Harvey and Whaley (1992) and Diz and Finucane (1993) for the S&P100 contract and Taylor and Xu (1994), Campa and Chang (1995) and Jorion (1995) for currency options). It was noted from equation (2.1.25) that implied (average) volatilities are expected to contain information regarding future volatility and therefore should predict the latter. One typically tests such hypotheses by regressing realized volatilities on past implied ones.

The empirical evidence regarding the predictable content of implied volatilities is mixed. The time series study of Lamoureux and Lastrapea (1993) considers options on non-dividend paying stocks and compared the forecasting performance of GARCH, implied volatility and historical volatility estimates and found that implied volatilities forecasts, though they are biased as one would expect from (2.1.25), outperform the others. In sharp contrast, Canina and Figlewski (1993) studied S&P100 index call options for which there is an extremely active market. They found that implied volatilities were virtually useless in forecasting future realized volatilities of the S&P100 index. In a different setting using weekly sampling intervals for S&P100 option contracts and a different sample Day and Lewis (1992) not only found that implied volatilities had a predictive content but also were unbiased. Studies examining options on foreign currencies, such as Jorion (1995) also found that implied volatilities were predicting future realizations and GARCH as well as historical volatilities were not outperforming the implied measures of volatility.

(h) The term structure of implied volatilities

The Black-Scholes model predicts a flat term structure of volatilities. In reality, the term structure of at-the-money implied volatilities is typically upward sloping when short term volatilities are low and the reverse when they are high (see Stein(1989)). Taylor and Xu (1994) found that the term structure of implied volatilities from foreign currency options reverses slope every few months. Stein (1989) also found that the actual sensitivity of medium to short term implied volatilities was greater than the estimated sensitivity from the forecast term structure and concluded that medium term implied volatilities overreacted to information. Diz and Finucane (1993) used different estimation techniques and rejected the overreaction hypothesis, even reported evidence suggesting underreaction.

(i) Smiles

If option prices in the market were conformable with the Black-Scholes formula, all the Black-Scholes implied volatilities corresponding to various options written on the same asset would coincide with the volatility parameter σ of the underlying asset. In reality this is not the case, and the Black-Scholes implied volatility $w^{imp}(t,t+h)$ defined by (2.1.23) heavily depends on the calendar time t, the time to maturity h and the moneyness $x_t = LogS_t/KB(t,t+h)$ of the option. This may produce various biases in option pricing or hedging when BS implied volatilities are used to evaluate new options with different strike prices K and maturities h. These price distortions, well-known to practitioners, are usually documented in the empirical literature under the terminology of the *smile effect*, where the so-called "smile" refers to the U-shaped pattern of implied volatilities across different strike prices. More precisely, the following stylized facts are extensively documented (see for instance Rubinstein (1985), Clewlow and Xu (1993), Taylor and

¹⁸Different weighting schemes have been suggested, see for instance Latane and Rendleman (1976), Chiras and Manaster (1978), Beckers (1981), Whaley (1982), Day and Lewis (1988), Engle and Mustafa (1992) and Bates (1995b).

Xu (1993)):

- The U-shaped pattern of w^{imp} (t, t + h) as a function of K (or log K) has its minimum centered at near - the - money options (discounted K close to S_t, i.e. x_t close to zero).
- The volatility smile is often but not always symmetric as a function of log K (or of x_t). When the smile is asymmetric, the skewness effect can often be described as the addition of a monotonic curve to the standard symmetric smile: if a decreasing curve is added, implied volatilities tend to rise more for decreasing than for increasing strike prices and the implied volatility curve has its minimum out of the money. In the reverse case (addition of an increasing curve), implied volatilities tend to rise more with increasing strike prices and their minimum is in the money.
- The amplitude of the smile increases quickly when time to maturity decreases. Indeed, for short
 maturities the smile effect is very pronounced (BS implied volatilities for synchronous option prices
 may vary between 15% and 25%) while it almost completely disappears for longer maturities.

It is widely believed that volatility smiles have to be explained by a model of stochastic volatility. This is natural for several reasons: First, it is tempting to propose a model of stochastically time varying volatility to account for stochastically time varying BS implied volatilities. Moreover, the decreasing amplitude of the smile being a function of time to maturity is conformable with formula like (2.1.25). Indeed, it shows that, when time to maturity is increased, temporal aggregation of volatilities erases conditional heteroskedasticity, which decreases the smile phenomenon. Finally, the skewness itself may also be attributed to the stochastic feature of the volatility process and overall to the correlation of this process with the price process (the so-called leverage effect). Indeed, this effect, while sensible for stock prices data, is small for interest rate and exchange rate series which is why the skewness of the smile is more often observed for options written on stocks.

Nevertheless, it is important to be cautious about tempting associations: stochastic implied volatility and stochastic volatility; asymmetry in stocks and skewness in the smile. As will be discussed in section 4, such analogies are not always rigorously proven. Moreover, other arguments to explain the smile and its skewness (jumps, transaction costs, bid-ask spreads, non-synchronous trading, liquidity problems, ...) have also to be taken in account both for theoretical reasons and empirical ones. For instance, there exists empirical evidence suggesting that the most expensive options (the upper parts of the smile curve) are also the least liquid; skewness may therefore be attributed to specific configurations of liquidity in option markets.

2.3. Information sets

So far we left the specification of information sets vague. This was done on purpose to focus on one issue at the time. In this section we need to be more formal regarding the definition of information since it will allow us to clarify several missing links between the various SV models introduced in the literature and also between SV and ARCH models. We know that SV models emerged from research looking at a very diverse set of issues. In this section we will try to define a common thread and a general unifying framework. We will accomplish this through a careful analysis of information sets and associate with it notions of non-causality in the Granger sense. These causality conditions will allows us to characterize in section 2.4 the distinct features of ARCH and SV models. 16

¹⁶The analysis in this section has some features in common with Andersen (1992) regarding the use of information sets to clarify the difference between SV and ARCH type models.

2.3.1. State variables and information sets

The Hull and White (1987) model is a simple example of a derivative asset pricing model where the stock price dynamics are governed by some unobservable state variables, such as random volatility. More generally, it is convenient to assume that a multivariate diffusion process U_l summarizes the relevant state variables in the sense that:

$$\begin{cases}
dS_t/S_t = \mu_t dt + \sigma_t dW_t \\
dU_t = \gamma_t dt + \delta_t dW_t^U \\
Cov \left(dW_t, dW_t^U\right) = \rho_t dt
\end{cases}$$
(2.3.1)

where the stochastic processes μ_t , σ_t , γ_t , δ_t and ρ_t are $I_t^U = [U_\tau, \tau \le t]$ adapted (Assumption 2.3.1). This means that the process U summarizes the whole dynamics of the stock price process S (which justifies the terminology "state" variable) since, for a given sample path $(U_\tau)_{0 \le \tau \le T}$ of state variables, consecutive returns $S_{t_{k+1}}/S_{t_k}$, $0 \le t_1 < t_2 < ... < t_k \le T$ are stochastically independent and log-normal (as in the benchmark BS model).

The arguments of section 2.1.2 can be extended to the state variables framework (see Garcia and Renault (1995)) discussed here. Indeed, such an extension provides a theoretical justification for the common use of the Black and Scholes model as a standard method of quoting option prices via their implied volatilities.¹⁷ In fact, it is a way of introducing neglected heterogeneity in the BS option pricing model (see Renault (1995) who draws attention to the similarities with introducing heterogeneity in microeconometric models of labor markets, etc.).

In continuous time models, available information at time t for traders (whose information determines option prices) is characterized by continuous time observations of both the state variable sample path and stock price process sample path; namely:

$$I_{\epsilon} = \sigma \left[U_{\tau}, S_{\tau} \; ; \; \tau \le t \right] \tag{2.3.2}$$

2.3.2. Discrete sampling and Granger noncausality

In the next section we will treat explicitly discrete time models. It will necessitate formulating discrete time analogues of equation (2.3.1). The discrete sampling and Granger non causality conditions discussed here will bring us a step closer to building a formal framework for statistical modelling using discrete time data. Clearly, a discrete time analogue of equation (2.3.1) is:

$$\log S_{t+1}/S_t = \mu(U_t) + \sigma(U_t)\varepsilon_{t+1} \tag{2.3.3}$$

provided we impose some restrictions on the process ϵ_i . The restrictions we want to impose must be flexible enough to accommodate phenomena such as leverage effects for instance. A setup that does this is the following:

Assumption 2.3.2.A: The process ε_t in (2.3.3) is i.i.d. and not Granger caused by the state variable process U_t .

¹⁷Garcia and Renault (1995) argued that Assumption 2.3.1 is essential to ensure the homogeneity of option prices with respect to the pair (stock price, strike price) which in turn ensures that BS implied volatilities do not depend on the stock price level but only on the moneyness S/K. This homogeneity property was first emphasized by Merton (1973).

Assumption 2.3.2.B: The process ϵ_t in (2.3.3) does not Granger cause U_t .

Assumption 2.3.2.B is useful for the practical use of BS implied volatilities as it is the discrete time analogue of Assumption 2.3.1 where it is stated that the coefficients of the process U are I_t^U adapted (for further details see Garcia and Renault (1995)). Assumption 2.3.2.A is important for the statistical interpretation of the functions $\mu(U_t)$ and $\sigma(U_t)$ respectively as trend and volatility coefficients. Namely,

$$E [\log S_{t+1}/S_t \mid (S_{\tau}/S_{\tau-1}; \tau \le t)] = E [E [\log S_{t+1}/S_t \mid (U_{\tau}, \varepsilon_{\tau}; \tau \le t)] \mid (S_{\tau}/S_{\tau-1}; \tau \le t)] = E [\mu(U_t) \mid (S_t/S_{t-1}; \tau \le t)]$$
(2.3.4)

since $E\left[\epsilon_{i+1}\mid (U_{\tau}, \epsilon_{\tau}; \tau \leq t)\right] = E\left[\epsilon_{i+1}\mid \epsilon_{i}; \tau \leq t\right] = 0$ due to the Granger noncausality from U_{t} to ϵ_{t} of Assumption 2.3.2.A. Likewise, one can easily show that

$$Var \left[\log S_{t+1} / S_t - \mu(U_t) \mid (S_{\tau} / S_{\tau-1}; \tau \le t) \right]$$

$$= E \left[\sigma^2(U_t) \mid (S_{\tau} / S_{\tau-1}; \tau \le t) \right]$$
(2.3.5)

Implicitly we have introduced a new information set in (2.3.4) and (2.3.5) which besides I_t defined in (2.3.2) will be useful as well for further analysis. Indeed, one often confines (statistical) analysis to information conveyed by a discrete time sampling of stock return series which will be denoted by the information set

$$I_t^R \equiv \sigma \left[S_\tau / S_{\tau-1} : \tau = 0, 1, ..., t-1, t \right]$$
 (2.3.6)

where the superscript R stands for returns. By extending Andersen (1994), we shall adopt as the most general framework for univariate volatility modelling, the setup given by the Assumptions 2.3.2.A, 2.3.2.B and:

Assumption 2.3.2. $C: \mu(U_t)$ is I_t^R measurable.

Therefore in (2.3.4) and (2.3.5) we have essentially shown that :

$$E\left[\log S_{t+1}/S_t \mid I_t^R\right] = \mu(U_t)$$
 (2.3.7)

$$Var\left[\left(\log S_{t+1}/S_{t}\right)\left|I_{t}^{R}\right|=E\left|\sigma^{2}\left(U_{t}\right)\right|I_{t}^{R}\right]$$
(2.3.8)

2.4. Statistical Modelling of Stochastic Volatility

Financial time series are observed at discrete time intervals while a majority of theoretical models are formulated in continuous time. Generally speaking there are two statistical methodologies to resolve this tension. Either one considers for the purpose of estimation statistical discrete time models of the continuous time processes. Alternatively, the statistical model may be specified in continuous time and inference is done via a discrete time approximation. In this section we will discuss in detail the former approach while the latter will be introduced in section 4. The class of discrete time statistical models discussed here is general. In section 2.4.1 we introduce some notation and terminologies. The next section discuss the so-called stochastic autoregressive volatility model introduced by Andersen (1994) as a reather general and flexible semi-parametric framework to encompass various representations of stochastic volatility already

available in the literature. Identification of parameters and the restrictions required for it are discussed in section 2.4.3

2.4.1. Notation and Terminology

In section 2.3, we left unspecified the functional forms which the trend $\mu(\cdot)$ and volatility $\sigma(\cdot)$ take. Indeed, in some sense we built a nonparametric framework recently proposed by Lezan, Renault and de Vitry (1995) which they introduced to discuss a notion of stochastic volatility of unknown form. This nonparametric framework encompasses standard parametric models (see section 2.4.2 for more formal discussion). For the purpose of illustration let us consider two extreme cases, assuming for simplicity that $\mu(U_t) = 0$: (i) the discrete time analogue of the Hull and White model (2.1.16) is obtained when $\sigma(U_t) = \sigma_t$ is a stochastic process independent from the stock return standardized innovation process ϵ and (ii) σ_t may be a deterministic function $h(\epsilon_t, \tau \leq t)$ of past innovations. The latter is the complete opposite of (i) and leads to a large variety of choices of parametrized functions for h yielding X-ARCH models (GARCH, EGARCH, QTARCH, Periodic GARCH, etc.).

Besides these two polar cases where Assumption 2.3.2.A is fulfilled in a trivial degenerate way, one can also accommodate leverage effects. ¹⁹ In particular the contemporaneous correlation structure between innovations in U and the return process can be nonzero, since the Granger non-causality assumptions deal with temporal causal links rather than contemporaneous ones. For instance, we may have $\sigma(U_t) = \sigma_t$ with:

$$\log S_{t+1}/S_t = \sigma_t \varepsilon_{t+1} \tag{2.4.1}$$

$$Cov\left(\sigma_{t+1}, \varepsilon_{t+1} | \mathbf{I}_{t}^{R}\right) \neq 0$$
 (2.4.2)

A negative covariance in (2.4.2) is a standard case of leverage effect, without violating the non-causality Assumptions 2.3.2.A and B.

A few concluding observations are worth making to deal with the burgeoning variety of terminologies in the literature. First, we have not considered the distinction due to Taylor (1994) between "lagged autoregressive random variance models" given by (2.4.1) and "contemporaneous autoregressive random variance models" defined by:

$$\log S_{t+1}/S_t = \sigma_{t+1}\varepsilon_{t+1} \tag{2.4.3}$$

Indeed, since the volatility process σ_t is unobservable, the settings (2.4.1) and (2.4.3) are observationally equivalent as long as they are not completed by precise (non)-causality assumptions. For instance: (i) (2.4.1) and assumption 2.3.2.A together appear to be a correct and very general definition of a SV model possibly completed by Assumption 2.3.2.B for option pricing and (2.4.2) to introduce leverage effects, (ii) (2.4.3) associated with (2.4.2) would not be a correct definition of a SV model since in this case in general: $E\begin{bmatrix} \log S_{t+1}/S_t \mid I_t^R \end{bmatrix} \neq 0$, and the model would introduce via the process σ a forecast which is related not only to volatility but also to the expected return.

¹⁸Lesan, Renault and de Vitry (1995) discuss in detail how to recover phenomena such as volatility clustering in this framework. As a nonparametric framework it also has certain advantages regarding (robust) estimation. They develop for instance methods that can be useful as a first estimation step for efficient algorithms assuming a specific parametric model (see Section 5).

¹⁹Assumption 2.3.2.B is fulfilled in case (i) but may fail in the GARCH case (ii). When it fails to hold in the latter case it makes the GARCH framework not very well-suited for option pricing.

For notational simplicity, the framework (2.4.3) will be used in section 3 with the leverage effect captured by Cov $(\sigma_{t+1}, \varepsilon_t) \neq 0$ instead of Cov $(\sigma_{t+1}, \varepsilon_{t+1}) \neq 0$. Another terminology was introduced by Amin and Ng (1993) for option pricing. Their distinction between "predictable" and "unpredictable" volatility is very close to the leverage effect concept and can also be analyzed through causality concepts as discussed in Garcia and Renault (1995). Finally, it will not be necessary to make a distinction between weak, semi-strong and strong definitions of SV models in analogy with their ARCH counterparts (see Drost and Nijman (1993)). Indeed, the class of SV models as defined here can accommodate parametrizations which are closed under temporal aggregation (see also section 4.1 on the subject of temporal aggregation).

2.4.2. Stochastic Autoregressive Volatility

For simplicity, let us consider the following univariate volatility process:

$$y_{t+1} = \mu_t + \sigma_t \varepsilon_{t+1} \tag{2.4.4}$$

where μ_t is a measurable function of observables $y_t \in I_t^R$, $\tau \leq t$. While our discussion will revolve around (2.4.4), we will discuss several issues which are general and not confined to that specific model; extensions will be covered more explicitly in section 3.5. Following the result in (2.3.8) we know that:

$$Var\left[y_{t+1}|I_t^R\right] = E\left[\sigma_t^2|I_t^R\right] \tag{2.4.5}$$

suggesting (1) that volatility clustering can be captured via autoregressive dynamics in the conditional expectation (2.4.5) and (2) that thick tails can be obtained in either one of three ways, namely (a) via heavy tails of the white noise ε_t distribution, (b) via the stochastic features of $E\left[\sigma_t^2|I_t^R\right]$ and (c) via specific randomness of the volatility process σ_t which makes it latent i.e. $\sigma_t \notin I_t^{R,20}$. The volatility dynamics that follow from (1) and (2) are usually an AR(1) model for some nonlinear function of σ_t . Hence, the volatility process is assumed to be stationary and Markovian of order one but not necessarily linear AR(1) in σ_t itself. This is precisely what motivated Andersen (1994) to introduce the Stochastic Autoregressive Variance or SARV class of models where σ_t (or σ_t^2) is a polynomial function $g\left(K_t\right)$ of a Markov process K_t with the following dynamic specification:

$$K_{t} = w + \beta K_{t-1} + [\gamma + \alpha K_{t-1}] u_{t}$$
 (2.4.6)

where $\bar{u}_t = u_t - 1$ is zero-mean white noise with unit variance. Andersen (1994) discusses sufficient regularity conditions which ensure stationarity and ergodicity for K_t . Without entering into the details, let us note that the fundamental non-causality Assumption 2.3.2A implies that the u_t process in (2.4.6) does not Granger cause ε_t in (2.4.4). In fact, the non-causality condition suggests a slight modification of Andersen's (1994) definition. Namely, it suggests assuming ε_{t+1} independent of u_{t-1} , $j \geq 0$ for the conditional probability distribution, given ε_{t-j} , $j \geq 0$ rather than for the unconditional distribution. This modification does not invalidate Andersen's SARV class of models as the most general parametric statistical model studied so far in the volatility literature. The GARCH(1,1) model is straightforwardly obtained from (2.4.6) by letting $K_t = \sigma_t^2$, $\gamma = 0$ and $u_t = \varepsilon_t^2$. Note that the deterministic relationship $u_t = \varepsilon_t^2$ between the stochastic components of (2.4.4) and (2.4.6) emphasizes that, in GARCH models, there is no randomness

²⁰Kim and Shephard (1994), using data on weekly returns on the S&P500 Index, found that a t-GARCH model has an almost identical likelihood as the normal based SV model. This example shows that a specific randomness in σ_t may produce the same level of marginal kurtoeis as a heavy tailed student distribution of the white noise ε .

specific to the volatility process. The Autoregressive Random Variance model popularized by Taylor (1986) also belongs to the SARV class. Here:

$$\log \sigma_{t+1} = \xi + \phi \log \sigma_t + \eta_{t+1} \tag{2.4.7}$$

where η_{t+1} is a white noise disturbance such that $Cov(\eta_{t+1}, \varepsilon_{t+1}) \neq 0$ to accommodate leverage effects. This is a SARV model with $K_t = \log \sigma_t$, $\alpha = 0$ and $\eta_{t+1} = \gamma u_{t+1}$.²¹

2.4.3. Identification of parameters

Introducing a general class of processes for volatility, like the SARV class discussed in the previous section prompts questions regarding identification. Suppose again that

$$y_{t+1} = \sigma_t \varepsilon_{t+1}
\sigma_t^q = g(K_t), q \in \{1, 2\}
K_t = w + \beta K_{t-1} + [\gamma + \alpha K_{t-1}] u_t.$$
(2.4.8)

Andersen (1994), noted the model is better interpreted by considering the zero-mean white noise process $\hat{u}_t = u_t - 1$:

$$K_{t} = (w + \gamma) + (\alpha + \beta) K_{t-1} + (\gamma + \alpha K_{t-1}) \hat{\mathbf{u}}_{t}. \qquad (2.4.9)$$

It is clear from the latter that it may be difficult to distinguish empirically the constant w from the "stochastic" constant γu_t . Similarly, the identification of the α and β parameters separately is also problematic as $(\alpha + \beta)$ governs the persistence of shocks to volatility. These identification problems are usually resolved by imposing (arbitrary) restrictions on the pairs of parameters (w, γ) and (α, β) .

The GARCH(1,1) and Autoregressive Random Variance specifications assume that $\gamma=0$ and $\alpha=0$ respectively. Identification of all parameters without such restrictions generally requires additional constraints, for instance via some distributional assumptions on ϵ_{t+1} and u_t , which restrict the semi-parametric framework of (2.4.6) into a parametric statistical model.

To address more rigorously the issue of identification, it is useful to consider, according to Andersen (1994), the following reparametrization (assuming for notational convenience that $\alpha \neq 0$):

$$\begin{cases} K = (w+\gamma)/(1-\alpha-\beta) \\ \rho = \alpha+\beta \\ \delta = \gamma/\alpha \end{cases}$$
 (2.4.10)

Hence equation (2.4.9) can be rewritten as :

$$K_t = K + \rho (K_{t-1} - K) + (\delta + K_{t-1}) \bar{U}_t$$

where $\tilde{U}_t = \alpha \tilde{u}_t$.

It is clear from (2.4.10), that only three functions of the original parameters α, β, γ, w may be identified and that the three parameters K, ρ, δ are identified from the first three unconditional moments of the process K_i for instance.

²¹Andersen (1994) also shows that the SARV framework encompasses another type of random variance model that we have considered as ill-specified since it combines (2.4.2) and (2.4.3).

To give to these identification results an empirical content, it is essential to know: (1) how to go from the moments of the observable process Y_t to the moments of the volatility process σ_t and (2) how to go from the moments of the volatility process σ_t to the moments of the latent process K_t . The first point is easily solved by specifying the corresponding moments of the standardized innovation process ε . If we assume for instance a Gaussian probability distribution, we obtain that:

$$\begin{cases}
E |y_t| = \sqrt{2/\pi} & E\sigma_t \\
E |y_t| |y_{t-j}| = 2/\pi & E (\sigma_t\sigma_{t-j}) \\
E |y_t^2| |y_{t-j}| = \sqrt{2/\pi} & E (\sigma_t^2\sigma_{t-j})
\end{cases} (2.4.11)$$

The solution of the second point requires in general the specification of the mapping g and of the probability distribution of u_t in (2.4.6). For the so-called Log-normal SARV model, it is assumed that $\alpha = 0$ and $K_t = \log \sigma_t$ (Taylor's autoregressive random variance model) and that u_t is normally distributed (Log-normality of the volatility process). In this case, it is easy to show that:

$$\begin{cases} E\sigma_t^n = \exp\left[n EK_t + n^2 Var K_t/2\right] \\ E\left(\sigma_t^m \sigma_{t-j}^n\right) = E\sigma_t^m E\sigma_{t-j}^n \exp\left[mnCov\left(K_t, K_{t-j}\right)\right] \\ Cov\left(K_t, K_{t-j}\right) = \beta^j Var K_t \end{cases}$$
(2.4.12)

Without the normality assumption (i.e. QML, mixture of normal, Student distribution ...) this model will be studied in much more detail in sections 3 and 5 from both probabilistic and statistical points of view. Moreover, this is a template for studying other specifications of the SARV class of models. In addition, various specifications will be considered in section 4 as proxies of continuous time models.

3. Discrete Time Models

The purpose of this section will be to discuss the statistical handling of discrete time SV models, using simple univariate cases. We start by defining the most basic SV model corresponding the autoregressive random variance model discussed earlier in (2.4.7). We study its statistical properties in section 3.2 and provide a comparison with ARCH models in section 3.3. Section 3.4 is devoted to filtering, prediction and smoothing. Various extensions, including multivariate models, are covered in the last section. Estimation of the parameters governing the volatility process is discussed later in section 5.

3.1. The Discrete Time SV Model

The discrete time SV model may be written as

$$y_t = \sigma_t \varepsilon_t, \qquad t = 1, ..., T, \tag{3.1.1}$$

where y_t denotes the demeaned return process $y_t = \log(S_t/S_{t-1}) - \mu$ and $\log \sigma_t^2$ follows an AR(1) process. It will be assumed that e_t is a series of independent, identically distributed random disturbances. Usually e_t is specified to have a standard distribution so its variance σ_t^2 is known. Thus for a normal distribution σ_t^2 is unity while for a t-distribution with ν degrees of freedom it will be $\nu/(\nu-2)$. Following a convention often adopted in the literature we write for $h_t \equiv \log \sigma_t^2$:

$$y_t = \sigma \varepsilon_t e^{0.6h_t} \tag{3.1.2}$$

where σ is a scale parameter, which removes the need for a constant term in the stationary first-order autoregressive process

$$h_{t+1} = \phi h_t + \eta_t, \eta_t \sim IID(0, \sigma_\eta^2), \quad |\phi| < 1.$$
 (3.1.3)

It was noted before that if ε_t and η_t are allowed to be correlated with each other, the model can pick up the kind of asymmetric behavior which is often found in stock prices. Indeed a negative correlation between ε_t and η_t induces a leverage effect. As in section 2.4.1, the timing of the disturbance in (3.1.3) ensures that the observations are still a martingale difference, the equation being written in this way so as to tie in with the state space literature.

It should be stressed that the above model is only an approximation to the continuous time models of section 2 observed at discrete intervals. The accuracy of the approximation is examined in Dassios (1995) using Edgeworth expansions (see also sections 4.1 and 4.3 for further discussion).

3.2. Statistical Properties

The following properties of the SV model hold even if ε_t and η_t are contemporaneously correlated. Firstly, as noted, y_t is a martingale difference. Secondly, stationarity of h_t implies stationarity of y_t . Thirdly, if η_t is normally distributed, it follows from the properties of the lognormal distribution that $E[\exp(ah_t)] = \exp(a^2\sigma_h^2/2)$, where a is a constant and σ_h^2 is the variance of h_t . Hence, if ε_t has a finite variance, the variance of y_t is given by

$$Var(y_t) = \sigma^2 \sigma_t^2 \exp(\sigma_h^2/2) \tag{3.2.1}$$

where σ_{ϵ}^2 is assumed known, e.g. $\sigma_{\epsilon}^2 = 1$ when ϵ_t i.i.d. N(0,1). Similarly if the fourth moment of ϵ_t exists, the kurtosis of y_t is $\kappa \exp(\sigma_{k}^2)$, where κ is the kurtosis of ϵ_t , so y_t exhibit more kurtosis than ϵ_t . Finally all the odd moments are zero.

For many purposes we need to consider the moments of powers of absolute values. Again, η_t is assumed to be normally distributed. Then for ϵ_t having a standard normal distribution, the following expressions are derived in Harvey (1993):

$$E \mid y_t \mid^c = \sigma^c 2^{c/2} \frac{\Gamma\left(\frac{c}{2} + \frac{1}{2}\right)}{\Gamma\left(\frac{1}{2}\right)} \exp\left(\frac{c^2}{8} \sigma_h^2\right), c > -1, c \neq 0$$
(3.2.2)

and

$$Var\mid y_t\mid^c=\sigma^{2c}2^c\exp\left(\frac{c^2}{2}\sigma_h^2\right)\left\{\frac{\Gamma\left(c+\frac{1}{2}\right)}{\Gamma\left(\frac{1}{2}\right)}-\left[\frac{\Gamma\left(\frac{c}{2}+\frac{1}{2}\right)}{\Gamma\left(\frac{1}{2}\right)}\right]^2\right\},c>-0.5,c\neq0$$

Note that $\Gamma(1/2) = \sqrt{\pi}$ and $\Gamma(1) = 1$. Corresponding expressions may be computed for other distributions of ε_i including Student's t and the General Error Distribution (see Nelson (1991)).

Finally, the square of the coefficient of variation of σ_t^2 is often used as a measure of the relative strength of the SV process. This is $Var(\sigma_t^2)/[E(\sigma_t^2)]^2 = \exp(\sigma_h^2) - 1$. Jacquier, Polson and Rossi (1994) argue that this is more easily interpretable than σ_η^2 . In the empirical studies they quote it is rarely less than 0.1 or greater than 2.

3.2.1. Autocorrelation Functions

If we assume that the disturbances ε_t and η_t are mutually independent, and η_t is normal, the ACF of the absolute values of the observations raised to the power c is given by

$$\rho_{\tau}^{(c)} = \frac{E(\mid y_t\mid^c\mid y_{t-\tau}\mid^c) - \{E(\mid y_t\mid^c)\}^2}{E(\mid y_t\mid^2) - \{E(\mid y_t\mid^c)\}^2} = \frac{\exp(\frac{c^2}{4}\sigma_h^2\rho_{h,\tau}) - 1}{\kappa_c \exp(\frac{c^2}{4}\sigma_h^2) - 1}, \tau \ge 1, c > -0.5, c \ne 0$$
 (3.2.3)

where ar is

$$\kappa_c = E(|y_t|^{2\epsilon})/\{E(|y_t|^c)\}^2,$$
(3.2.4)

and $\rho_{h,\tau}$, $\tau=0,1,2,...$ denotes the ACF of h_t . Taylor (1986) gives this expression for c equal to one and two and ϵ_t normally distributed. When c=2, κ_c is the kurtosis and this is three for a normal distribution. More generally,

$$\kappa_c = \Gamma(c + \frac{1}{2})\Gamma(\frac{1}{2})/\{\Gamma(\frac{c}{2} + \frac{1}{2})\}^2, c \neq 0$$

For Student's t-distribution with ν degrees of freedom :

$$\kappa_{c} = \frac{\Gamma(c + \frac{1}{2})\Gamma(-c + \frac{\nu}{2})\Gamma(\frac{1}{2})\Gamma(\frac{\nu}{2})}{\{\Gamma(\frac{c}{2} + \frac{1}{2})\Gamma(-\frac{c}{2} + \frac{\nu}{2})\}^{2}}, |c| < \nu/2, c \neq 0$$
(3.2.5)

Note that ν must be at least five if c is two.

The ACF, $\rho_{\tau}^{(c)}$, has the following features. First, if σ_h^2 is small and/or $\rho_{h,\tau}$ is close to one,

$$\rho_{\tau}^{(c)} \simeq \rho_{h,\tau} \frac{\exp(\frac{c^2}{4}\sigma_h^2) - 1}{(\kappa_c \exp(\frac{c^2}{4}\sigma_h^2) - 1)}, \tau \ge 1;$$
(3.2.6)

compare Taylor (1986, p. 74-5). Thus the shape of the ACF of h_t is approximately carried over to $\rho_r^{(c)}$ except that it is multiplied by a factor of proportionality, which must be less than one for c positive as κ_c is greater than one. Secondly, for the t-distribution, κ_c declines as ν goes to infinity. Thus $\rho_r^{(c)}$ is a maximum for a normal distribution. On the other hand, a distribution with less kurtosis than the normal will give rise to higher values of $\rho_r^{(c)}$.

Although (3.2.6) gives an explicit relationship between $\rho_r^{(c)}$ and c, it does not appear possible to make any general statements regarding $\rho_r^{(c)}$ being maximized for certain values of c. Indeed different values of σ_h^c lead to different values of c maximizing $\rho_r^{(c)}$. If σ_h^2 is chosen so as to give values of $\rho_r^{(c)}$ of a similar size to those reported in Ding, Granger and Engle (1993) then the maximum appears to be attained for c slightly less than one. The shape of the curve relating $\rho_r^{(c)}$ to c is similar to the empirical relationships reported in Ding, Granger and Engle, as noted by Harvey (1993).

3.2.2. Logarithmic Transformation

Squaring the observations in (3.1.2) and taking logarithms gives

$$\log y_t^2 = \log \sigma^2 + h_t + \log \varepsilon_t^2. \tag{3.2.7}$$

Alternatively

$$\log y_t^2 = \omega + h_t + \xi_t, \tag{3.2.8}$$

where $\omega = \log \sigma^2 + E \log \varepsilon_t^2$, so that the disturbance ξ_t has zero mean by construction.

The mean and variance of $\log \varepsilon_t^2$ are known to be -1.27 and $\pi^2/2 = 4.93$ when ε_t has a standard normal distribution; see Abramovitz and Stegun (1970). However, the distribution of $\log \varepsilon_t^2$ is far from being normal, being heavily skewed with a long tail.

More generally, if ε_t has a t-distribution with ν degrees of freedom, it can be expressed as:

$$\varepsilon_t = \zeta_t \kappa_t^{-0.5}$$

where ζ_i is a standard normal variate and κ_i is independently distributed such that $\nu \kappa_i$ is chi-square with ν degrees of freedom. Thus

$$\log \varepsilon_i^2 = \log \zeta_i^2 - \log \kappa_i$$

and again using results in Abramovitz and Stegun (1970), it follows that the mean and variance of $\log \varepsilon_t^2$ are -1.27 $-\psi(\nu/2) - \log(\nu/2)$ and 4.93 $+\psi'(\nu/2)$ respectively, where $\psi(.)$ is the digamma function. Note that the moments of ξ_t exist even if the model is formulated in such a way that the distribution of ε_t is Cauchy, that is $\nu=1$. In fact in this case ξ_t is symmetric with excess kurtosis two, compared with excess kurtosis four when ε_t is Gaussian.

Since $\log e_t^2$ is serially independent, it is straightforward to work out the ACF of $\log y_t^2$ for h_t following any stationary process:

$$\rho_{\tau}^{(0)} = \rho_{h,\tau} / \{1 + \sigma_t^2 / \sigma_h^2\}, \tau \ge 1$$
(3.2.9)

The notation $\rho_r^{(0)}$ reflects the fact that the ACF of a power of an absolute value of the observation is the same as that of the Box-Cox transform, that is $\{|y_t|^c - 1\}/c$, and hence the logarithmic transform of an absolute value, raised to any (non-zero) power, corresponds to c = 0. (But note that one cannot simply set c = 0 in (3.2.3)).

Note that even if η_t and ε_t are not mutually independent, the η_t and ξ_t disturbances are uncorrelated if the joint distribution of ε_t and η_t is symmetric, that is $f(\varepsilon_t, \eta_t) = f(-\varepsilon_t, -\eta_t)$; see Harvey, Ruiz and Shephard (1994). Hence the expression for the ACF in (3.2.9) remains valid.

3.3. Comparison with ARCH models

The GARCH(1,1) model has been applied extensively to financial time series. The variance in (3.1.1) is assumed to depend on the variance and squared observation in the previous time period. Thus

$$\sigma_t^2 = \gamma + \alpha y_{t-1}^2 + \beta \sigma_{t-1}^2, \quad t = 1, ..., T.$$
 (3.3.1)

The GARCH model was proposed by Bollerslev (1986) and Taylor (1986), and is a generalization of the ARCH model formulated by Engle (1982). The ARCH(1) model is a special case of GARCH(1,1) with $\beta=0$. The motivation comes from forecasting, in an AR(1) model with independent disturbances, the optimal prediction of the next observation is a fraction of the current observation, and in ARCH(1) it is a fraction of the current squared observation (plus a constant). The reason is that the optimal forecast is constructed conditional on the current information and in an ARCH model the variance in the next period is assumed to be known. This construction leads directly to a likelihood function for the model once a distribution is assumed for ε_t . Thus estimation of the parameters upon which σ_t^2 depends is straightforward

in principle. The GARCH formulation introduces terms analogous to moving average terms in an ARMA model, thereby making forecasts a function of a distributed lag of past squared observations.

It is straightforward to show that y_t is a martingale difference with (unconditional) variance $\gamma/(1-\alpha-\beta)$. Thus $\alpha+\beta<1$ is the condition for covariance stationarity. As shown in Bollerslev (1986), the condition under which the fourth moment exists in a Gaussian model is $2\alpha^2+(\alpha+\beta)^2<1$. The model then exhibits excess kurtosis. However, the fourth moment condition may not always be satisfied in practice. Somewhat paradoxically, the conditions for strict stationarity are much weaker and, as shown by Nelson (1990), even include the case $\alpha+\beta=1$.

The specification of GARCH(1,1) means that we can write

$$y_t^2 = \gamma + \alpha y_{t-1}^2 + \beta \sigma_{t-1}^2 + v_t = \gamma + (\alpha + \beta) y_{t-1}^2 + v_t - \beta v_{t-1}$$

where $v_t = y_t^2 - \sigma_t^2$ is a martingale difference. Thus y_t^2 has the form of an ARMA(1,1) process and so its ACF can be evaluated in the same way. The ACF of the corresponding ARMA model seems to be indicative of the type of patterns likely to be observed in practice in correlograms of y_t^2 .

The GARCH model extends by adding more lags of σ_t^2 and y_t^2 . However, GARCH(1,1) seems to be the most widely used. It displays similar properties to the SV model, particularly if ϕ is close to one. This should be clear from (3.2.6) which has the pattern of an ARMA(1,1) process. Clearly ϕ plays a role similar to that of $\alpha + \beta$. The main difference in the ACFs seems to show up most at lag one. Jacquier et al. (1994, p. 373) present a graph of the correlogram of the squared weekly returns of a portfolio on the New York Stock Exchange together with the ACFs implied by fitting SV and GARCH(1,1) models. In this case the ACF implied by the SV model is closer to the sample values.

The SV model displays excess kurtosis even if ϕ is zero since y_t is a mixture of distributions. The σ_{η}^{2} parameter governs the degree of mixing independently of the degree of smoothness of the variance evolution. This is not the case with a GARCH model where the degree of kurtosis is tied to the roots of the variance equation, α and β in the case of GARCH(1,1). Hence, it is very often necessary to use a non-Gaussian GARCH model to capture the high kurtosis typically found in a financial time series.

The basic GARCH model does not allow for the kind of asymmetry captured by an SV model with contemporaneously correlated disturbances, though it can be modified as suggested in Engle and Ng (1993). The EGARCH model, proposed by Nelson (1991), handles asymmetry by taking $\log \sigma_t^2$ to be a function of past squares and absolute values of the observations.

3.4. Filtering, Smoothing and Prediction

For the purposes of pricing options, we need to be able to estimate and predict the variance, σ_t^2 , which of course, is proportional to the exponent of h_t . An estimate based on all the observations up to, and possibly including, the one at time t is called a *filtered* estimate. On the other hand an estimate based on all the observations in the sample, including those which came after time t is called a *smoothed* estimate. *Predictions* are estimates of future values. As a matter of historical interest we may wish to examine the evolution of the variance over time by looking at the smoothed estimates. These might be compared with the volatilities implied by the corresponding options prices as discussed in section 2.1.2. For pricing 'at the money' options we may be able to simply use the filtered estimate at the end of the sample and the predictions of future values of the variance, as in the method suggested for ARCH models by Noh, Engle and Kane (1994). More generally, it may be necessary to base prices on the full distribution of future values of the variance, perhaps obtained by simulation techniques; for further discussion see section 4.2.

One can think of constructing filtered and smoothed estimates in a very simple, but arbitrary way, by taking functions (involving estimated parameters) of moving averages of transformed observations. Thus:

$$\widehat{\sigma_t}^2 = g\{\sum_{j=t-1}^r w_{ij} f(y_{t-j})\}, t = 1, ..., T,$$
(3.4.1)

where r = 0 or 1 for a filtered estimate and r = t - T for a smoothed estimate.

Since we have formulated a stochastic volatility model, the natural course of action is to use this as the basis for filtering, smoothing and prediction. For a linear and Gaussian time series model, the state space form can be used as the basis for optimal filtering and smoothing algorithms. Unfortunately, the SV model is nonlinear. This leaves us with three possibilities:

- a. compute inefficient estimates based on a linear state space model;
- b. use computer intensive techniques to estimate the optimal filter to a desired level of accuracy;
- c. use an (unspecified) ARCH model to approximate the optimal filter.

We now turn to examine each of these in some detail.

3.4.1. Linear State Space Form

The transformed observations, the $\log y_t^{b}s$, can be used to construct a linear state space model as suggested by Nelson (1988) and Harvey, Ruiz and Shephard (1994). The measurement equation is (3.2.8) while (3.1.3) is the transition equation. The initial conditions for the state, h_t , are given by its unconditional mean and variance, that is zero and $\sigma_n^2/(1-\phi^2)$ respectively.

While it may be reasonable to assume that η_t is normal, ξ_t would only be normal if the absolute value of ε_t were lognormal. This is unlikely. Thus application of the Kalman filter and the associated smoothers yields estimators of the state, h_t , which are only optimal within the class of estimators based on linear combinations of the $\log y_t^{2s}$. Furthermore, it is not the $h_t's$ which are required, but rather their exponents. Suppose $h_{t|T}$ denotes the smoothed estimator obtained from the linear state space form. Then $\exp(h_{t|T})$ is of the form (3.4.1), multiplied by an estimate of the scaling constant, σ^2 . It can be written as a weighted geometric mean. This makes the estimates vulnerable to very small observations and is an indication of the limitations of this approach.

Working with the logarithmic transformation raises an important practical issue, namely how to handle observations which are zero. This is a reflection of the point raised in the previous paragraph, since obviously any weighted geometric mean involving a zero observation will be zero. More generally we wish to avoid very small observations. One possible solution is to remove the sample mean. A somewhat more satisfactory alternative, suggested by Fuller, and studied by Breidt and Carriquiry (1995), is to make the following transformation based on a Taylor series expansion:

$$\log y_t^2 \cong \log(y_t^2 + cs_v^2) - cs_v^2/(y_t^2 + cs_v^2), \quad t = 1, \dots, T, \tag{3.4.2}$$

where s_y^2 is the sample variance of the $y_t^i s$ and c is a small number, the suggested value being 0.02. The effect of this transformation is to reduce the kurtosis in the transformed observations by cutting down the long tail made up of the negative values obtained by taking the logarithms of the 'inliers'. In other words it is a form of trimming. It might be more satisfactory, to carry out this procedure after correcting the observations for heteroskedasticity by dividing by preliminary estimates, $\widehat{\sigma}_t^{2^{ij}} s$. The $\log \widehat{\sigma}_t^{2^{ij}} s$ are then added

to the transformed observations. The $\widehat{\sigma_t}^{y}$ s could be constructed from a first round or by using a totally different procedure, perhaps a nonparametric one.

The linear state space form can be modified so as to deal with asymmetric models. It was noted earlier that even if η_t and ε_t are not mutually independent, the disturbances in the state space form are uncorrelated if the joint distribution of ε_t and η_t is symmetric. Thus the above filtering and smoothing operations are still valid, but there is a loss of information stemming from the squaring of the observations. Harvey and Shephard (1993) show that this information may be recovered by conditioning on the signs of the observations denoted by s_t , a variable which takes the value +1 (-1) when y_t is positive (negative). These signs are, of course, the same as the signs of the ε_t 's. Let $E_+(E_-)$ denote the expectation conditional on ε_t being positive (negative), and assign a similar interpretation to variance and covariance operators. The distribution of ξ_t is not affected by conditioning on the signs of the ε_t 's, but, remembering that $E(\eta_t|\varepsilon_t)$ is an odd function of ε_t ,

$$\mu^* = E_+(\eta_t) = E_+[E\eta_t]\epsilon_t] = -E_-(\eta_t)$$

and

$$\gamma^* = Cov_+(\eta_t, \xi_t) = E_+(\eta_t \xi_t) - E_+(\eta_t) E(\xi_t) = E_+(\eta_t \xi_t) = -Cov_-(\eta_t, \xi_t),$$

because the expectation of ξ_t is zero and

$$E_{+}(\eta_{t}\xi_{t}) = E_{+}[E(\eta_{t}|\varepsilon_{t})\log\varepsilon_{t}] - \mu^{*}E(\log\varepsilon_{t}) = -E_{-}(\eta_{t}\xi_{t}).$$

Finally

$$Var_+\eta_t = E_+(\eta_t^2) - [E_+(\eta_t)]^2 = \sigma_n^2 - \mu^{*2}.$$

The linear state space form is now

$$\begin{aligned}
\log y_t^2 &= \omega + h_t + \xi_t \\
h_{t+1} &= \phi h_t + s_t \mu^* + \eta_t^* \\
\begin{pmatrix} \xi_t \\ \eta_t^* \end{pmatrix} \mid s_t \sim ID\left(\begin{pmatrix} 0 \\ 0 \end{pmatrix}, \begin{pmatrix} \sigma_{\ell}^2 & \gamma^* s_t \\ \gamma^* s_t & \sigma_{\eta}^2 - \mu^{*2} \end{pmatrix}\right).
\end{aligned} (3.4.3)$$

The Kalman filter may still be initialized by taking h_0 to have mean zero and variance $\sigma_0^2/(1-\phi^2)$.

The parameterization in (3.4.3) does not directly involve a parameter representing the correlation between e_t and η_t . The relationship between μ^* and γ^* and the original parameters in the model can only be obtained by making a distributional assumption about e_t as well as η_t . When e_t and η_t are bivariate normal with $Corr(e_t, \eta_t) = \rho$, $E(\eta_t|e_t) = \rho \sigma_{\eta} e_t$, and so

$$\mu^* = E_+(\eta_t) = \rho \sigma_\eta E_+(\varepsilon_t) = \rho \sigma_\eta \sqrt{2/\pi} = 0.7979 \rho \sigma_\eta.$$
 (3.4.4)

Furthermore,

$$\gamma^* = \rho \sigma_{\eta} E(|\varepsilon_t| \log \varepsilon_t^2) - 0.7979 \rho \sigma_{\eta} E(\log \varepsilon_t^2) = 1.1061 \rho \sigma_{\eta}. \tag{3.4.5}$$

When ϵ_t has a t-distribution, it can be written as $\zeta_t \kappa_t^{-0.5}$, and ζ_t and η_t can be regarded as having a bivariate normal distribution with correlation ρ , while κ_t is independent of both. To evaluate μ^* and γ^* one proceeds as before, except that the initial conditioning is on ζ_t rather than on ϵ_t , and the required expressions are found to be exactly as in the Gaussian case.

The filtered estimate of the log volatility h_t , written as $h_{t+1|t}$, takes the form:

$$h_{t+1|t} = \phi h_{t|t-1} + \frac{\phi(p_{t|t-1} + \gamma^* s_t)}{p_{t|t-1} + 2\gamma^* s_t + \sigma_t^2} (\log y_t^2 - \omega - h_{t|t-1}) + s_t \mu^*,$$

where p_{0k-1} is the corresponding mean square error of the h_{0k-1} . If $\rho < 0$, then $\gamma^* < 0$, and the filtered estimator will behave in a similar way to the EGARCH model estimated by Nelson (1991), with negative observations causing bigger increases in the estimated log volatility than corresponding positive values.

3.4.2. Nonlinear Filters

In principle, an exact filter may be written down for the original (3.1.2) and (3.1.3), with the former taken as the measurement equation. Evaluating such a filter requires approximating a series of integrals by numerical methods. Kitagawa (1987) has proposed a general method for implementing such a filter and Watanabe (1993) has applied it to the SV model. Unfortunately, it appears to be so time consuming as to render it impractical with current computer technology.

As part of their Bayesian treatment of the model as a whole, Jacquier, Polson and Rossi (1994) show how it is possible to obtain smoothed estimates of the volatilities by simulation. What is required is the mean vector of the joint distribution of the volatilities conditional on the observations. However, because simulating this joint distribution is not a practical proposition, they decompose it into a set of univariate distributions in which each volatility is conditional on all the others. These distributions may be denoted $p(\sigma_t|\sigma_{-t},y)$, where σ_{-t} denotes all the volatilities apart from σ_t . What one would like to do is to sample from each of these distributions in turn, with the elements of σ_{-t} set equal to their latest estimates, and repeat several thousand times. As such this is a Gibbs sampler. Unfortunately, there are difficulties. The Markov structure of the SV model may be exploited to write

$$p(\sigma_{t}|\sigma_{-t}, y) = p(\sigma_{t}|\sigma_{t-1}, \sigma_{t+1}, y_{t}) \propto p(y_{t}|h_{t}) p(h_{t}|h_{t-1}) p(h_{t+1}|h_{t})$$

but although the right hand side of the above expression can be written down explicitly, the density is not of a standard form and there is no analytic expression for the normalizing constant. The solution adopted by Jacquier, Polson and Rossi is to employ a series of Metropolis accept/reject independence chains.

Kim and Shephard (1994) argue that the single mover algorithm employed by Jacquier, Polson and Rossi will be slow if ϕ is close to one and/or σ_{η}^2 is small. This is because σ_t changes slowly; in fact when it is constant, the algorithm will not converge at all. Another approach based on the linear state space form, is to capture the non-normal disturbance term in the measurement equation, ξ_t , by a mixture of normals. Watanabe (1993) suggested an approximate method based on a mixture of two moments. Kim and Shephard (1994) propose a multimove sampler based on the linear state space form. Blocks of the $h_t's$ are sampled, rather than taking them one at a time. The technique they use is based on mixing an appropriate number of normal distributions to get the required level of accuracy in approximating the disturbance in (3.2.7). Mahieu and Schotman (1994a) extend this approach by introducing more degrees of freedom in the mixture of normals where the parameters are estimated rather than fixed a priori. Note that the distribution of the $\sigma_t's$ can be obtained from the simulated distribution of the $h_t's$.

Jacquier, Polson and Rossi (1994, p.416) argue that no matter how many mixture components are used in the Kim and Shephard method, the tail behavior of $\log \varepsilon_t^2$ can never be satisfactorily approximated. Indeed, they note that given the discreteness of the Kim and Shephard state space, not all states can have been visited enough in the small number of draws mentioned, i.e. the socalled inlier problem (see also section 3.4.1 and Nelson (1994)) is still present.

As a final point it should be noted that when the hyperparameters are unknown, the simulated distribution of the state produced by the Bayesian approach allows for their sampling variability.

3.4.3. ARCH Models as Approximate Filters

The purpose here is to draw attention to a subject that will be discussed in greater detail in section 4.3. In an ARCH model the conditional variance is assumed to be an exact function of past observations. As pointed out by Nelson and Foster (1994, p.32) this assumption is ad hoc on both economic and statistical grounds. However, because ARCH models are relatively easy to estimate, Nelson (1992) and Nelson and Foster (1994) have argued that a useful strategy is to regard them as filters which produce estimates of the conditional variance. Thus even if we believe we have a continuous time or discrete time SV model, we may decide to estimate a GARCH(1,1) model and treat the σ_t^{TS} as an approximate filter, as in (3.4.1). Thus the estimate is a weighted average of past squared observations. It delivers an estimate of the mean of the distribution of σ_t^T , conditional on the observations at time t-1. As an alternative, the model suggested by Taylor (1986) and Schwert (1989), in which the conditional standard deviation is set up as a linear combination of the previous conditional standard deviation and the previous absolute value, could be used. This may be more robust to outliers as it is a linear combination of past absolute values.

Nelson and Foster derive an ARCH model which will give the closest approximation to the continuous time SV formulation (see section 4.3 for more details). This does not correspond to one of the standard models, though it is fairly close to EGARCH. For discrete time SV models the filtering theory is not as extensively developed. Indeed, Nelson and Foster point out that a change from stochastic different equations to difference equations makes a considerable difference in the limit theorems and optimality theory. They study the case of near diffusions as an example to illustrate these differences.

3.5. Extensions of the Model

3.5.1. Persistence and Seasonality

The simplest nonstationary SV model has h_t following a random walk. The dynamic properties of this model are easily obtained if we work in terms of the logarithmically transformed observations, $\log y_t^2$. All we have to do is first difference to give a stationary process. The untransformed observations are nonstationary but the dynamic structure of the model will appear in the ACF of $|y_t/y_{t-1}|^c$, provided that c < 0.5.

The model is an alternative to IGARCH, that is (3.3.1) with $\alpha + \beta = 1$. The IGARCH model is such that the squared observations have some of the features of an integrated ARMA process and it is said to exhibit persistence; see Bollerslev and Engle (1993). However, its properties are not straightforward. For example it must contain a constant, γ , otherwise, as Nelson (1990) has shown, σ_t^2 converges almost surely to zero and the model has the peculiar feature of being strictly stationary but not weakly stationary. The nonstationary SV model, on the other hand, can be analyzed on the basis that h_t is a standard integrated of order one process.

Filtering and smoothing can be carried out within the linear state space framework, since $\log y_t^2$ is just a random walk plus noise. The initial conditions are handled in the same way as is normally done with nonstationary structural time series models, with a proper prior for the state being effectively formed from the first observation; see Harvey (1989). The optimal filtered estimate of h_t within the class of estimates which are linear in past $\log y_t^2$'s, that is $h_{t|t-1}$, is a constant plus an equally weighted moving average (EWMA) of past $\log y_t^2$'s. In IGARCH σ_t^2 is given exactly by a constant plus an EWMA of past squared observations.

The random walk volatility can be replaced by other nonstationary specifications. One possibility is the doubly integrated random walk in which $\Delta^2 h_i$ is white noise. When formulated in continuous time, this model is equivalent to a cubic spline and is known to give a relatively smooth trend when applied

in levels models. It is attractive in the SV context if the aim is to find a weighting function which fits a smoothly evolving variance. However, it may be less stable for prediction,

Other nonstationary components can easily be brought into h_t . For example, a seasonal or intra-daily component can be included; the specification is exactly as in the corresponding levels models discussed in Harvey (1989) and Harvey and Koopman (1993). Again the dynamic properties are given straightforwardly by the usual transformation applied to $\log y_t^2$, and it is not difficult to transform the absolute values suitably. Thus if the volatility consists of a random walk plus a slowly changing, nonstationary seasonal as in Harvey (1989, p. 40-3), the appropriate transformations are $\Delta_s \log y_t^2$ and $|y_t/y_{t-s}|^c$ where s is the number of seasons. The state space formulation follows along the lines of the corresponding structural time series models for levels. Handling such effects is not so easy within the GARCH framework.

Different approaches to seasonality can also be incorporated in SV models using ideas of time deformation as discussed in a later sub-section. Such approaches may be particularly relevant when dealing with the kind of abrupt changes in seasonality which seem to occur in high frequency, like five minute or tick-by-tick, foreign exchange data.

3.5.2. Interventions and other deterministic effects

Intervention variables are easily incorporated into SV models. For example, a sudden structural change in the volatility process can be captured by assuming that

$$\log \sigma_i^2 = \log \sigma^2 + h_i + \lambda w_i$$

where w_t is zero before the break and one after and λ is an unknown parameter. The logarithmic transformation gives (3.2.8) but with λw_t added to the right hand side. Care needs to be taken when incorporating such effects into ARCH models. For example, in the GARCH(1,1) a sudden break has to be modelled as

$$\sigma_t^2 = \gamma + \lambda w_t - (\alpha + \beta)\lambda w_{t-1} + \alpha y_{t-1}^2 + \beta \sigma_{t-1}^2$$

with λ constrained so that σ_i^2 is always positive.

More generally observable explanatory variables, as opposed to intervention dummies, may enter into the model for the variance.

3.5.3. Multivariate Models

The multivariate model corresponding to (3.1.2) assumes that each series is generated by a model of the form

$$y_{it} = \sigma_i \varepsilon_{it} e^{0.5h_{it}}, t = 1, ..., T,$$
 (3.5.1)

with the covariance (correlation) matrix of the vector $\boldsymbol{\varepsilon}_{t} = (\varepsilon_{1t}, ..., \varepsilon_{Nt})'$ being denoted by $\boldsymbol{\Sigma}_{t}$. The vector of volatilities, \boldsymbol{h}_{t} , follows a VAR(1) process, that is

$$h_{t+1} = \Phi h_t + \eta_t$$

where $\eta_t \sim IID(0, \Sigma_\eta)$. This specification allows the movements in volatility to be correlated across different series via Σ_η . Interactions can be picked up by the off-diagonal elements of Φ .

The logarithmic transformation of squared observations leads to a multivariate linear state space model from which estimates of the volatilities can be computed as in section 3.4.1.

A simple nonstationary model is obtained by assuming that the volatilities follow a multivariate random walk, that is $\Phi = 1$. If Σ_n is singular, of rank K<N, there are only K components in volatility, that is each h_K in (3.5.1) is a linear combination of K < N common trends, that is

$$\mathbf{h}_{i} = \mathbf{\Theta} \mathbf{h}_{i}^{\dagger} + \overline{\mathbf{h}} \tag{3.5.2}$$

where \mathbf{h}_{i}^{1} is the $K \times 1$ vector of common random walk volatilities, \overline{h} is a vector of constants and Θ is an $N \times K$ matrix of factor loadings. Certain restrictions are needed on Θ and \overline{h} to ensure identifiability; see Harvey, Ruiz and Shephard (1994). The logarithms of the squared observations are 'co-integrated' in the sense of Engle and Granger (1987) since there are N-K linear combinations of them which are white noise and hence stationary. This implies, for example, that if two series of returns exhibit stochastic volatility, but this volatility is the same with $\Theta' = (1,1)$, then the ratio of the series will have no stochastic volatility. The application of the related concept of 'co-persistence' can be found in Bollerslev and Engle (1993). However, as in the univariate case there is some ambiguity about what actually constitutes persistence.

There is no reason why the idea of common components in volatility should not extend to stationary models. The formulation of (3.5.2) would apply, without the need for \overline{h} , and with h_{ℓ}^{\dagger} modelled, for example, by a VAR(1).

Bollerslev, Engle and Wooldridge (1988) show that a multivariate GARCH model can, in principle, be estimated by maximum likelihood, but because of the large number of parameters involved computational problems are often encountered unless restrictions are made. The multivariate SV model is much simpler than the general formulation of a multivariate GARCH. However, it is limited in that it does not model changing covariances. In this sense it is analogous to the restricted multivariate GARCH model of Bollerslev (1986) in which the conditional correlations are assumed to be constant.

Harvey, Ruiz and Shephard (1994) apply the nonstationary model to four exchange rates and find just two common factors driving volatility. Another application is in Mahieu and Schotman (1994b). A completely different way of modelling exchange rate volatility is to be found in the latent factor ARCH model of Diebold and Nerlove (1989).

3.5.4. Observation intervals, aggregation and time deformation

Suppose that an SV model is observed every δ time periods. In this case, h_{τ} , where τ denotes the new observation (sampling) interval, is still AR(1) but with parameter ϕ^{δ} . The variance of the disturbance, η_{ϵ} , increases, but σ_h^2 remains the same. This property of the SV model makes it easy to make comparisons across different sampling intervals; for example it makes it clear why if ϕ is around 0.98 for daily observations, a value of around 0.9 can be expected if an observation is made every week (assuming a week has 5 days).

If averages of observations are observed over the longer period, the comparison is more complicated, as h_{τ} will now follow an ARMA(1,1) process. However, the AR parameter is still ϕ^{δ} . Note that it is difficult to change the observation interval of ARCH processes unless the structure is weakened as in Drost and Nijman (1993); see also section 4.4.1.

Since, as noted in section 2.4, one typically uses a discrete time approximation to the continuous time model, it is quite straightforward to handle irregularly spaced observations by using the linear state space form as described, for example, in Harvey (1989). Indeed the approach originally proposed by Clark (1973) based on subordinated processes to describe asset prices and their volatility fits quite well into this

framework. The techniques for handling irregularly spaced observations can be used as the basis for dealing with time deformed observations, as noted by Stock (1988). Ghysels and Jasiak (1994a, b) suggest a SV model in which the operational time for the continuous time volatility equation is determined by the flow of information. Such time deformed processes may be particularly suited to dealing with high frequency data. If $\tau = q(t)$ is the mapping between calendar time τ and operational time t, then

$$dS_t = \mu S_t dt + \sigma \left(g(t) \right) S_t dW_{1t}$$

and

$$d\log\sigma(\tau) = a\left((b - \log\sigma(\tau)\right)d\tau + cdW_{2\tau}$$

where W_{11} and W_{27} are standard, independent Wiener processes. The discrete time approximation generalizing (3.1.3), but including a term which in (3.1.2) is incorporated in the constant scale factor σ , is then

$$h_{t+1} = [1 - e^{-a\Delta g(t)}]b + e^{-a\Delta g(t)}h_t + \eta_t$$

where $\Delta g(t)$ is the change in operational time between two consecutive calendar time observations and η_t is normally distributed with mean zero and variance $c^2(1-e^{-2\alpha\Delta g(t)})/2a$. Clearly if $\Delta g(t)=1$, $\phi=e^{-a}$ in (3.1.3). Since the flow of information, and hence $\Delta g(t)$, is not directly observable, a mapping to calendar time must be specified to make the model operational. Ghysels and Jasiak (1994a) discuss several specifications revolving around a scaled exponential function relating g(t) to observables such as past volume of trade and past price changes with asymmetric leverage effects. This approach was also used by Ghysels and Jasiak (1994b) to model return-volume co-movements and by Ghysels, Gouriéroux and Jasiak (1995b) for modeling intra-daily high frequency data which exhibit strong seasonal patterns (cfr. section 3.5.1).

3.5.5. Long Memory

Baillie, Bollerslev and Mikkelsen (1993) propose a way of extending the GARCH class to account for long memory. They call their models Fractionally Integrated GARCH (FIGARCH), and the key feature is the inclusion of the fractional difference operator, $(1-L)^d$, where L is the lag operator, in the lag structure of past squared observations in the conditional variance equation. However, this model can only be stationary when d=0 and it reduces to GARCH. In a later paper, Bollerslev and Mikkelsen (1995) consider a generalization of the EGARCH model of Nelson (1991) in which $\log \sigma_i^2$ is modelled as a distributed lag of past $\varepsilon_i s$ involving the fractional difference operator. This FIEGARCH model is stationary and invertible if |d| < 0.5.

Breidt, Crato and de Lima (1993) and Harvey (1993) propose an SV model with h_{ℓ} generated by fractional noise

$$h_t = \eta_t/(1-L)^d, \eta_t \sim NID(0, \sigma_n^2), 0 \le d \le 1.$$
 (3.5.1)

Like the AR(1) model in (3.1.3), this process reduces to white noise and a random walk at the boundary of the parameter space, that is d=0 and 1 respectively. However, it is only stationary if d<0.5. Thus the transition from stationarity to nonstationarity proceeds in a different way to the AR(1) model. As in the AR(1) case it is reasonable to constrain the autocorrelations in (3.5.1) to be positive. However, a

negative value of d is quite legitimate and indeed differencing h_i when it is nonstationary gives a stationary intermediate memory' process in which $-0.5 \le d \le 0$.

The properties of the long memory SV model can be obtained from the formulae in sub-section 3.2. A comparison of the ACF for h_t following a long memory process with d=0.45 and $\sigma_h^2=2$ with the corresponding ACF when h_t is AR(1) with $\phi=0.99$ can be found in Harvey (1993). Recall that a characteristic property of long memory is a hyperbolic rate of decay for the autocorrelations instead of an exponential rate, a feature observed in the data (see section 2.2(e)). The slower decline in the long memory model is very clear and, in fact, for $\tau=1000$, the long memory autocorrelation is still 0.14, whereas in the AR case it is only 0.000013. The long memory shape closely matches that in Ding, Granger and Engle (1993, p. 86-8).

The model may be extended by letting η_t be an ARMA process and/or by adding more components to the volatility equation.

As regards smoothing and filtering, it has already been noted that the state space approach is approximate because of the truncation involved and is relatively cumbersome because of the length of the state vector. Exact smoothing and filtering, which is optimal within the class of estimators linear in the $\log y_t^{\mathcal{X}}s$, can be carried out by a direct approach if one is prepared to construct and invert the $T \times T$ covariance matrix of the $\log y_t^{\mathcal{X}}s$.

4. Continuous Time Models

At the end of section 2 we presented a framework for statistical modelling of SV in discrete time and devoted the entire section 3 to specific discrete time SV models. To motivate the continuous time models we study first of all the exact relationship (i.e. without approximation error) between differential equations and SV models in discrete time. We examine this relationship in section 4.1 via a class of statistical models which are closed under temporal aggregation and proceed (1) from high frequency discrete time to lower frequencies and (2) from continuous time to discrete time. Next, in section 4.2, we study option pricing and hedging with continuous time models and elaborate on features such as the smile effect. The practical implementation of option pricing formulae with SV often requires discrete time SV and/or ARCH models as filters and forecasters of the continuous time volatility processes. Such filters, covered in section 4.3, are in general discrete time approximations (and not exact discretizations as in section 4.1) of continuous time SV models.

4.1. From discrete to continuous time

The purpose of this section is to provide a rigorous discussion of the relationship between discrete and continuous time SV models. The presentation will proceed first with a discussion of temporal aggregation in the context of the SARV class of models and focus on specific cases including GARCH models. This material is covered in section 4.1.1. Next we turn our attention to the aggregation of continuous time SV models to yield discrete time representations. This is the subject matter of section 4.1.2.

4.1.1. Temporal Aggregation of Discrete Time Models

Andersen's SARV class of models was presented in section 2.4 as a general discrete time parametric SV statistical model. Let us consider the zero-mean case, namely:

$$y_{t+1} = \sigma_t \varepsilon_{t+1} \tag{4.1.1}$$

and σ_t^q for q=1 or 2 is a polynomial function $g(K_t)$ of the Markov process K_t with stationary autoregressive representation :

$$K_t = \omega + \beta K_{t-1} + \upsilon_t \tag{4.1.2}$$

where $|\beta| < 1$ and

$$E\left[\varepsilon_{t+1}\middle|\varepsilon_{\tau},\upsilon_{\tau}\;\tau\leq t\right]=0\tag{4.1.3a}$$

$$\mathbb{E}\left[\varepsilon_{t+1}^{2}\middle|\varepsilon_{\tau}, \upsilon_{\tau}\,\tau \leq t\right] = 1 \tag{4.1.3b}$$

$$\mathbf{E}\left[v_{t+1}|\,\varepsilon_{\tau},v_{\tau}\,\tau\leq t\right]=0\tag{4.1.3c}$$

The restrictions (4.1.3a-c) imply that v is a martingale difference sequence with respect to the filtration $\mathcal{F}_t = \sigma\left[\varepsilon_\tau, v_\tau, \tau \leq t\right].^{22}$ Moreover, the conditional moment conditions in (4.3.1a-c) also imply that ε in (4.1.1) is a white noise process in a semi-strong sense, i.e. $\mathrm{E}[\varepsilon_{t+1}|\varepsilon_\tau, \tau \leq t] = 0$ and $\mathrm{E}\left[\varepsilon_{t+1}^2|\varepsilon_\tau, \tau \leq t\right] = 1$, and is not Granger caused by $v.^{23}$ From the very beginning of section 2 we choose the continuously compounded rate of return over a particular time horizon as the starting point for continuous time processes. Therefore, let y_{t+1} in (4.1.1) be the continuously compounded rate of return for [t,t+1] of the asset price process S_t , consequently:

$$y_{t+1} = \log S_{t+1} / S_t \tag{4.1.4}$$

Since the unit of time of the sampling interval is to a large extend arbitrary, we would surely want the SV model defined by equations (4.1.1) through (4.1.3), (for given q and function g) to be closed under temporal aggregation. As rates of return are flow variables, closedness under temporal aggregation means that for any integer m:

$$y_{tm}^{(m)} \equiv \log S_{tm} / S_{tm-m} = \sum_{k=0}^{m-1} y_{tm-k}$$

is again conformable to a model of the type (4.1.1) through (4.1.3) for the same choice of q and g involving suitably adapted parameter values. The analysis in this section follows Meddahi and Renault (1995) who study temporal aggregation of SV models in detail, particularly the cases (1) $\sigma_t^2 = K_t$, i.e. q = 2 and g is the identity function and (2) $\sigma_t^2 = \exp(K_t)$ which is the leading discrete time SV model. We will focus here on the former as it is related to the socalled continuous time GARCH approach of Drost and Werker (1994). Hence, we have (4.1.1) with:

$$\sigma_i^2 = \omega + \beta \sigma_{i-1}^2 + \upsilon_{i}. \tag{4.1.5}$$

²²Note that we do not use here the decomposition appearing in (2.4.9) namely, $v_t = [\gamma + aK_{t-1}]\hat{u}_t$.

²³The Granger poscausality considered here for e_t is weaker than Assumption 2.3.2.A as it applies only to the first two conditional moments.

With conditional moment restrictions (4.1.3a-c) this model is closed under aggregation. For instance, for m=2:

$$y_{t+1}^{(2)} = y_{t+1} + y_t = \sigma_{t-1}^{(2)} \varepsilon_{t+1}^{(2)}$$

with :

$$\left(\sigma_{t-1}^{(2)}\right)^2 = w^{(2)} + \beta^{(2)} \left(\sigma_{t-3}^{(2)}\right)^2 + \upsilon_{t-1}^{(2)}$$

where

$$\begin{cases} w^{(2)} &= 2\omega (1+\beta) \\ \beta^{(2)} &= \beta^2 \\ v^{(2)}_{t-1} &= (\beta+1) [\beta v_{t-2} + v_{t-1}]. \end{cases}$$

Moreover, it also worth noting that whenever a leverage effect is present at the aggregate level, i.e. :

$$Cov\left[v_{t-1}^{(2)},\varepsilon_{t-1}^{(2)}\right]\neq 0$$

with $\varepsilon_{t-1}^{(2)} = \left(y_{t-1} + y_{t-2}\right) / \sigma_{t-3}^{(2)}$, it necessarily appears at the disaggregate level, i.e. $Cov\left(v_{t}, \varepsilon_{t}\right) \neq 0$.

For the general case Meddahi and Renault (1995) show that model (4.1.5) together with conditional moment restrictions (4.1.3a-c) is a class of processes closed under aggregation. Given this result, it is of interest to draw a comparison with the work of Drost and Nijman (1993) on temporal aggregation of GARCH. While establishing this link between Meddahi and Renault (1995) and Drost and Nijman (1993) we will also uncover issues of leverage properties in GARCH models. Indeed, contrary to what is often believed, we will find leverage effect restrictions in socalled weak GARCH processes defined below. Moreover, we will also find from the results of Meddahi and Renault that the class of weak GARCH processes includes certain SV models.

To find a class of GARCH processes which is closed under aggregation Drost and Nijman (1993) weakened the definition of GARCH, namely for a positive stationary process σ_t :

$$\sigma_t^2 = w + ay_{t-1}^2 + b\sigma_{t-1}^2 \tag{4.1.6}$$

where a + b < 1, they defined:

- strong GARCH if y_{t+1}/σ_t is i.i.d. with mean zero and variance 1
- semi-strong GARCH if $E[y_{t+1}|y_{\tau}, \tau \leq t] = 0$ and $E[y_{t+1}^2|y_{\tau}, \tau \leq t] = \sigma_t^2$
- weak GARCH if $\mathrm{EL}[y_{t+1}|\ y_{\tau},y_{\tau}^2,\tau\leq t]=0$; $\mathrm{EL}\left[y_{t+1}^2|\ y_{\tau},y_{\tau}^2,\tau\leq t\right]=\sigma_t^2.^{24}$

Drost and Nijman show that weak GARCH processes temporally aggregate and provide explicit formula for their coefficients. In section 2.4 it was noted that the framework of SARV includes GARCH processes whenever there is no randomness specific to the volatility process. This property will allow us to show that the class of weak GARCH processes -as defined above- in fact includes more general SV processes which are strictly speaking not GARCH. The arguments, following Meddahi and Renault (1995), require a

²⁴For any Hilbert space H of L², $EL[x_1|x, x \in H]$ is the best linear predictor of x_i in terms of 1 and $x \in H$. It should be noted that a strong GARCH process is a fortiori semi-strong which itself is also a weak GARCH process.

classification of the models defined by (4.1.3) and (4.1.5) according to the value of the correlation between v_t and y_t^2 , namely :

(a) Models with perfect correlation: This first class, henceforth denoted C_1 , is characterized by a linear correlation between v_t and y_t^2 conditional on $(\varepsilon_\tau, v_\tau, \tau < t)$ which is either 1 or -1 for the model in (4.1.5).

(b) Models without perfect correlation: This second class, henceforth denoted C_2 has the above conditional correlation less than one in absolute value.

The class C_1 contains all semi-strong GARCH processes, indeed whenever $Var\left[y_i^2|\varepsilon_i, \upsilon_\tau, \tau < t\right]$ is proportional to $Var\left[\upsilon_i|\varepsilon_\tau, \upsilon_\tau, \tau < t\right]$ in C_1 we have a semi-strong GARCH. Consequently, a semi-strong GARCH processes is a model (4.1.5) with (1) restrictions (4.1.3), (2) a perfect conditional correlation as in C_1 , and (3) restrictions on the conditional kurtosis dynamics.

Let us consider now the following assumption :

Assumption 4.1.1: The following two conditional expectations are zero:

$$E\left[\varepsilon_{t}\upsilon_{t}\left|\varepsilon_{\tau},\upsilon_{\tau},\tau< t\right.\right]=0\tag{4.1.7a}$$

$$E\left[\varepsilon_{t}^{3}\left|\varepsilon_{\tau}, v_{\tau}, \tau < t\right.\right] = 0. \tag{4.1.7b}$$

This assumption amounts to an absence of leverage effects, where the latter is defined in a conditional covariance sense to capture the notion of instantaneous causality discussed in section 2.4.1 and applied here in the context of weak white noise. 26 It should also parenthetically be noted that (4.1.7a) and (4.1.7b) are in general not equivalent except for the processes of class C_1 .

The class C_2 allows for randomness proper to the volatility process due to the imperfect correlation. Yet, despite this volatility-specific randomness one can show that under Assumption 4.1.1 processes of C_2 satisfy the weak GARCH definition. A fortiori, any SV model conformable to (4.1.3a-c), (4.1.5), (4.7.1a-b) and Assumption 4.1.1 is a weak GARCH process. It is indeed the symmetry assumption (4.1.7a-b), or restrictions on leverage in GARCH, that makes that $\mathrm{EL}[y_{t+1}^2|y_\tau,y_\tau^2,\tau\leq t]=\sigma_t^2$ (together with the conditional moment restrictions (4.1.3a-c)) and yields the internal consistency for temporal aggregation found by Drost and Nijman (1993, example 2, p.915) for the class of socalled symmetric weak GARCH(1,1). Hence, this class of weak GARCH(1,1) processes can be viewed as a subclass of processes satisfying (4.1.3) and (4.1.5).

4.1.2. Temporal aggregation of continuous time models

To facilitate our discussion we will specialize the general continuous time model (2.3.1) to processes with zero drift, i.e. :

$$d\log S_t = \sigma_t dW_t \tag{4.1.8a}$$

²⁵In fact, Nelson and Foster (1994) observed that the most commonly used ARCH models effectively assume that the variance of the variance rises linearly in σ_t^4 , which is the main drawback of ARCH models to approximate SV models in continuous time (see also section 4.3).

²⁶The conditional expectation (4.1.7b) can be viewed as a conditional covariance between ε_t and ε_t^2 . It is this conditional covariance which, if nonzero, produces leverage effects in GARCH.

²⁷As noted before, the class of processes satisfying (4.1.3) and (4.1.5) is closed under temporal aggregation, including processes with leverage effects not satisfying Assumption 4.1.1.

$$d\sigma_t = \gamma_t dt + \delta_t dW_t^{\sigma} \tag{4.1.8b}$$

$$Cov\left(dW_{t}, dW_{t}^{\sigma}\right) = \rho_{t}dt \tag{4.1.8c}$$

where the stochastic processes σ_t , γ_t , δ_t and ρ_t are $I_t^a = [\sigma_\tau; \tau \leq t]$ adapted. To ensure that σ_t is a nonnegative process one typically follows either one of two strategies: (1) considering a diffusion for $\log \sigma_t^2$ or (2) describing σ_t^2 as a CEV process (or Constant Elasticity of Variance process following Cox (1975) and Cox and Ross (1976)).²⁸ The former is frequently encountered in the option pricing literature (see e.g. Wiggins (1987)) and is also clearly related to Nelson (1991), who introduced EGARCH, and to the log-Normal SV model of Taylor (1986). The second class of CEV processes can be written as

$$d\sigma_t^2 = k \left(\theta - \sigma_t^2\right) dt + \gamma \left(\sigma_t^2\right)^{\delta} dW_t^{\sigma} \tag{4.1.9}$$

where $\delta \geq \frac{1}{2}$ ensures that σ_t^2 is a stationary process with nonnegative values. Equation (4.1.9) can be viewed as the continuous time analogue of the discrete time SARV class of models presented in section 2.4. This observation establishes links with the discussion of the previous section 4.1.1 and yields exact discretization results of continuous time SV models. Here, as in the previous section it will be tempting to draw comparisons with the GARCH class of models, in particular the diffusions proposed by Drost and Werker (1994) in line with the temporal aggregation of weak GARCH processes.

Firstly, one should note that the CEV process in (4.1.9) implies an autoregressive model in discrete time for σ_t^2 , namely :

$$\sigma_{t+\Delta t}^{2} = \theta \left(1 - e^{-k\Delta t}\right) + e^{-k\Delta t}\sigma_{t}^{2} + e^{-k\Delta t} \int_{t}^{t+\Delta t} e^{k(u-t)} \gamma \left(\sigma_{u}^{2}\right)^{\delta} dW_{u}^{\sigma}$$
(4.1.10)

Meddahi and Renault (1995) show that whenever (4.1.9) and its discretization (4.1.10) govern volatility then the discrete time process $\log S_{t+(k+1)\Delta t}/S_{t+k\Delta t}, k\in\mathbb{Z}$ is a SV process satisfying the model restrictions (4.1.3a-c) and (4.1.5). Hence, from the diffusion (4.1.9) we obtain the class of discrete time SV models which is closed under temporal aggregation, as discussed in the previous section. To be more specific, consider for instance $\Delta t = 1$, then from (4.1.10) it follows that:

$$y_{t+1} = \log S_{t+1} / S_t = \sigma_i \varepsilon_{t+1}$$

 $\sigma_t^2 = w + \beta \sigma_{t-1}^2 + v_t$ (4.1.11)

where from (4.1.10):

$$\begin{cases} \beta = e^{-k}, w = \theta \left(1 - e^{-k} \right), \\ v_{t+1} = e^{-k} \int_{1}^{t+1} e^{k(u-t)} \gamma \left(\sigma_{u}^{2} \right)^{\delta} W_{u}^{\sigma}. \end{cases}$$
(4.1.12)

²⁶Occasionally one encounters specifications which do not ensure nonnegativity of the σ_t process. For the sake of computational simplicity some authors for instance have considered Ornstein-Uhlenbeck processes for σ_t or σ_t^2 (see e.g. Stein and Stein (1991)).

It is important to note from (4.1.12) that absence of leverage effect in continuous time, i.e. $\rho_t = 0$ in (4.1.8c), means no such effect at low frequencies and the two symmetry conditions of Assumption 4.1.1 are fulfilled. This line of reasoning also explains the temporal aggregation result of Drost and Werker (1994), but one more generally can interpret discrete time SV models with leverage effects as exact discretizations of continuous time SV models with leverage.

4.2. Option pricing and hedging

Section 4.2.1 is devoted to the basic option pricing model with SV, namely the Hull and White model of section 2. We are better equipped now to elaborate on its theoretical foundations. The practical implications appear in section 4.2.2 while 4.2.3 concludes with some extensions of the basic model.

4.2.1. The Basic Option Pricing Formula

Consider again formula (2.1.10) for a European option contract maturing at time t + h = T. As noted in section 2.1.2, we assume continuous and frictionless trading. Moreover no arbitrage profits can be made from trading in the underlying asset and riskless bonds; interest rates are nonstochastic so that B(t,T) defined by (2.1.12) denotes the time t price of a unit discount bond maturing at time T. Consider now the probability space (Ω, \mathcal{F}, P) , which is the fundamental space of the underlying asset price process S:

$$dS_t/S_t = \mu(t, S_t, U_t) dt + \sigma_t dW_t^s$$

$$\sigma_t^2 = f(U_t)$$

$$dU_t = a(t, U_t) dt + b(t, U_t) dW_t^\sigma$$
(4.2.1)

where $W_t = \left(W_t^S, W_t^\sigma\right)$, is a standard two dimensional Brownian Motion $\left(W_t^B \text{ and } W_t^\sigma\right)$ are independent, zero-mean and unit variance) defined on (Ω, \mathcal{F}, P) . The function f, called the volatility function, is assumed to be one-to-one. In this framework (under suitable regularity conditions) the no free lunch assumption is equivalent to the existence of a probability distribution Q on (Ω, \mathcal{F}) , equivalent to P, under which discounted price processes are martingales (see Harrison and Kreps (1979)). Such a probability is called an equivalent martingale measure and is unique if and only if the markets are complete (see Harrison and Pliska (1981)). From the integral form of martingale representations (see Karatzas and Shreve (1988), p. 184), the (positive) density process of any probability measure Q equivalent to P can be written as:

$$M_{t} = \exp \left[-\int_{0}^{t} \lambda_{u}^{S} dW_{u}^{S} - \frac{1}{2} \int_{0}^{4} \left(\lambda_{u}^{S} \right)^{2} du - \int_{0}^{4} \lambda_{u}^{\sigma} dW_{u}^{\sigma} - \frac{1}{2} \int_{0}^{4} \left(\lambda_{u}^{\sigma} \right)^{2} du \right]$$
(4.2.2)

where the processes λ^g and λ^σ are adapted to the natural filtration $\sigma_t = \sigma[W_\tau, \tau \leq t], t \geq 0$, and satisfy the integrability conditions (almost surely):

$$\int_0^t \left(\lambda_u^S\right)^2 du < + \infty \text{ and } \int_0^t \left(\lambda_u^\sigma\right)^2 du < + \infty$$

By Girsanov's theorem the process $\tilde{W}=\left(\tilde{W}^S,\tilde{W}''\right)'$ defined by :

$$\hat{W}_t^S = \hat{W}_t^S + \int_0^t \lambda_u^S du \text{ and } \hat{W}_t^\sigma = W_t^\sigma + \int_0^t \lambda_u^\sigma du \qquad (4.2.3)$$

²⁶Here, the market is seen as incomplete (before taking into account the market pricing of the option) so that we have to characterize a set of equivalent martingale measures.

is a two dimensional Brownian Motion under Q. The dynamic of the underlying asset price under Q is obtained directly from (4.2.1) and (4.2.3). Moreover, the discounted asset price process $S_t B(0,t)$, $0 \le t \le T$, is a Q-martingale if and only if for r_t defined in (2.1.11):

$$\lambda_t^S = \frac{\mu(t, S_t, U_t) - r_t}{\sigma_t} \tag{4.2.4}$$

Since S is the ordy traded asset, the process λ^{σ} is not fixed. The process λ^{σ} defined by (4.2.4) is called the asset risk premium. By analogy, any process λ^{σ} satisfying the required integrality condition can be viewed as a volatility risk premium and for any choice of λ^{σ} , the probability $Q(\lambda^{\sigma})$ defined by the density process M in (4.2.2) is an equivalent martingale measure. Therefore, given the volatility risk premium process λ^{σ} :

$$C_t^{\lambda^{\sigma}} = B(t, T) E_t^{Q(\lambda^{\sigma})} [Max[0, S_T - K]], 0 \le t \le T$$
(4.2.5)

is an admissible price process of the European call option.³⁰

The Hull and White option pricing model relies on the following assumption, which restricts the set of equivalent martingale measures:

Assumption 4.2.1: The volatility risk premium λ_i^{σ} only depends on the current value of the volatility process: $\lambda_i^{\sigma} = \lambda^{\sigma}(t, U_t)$, $\forall t \in [0, T]$.

This assumption is consistent with an intertemporal equilibrium model where the agent preferences are described by time separable isoclastic utility functions (see He (1993)) and Pham and Touzi (1993)). It ensures that \tilde{W}^S and \tilde{W}^σ are independent, so that the $Q(\lambda^\sigma)$ distribution of $\log S_T/S_t$, conditionally on F_t and the volatility path $(\sigma_t, 0 \le t \le T)$ is normal with mean $\int_t^T r_u du - \frac{1}{2} \gamma^2(t, T)$ and variance $\gamma^2(t, T) = \int_t^T \sigma_u^2 du$. Under Assumption 4.2.1 one can compute the expectation in (4.2.5) conditionally on the volatility path, and we obtain finally:

$$C_t^{\lambda''} = S_t E_t^{Q(\lambda'')} \left[\phi(d_{1t}) - e^{-x_t} \phi(d_{2t}) \right]$$
 (4.2.6)

with the same notation as in (2.1.20). To conclude it is worth noting that many option pricing formulae available in the literature have a feature common with (4.2.6) as they can be expressed as an expectation of the Black-Scholes price over an heterogeneity distribution of the volatility parameter (see Renault (1995) for an elaborate discussion on this subject).

4.2.2. Pricing and Hedging with the Hull and White model

The Markov feature of the process (S,σ) implies that the option price (4.2.6) only depends on the contemporaneous values of the underlying asset prices and its volatility. Moreover, under mild regularity conditions, this function is differentiable. Therefore, a natural way to solve the hedging problem in this stochastic volatility context is to hedge a given option of price C_t^1 by Δ_t^* units of the underlying asset and Σ_t^* units of any other option of price C_t^2 where the hedging ratios solve:

$$\begin{cases} \partial C_t^1/\partial S_t - \Delta_t^* - \sum_t^* \partial C_t^2/\partial S_t = 0\\ \partial C_t^1/\partial \sigma_t - \sum_t^* \partial C_t^2/\partial \sigma_t = 0 \end{cases}$$
(4.2.7)

²⁰Here elsewhere $E_t^Q(\cdot) = E^Q(\cdot|\mathcal{F}_t)$ stands for the conditional expectation operator given \mathcal{F}_t when the price dynamics are governed by Q.

Such a procedure, known as the delta-sigma hedging strategy, has been studied by Scott (1991). By showing that any European option completes the market, i.e. $\partial C_t^2/\partial \sigma_t \neq 0$, $0 \leq t \leq T$, Bajeux and Rochet (1992) justify the existence of an unique solution to the delta-sigma hedging problem (4.2.7) and the implicit assumption in the previous sections that the available information I_t contains the past values (S_t, σ_t) , $\tau \leq t$. In practice, option traders often focus on the risk due to the underlying asset price variations and consider the imperfect hedging strategy $\sum_t = 0$ and $\Delta_t = \partial C_t^1/\partial S_t$. Then, the Hull and White option pricing formula (4.2.6) provides directly the theoretical value of Δ_t :

$$\Delta_t = \partial C_t^{\Lambda^{\sigma}} / \partial S_t = E_t^{Q(\Lambda^{\sigma})} \phi(d_{1t}) \tag{4.2.8}$$

This theoretical value is hard to use in practice since: (1) even if we knew the $Q(\lambda^{\sigma})$ conditional probability distribution of d_{1t} given I_t (summarized by σ_t), the derivation of the expectation (4.2.8) might be computationally demanding and (2) the conditional probability is directly related to the conditional probability distribution of $\gamma^2(t,T) = \int_t^T \sigma_{\sigma}^2 du$ given σ_t , which in turn may involve nontrivially the parameters of the latent process σ_t . Moreover, these parameters are those of the conditional probability distribution of $\gamma^2(t,T)$ given σ_t under the risk-neutral probability $Q(\lambda^{\sigma})$ which is generally different from the Data Generating Process P. The statistical inference issues are therefore quite complicated. We will argue in section 5 that only tools like simulation-based inference methods involving both asset and option prices (via an option pricing model) may provide some satisfactory solutions.

Nevertheless, a practical way to avoid these complications is to use the Black-Scholes option pricing model, even though it is known to be misspecified. Indeed, option traders know that they cannot generally obtain sufficiently accurate option prices and hedge ratios by using the BS formula with historical estimates of the volatility parameters based on time series of the underlying asset price. However, the concept of Black-Scholes implied volatility (2.1.23) is known to improve the pricing and hedging properties of the BS model. This raises two issues: (1) what is the internal consistency of the simultaneous use of the BS model (which assumes constant volatility) and of BS implied volatility which is clearly time-varying and stochastic and (2) how to exploit the panel structure of option pricing errors?³¹

Concerning the first issue, we noted in section 2 that the Hull and White option pricing model can indeed be seen as a theoretical foundation for this practice of pricing. Hedging issues and the panel structure of option pricing errors are studied in detail in Renault and Touzi (1992) and Renault (1995).

4.2.3. Smile or Smirk?

As noted in section 2.2, the smile effect is now a well documented empirical stylized fact. Moreover the smile becomes sometimes a smirk since it appears more or less lopsided (the socialled skewness effect). We cautioned in section 2 that some explanations of the smile/smirk effect are often founded on tempting analogies rather than rigorous proofs.

To the best of our knowledge, the state of the art is the following: (i) the first formal proof that a Hull and White option pricing formula implies a symmetric smile was provided by Renault and Touzi (1992), (ii) the first complete proof that the smile/smirk effects can alternatively be explained by liquidity problems the upper parts of the smile curve, i.e. the most expensive options are the least liquid) was provided by Platten and Schweizer (1994) using a microstructure model, (iii) there is no formal proof that asymmetries of the probability distribution of the underlying asset price process (leverage effect, non-normality,...) are

³¹The value of σ which equates the BS formula to the observed market price of the option heavily depends on the actual date t, the strike price K, the time to maturity (T-t) and therefore creates a panel data structure.

able to capture the observed skewness of the smile. A different attempt to explain the observed skewness is provided by Renault (1995). He showed that a slight discrepancy between the underlying asset price \tilde{S}_t used to infer BS implied volatilities and the stock price S_t considered by option traders may generate an empirically plausible skewness in the smile. Such nonsynchronous \tilde{S}_t and S_t may be related to various issues: bid-ask spreads, non-synchronous trading between the two markets, forecasting strategies based on the leverage effect, etc.

Finally, to conclude it is also worth noting that a new approach initiated by Gouriéroux, Monfort, Tenreiro (1994) and followed also by Ait-Sahalia, Bickel, Stoker (1994) is to explain the BS implied volatility using a nonparametric function of some observed state variables. Gouriéroux, Monfort, Tenreiro (1995) obtain for example a good nonparametric fit of the following form:

$$\sigma_t(S_t, K) = \mathbf{a}(K) + \mathbf{b}(K) (\log S_t / S_{t-1})^2.$$

A classical smile effect is directly observed on the intercept $\mathbf{a}(K)$ but an inverse smile effect appears for the path-dependent effect parameter $\mathbf{b}(K)$. For American options a different nonparametric approach is pursued by Broadie, Detemple, Ghysels and Torrès (1995) where besides volatility also exercise boundaries for the option contracts are nonparametrically obtained.³²

4.3. Filtering and Discrete Time Approximations

In section 3.4.3 it was noted that the ARCH class of models could be viewed as filters to extract the (continuous time) conditional variance process from discrete time data. Several papers were devoted to the subject, namely Nelson (1990, 1992, 1995a,b) and Nelson and Foster (1994, 1995). It was one of Nelson's seminal contributions to bring together ARCH and continuous time SV. Nelson's first contribution in his 1990 paper was to show that ARCH models, which model volatility as functions of past (squared) returns, converge weakly to a diffusion process, either a diffusion for $\log \sigma_t^2$ or a CEV process as described in section 4.1.2. In particular, it was shown that a GARCH(1,1) model observed at finer and finer time intervals $\Delta t = \hbar$ with conditional variance parameters $\omega_h = \hbar \omega$, $\alpha_h = \alpha \left(h/2 \right)^{\frac{1}{2}}$ and $\beta_h = 1 - \alpha \left(h/2 \right)^{\frac{1}{2}} - \theta h$ and conditional mean $\mu_h = \hbar c \sigma_t^2$ converges to a diffusion limit quite similar to equations (4.1.8a) combined with (4.1.9) with $\delta = 1$, namely

$$d \log S_t = c\sigma_t^2 dt + \sigma_t dW_t$$

$$d\sigma_t^2 = (\omega - \theta \sigma_t^2) dt + \sigma_t^2 dW_t^{\sigma}.$$

Similarly, it was also shown that a sequence of AR(1)-EGARCH(1,1) models converges weakly to an Ornstein-Uhlenbeck diffusion for $\ln \sigma_i^2$:

$$d\ln\sigma_t^2 = \alpha \left(\beta - \ln\sigma_t^2\right) dt + dW_t^{\sigma}.$$

Hence, these basic insights showed that the continuous time stochastic difference equations emerging as diffusion limits of ARCH models were no longer ARCH but instead SV models. Moreover, following Nelson (1992) even when misspecified ARCH models still kept desirable properties regarding extracting the continuous time volatility. The argument was that for a wide variety of misspecified ARCH models the difference between the ARCH filter volatility estimates and the true underlying diffusion volatilities

³²See also Bossserts and Hillion (1995) for the use of a nonparametric hedging procedure and the smile effect

converges to zero in probability as the length of the sampling time interval goes to zero at an appropriate rate. For instance the GARCH(1,1) model with ω_h , α_h and β_h described before estimates $\ddot{\sigma}_i^2$ as follows:

$$\hat{\sigma}_{t}^{2} = \omega_{h} (1 - \beta_{h})^{-1} + \sum_{i=0}^{\infty} \alpha_{h} \beta_{h}^{i} y_{t-h(i+1)}^{2}$$

where $y_t = \log S_t / S_{t-h}$. This filter can be viewed as a particular case of equation (3.4.1). The GARCH(1,1) and many other models, effectively achieve consistent estimation of σ_t via a lag polynomial function of past squared returns close to time t.

The fact that a wide variety of misspecified ARCH models consistently extract σ_t from high frequency data raises questions regarding efficiency of filters. The answers to such questions are provided in Nelson (1955a,b) and Nelson and Foster (1994, 1995). In section 3.4 it was noted that the linear state space Kalman filter can also be viewed as a (suboptimal) extraction filter for σ_t . Nelson and Foster (1994) show that the asymptotically optimal linear Kalman filter has asymptotic variance for the normalized estimation error $h^{-\frac{1}{4}}$ [ln $(\hat{\sigma}_t^2) - \ln \sigma_t^2$] equal to $\lambda Y(1/2)^{\frac{1}{2}}$ where $Y(x) = d [\ln \Gamma(x)] / dx$ and λ is scaling factor. A model, closely related to EGARCH of the following form:

$$\begin{split} \ln\left(\hat{\sigma}_{t+h}^{2}\right) &= \ln\left(\hat{\sigma}_{t}^{2}\right) + \rho\lambda\left(S_{t+h} - S_{t}\right)\hat{\sigma}_{t}^{-1} \\ &+ \lambda\left(1 - \rho^{2}\right)^{\frac{1}{2}} \left[\Gamma\left(1/2\right)^{\frac{1}{2}} \Gamma\left(3/2\right)^{\frac{1}{2}} \left|S_{t+h} - S_{t}\right| \hat{\sigma}_{t}^{-1} - 2^{-\frac{1}{2}}\right] \end{split}$$

yields the asymptotically optimal ARCH filter with asymptotic variance for the normalized estimation error equal to $\lambda [2(1-\rho^2)]^{\frac{1}{2}}$ where the parameter ρ measure the leverage effect. These results also show that the differences between the most efficient suboptimal Kalman filter and the optimal ARCH filter can be quite substantial. Besides filtering one must also deal with smoothing and forecasting. Both of these issues were discussed in section 3.4 for discrete time SV models. The prediction properties of (misspecified) ARCH models were studied extensively by Nelson and Foster (1995). Nelson (1995) takes ARCH models a step further by studying smoothing filters, i.e. ARCH models involving not only lagged squared returns but also future realizations, i.e. r = t - T in equation (3.4.1).

4.4. Long Memory

We conclude this section with a brief discussion of long memory in continuous time SV models. The purpose is to build continuous time long memory stochastic volatility models which are relevant for high frequency financial data and for (long term) option pricing. The reasons motivating the use of long memory models were discussed in sections 2.2 and 3.5.5. The advantage of considering continuous time long memory is their relative ability to provide a more structural interpretation of the parameters governing short term and long term dynamics. The first subsection defines fractional Brownian Motion. Next we will turn our attention to the fractional SV model followed by a section on filtering and discrete time approximations.

4.4.1. Stochastic integration with respect to fractional Brownlan Motion

We recall in this subsection a few definitions and properties of fractional and long memory processes in continuous time, extensively studied for instance in Comte and Renault (1993). Consider the scalar process:

$$x_t = \int_0^t \mathbf{a} \left(t - s \right) dW_s \tag{4.4.1}$$

Such a process is asymptotically equivalent in quadratic mean to the stationary process :

$$y_t = \int_{-\infty}^{t} a(t-s) dW_s \qquad (4.4.2)$$

whenever $\int_0^{+\infty} a^2(x) dx < +\infty$. Such processes are called fractional processes if $a(x) = x^{\alpha}\tilde{a}(x) / \Gamma(1+\alpha)$ for $|\alpha| < 1/2$, \tilde{a} continuously differentiable on [0,T] and where $\Gamma(1+\alpha)$ is a scaling factor useful for normalizing fractional derivation operators on [0,T]. Such processes admit several representations, and in particular that they can also be written:

$$x_t = \int_0^t c (t - s) dW_{\alpha s}, \quad W_{\alpha t} = \int_0^t \frac{(t - s)^{\alpha}}{\Gamma(1 + \alpha)} dW_s$$
 (4.4.3)

where W_{α} is the so-called fractional Brownian Motion of order α (see Mandelbrot and Van Ness (1968)).

The relation between the functions a and c is one-to-one. One can show that W_{α} is not a semi-martingale (see e.g. Rogers (1995)) but stochastic integration with respect to W_{α} can be defined properly. The processes x_i are long memory if:

$$\lim_{x \to +\infty} \tilde{x}(x) = a_{\infty}, \ 0 < \alpha < 1/2 \text{ and } 0 < a_{\infty} < +\infty.$$

$$(4.4.4)$$

for instance,

$$dx_t = -kx_t dt + \sigma dW_{at}$$
 $x_t = 0, k > 0, 0 < \alpha < 1/2$ (4.4.5)

with its solution given by:

$$x_{t} = \int_{0}^{t} (t - s)^{\alpha} (\Gamma(1 + \alpha))^{-1} dx_{t}^{(\alpha)}$$
 (4.4.6a)

$$x_t^{(a)} = \int_0^t e^{-k(t-s)} \sigma dW_s$$
 (4.4.6b)

Note that, $x_t^{(\alpha)}$ the derivative of order α of x_t , is a solution of the usual SDE: $dz_t = -kz_t dt + \sigma dW_t$.

4.4.2. The fractional SV model

To facilitate comparison with both the FIEGARCH model and the fractional extensions of the log-Normal SV model discussed in section 3.5.5 let us consider the following fractional SV model (henceforth FSV):

$$dS_t/S_t = \sigma_t dW_t \tag{4.4.7a}$$

$$d\log\sigma_t = -k\log\sigma_t dt + \gamma dW_{ot} \tag{4.4.7b}$$

where k > 0 and $0 \le \alpha < 1/2$. If nonzero, the fractional exponent α will provide some degree of freedom in the order of regularity of the volatility process, namely the greater α the smoother the path of the volatility process. If we denote the autocovariance function of σ by $r_{\sigma}(\cdot)$ then:

$$\alpha > 0 \Rightarrow (r_{\sigma}(h) - r_{\sigma}(0))/h \rightarrow 0 \text{ as } h \rightarrow 0$$
.

This would be incorrectly interpreted as near-integrated behavior, widely found in high frequency data for instance, when:

$$r_{\sigma}(h) - r_{\sigma}(0)/h = (\rho^{h} - 1)/h \rightarrow \log \rho \text{ as } h \rightarrow 0,$$

and σ_i is a continuous time AR(1) with correlation ρ near 1.

The long memory continuous time approach allows us to model persistence with the following features

- (1) the volatility process itself (and not only its logarithm) has hyperbolic decay of the correlogram;
- (2) the persistence of volatility shocks yields leptokurtic features for return which vanishes with temporal aggregation at a slow hyperbolic rate of decay.³³ Indeed for rate of return on [0, h]:

$$\mathbb{E}\left[\log S_{t+h}/S_t - \mathbb{E}\left(\log S_{t+h}/S_t\right)\right]^4 / \left(\mathbb{E}\left[\log S_{t+h}/S_t - \mathbb{E}\left(\log S_{t+h}/S_t\right)\right]^2\right)^2 \longrightarrow 3$$

as $h \to \infty$ at a rate $h^{2\alpha-1}$ if $\alpha \in [0, 1/2]$ and a rate $\exp(-kh/2)$ if $\alpha = 0$.

4.4.3. Filtering and Discrete Time Approximations

The volatility process dynamics are described by the solution to the SDE (4.4.5), namely :

$$\log \sigma_t = \int_0^t (t-s)^{\alpha} / \Gamma (1+\alpha) d \log \sigma_s^{(\alpha)}$$
 (4.4.6)

where $\log \sigma^{(a)}$ follows the O-U process :

$$d\log\sigma_t^{(a)} = -k\log\sigma_t^{(a)}dt + \gamma dW_t \tag{4.4.7}$$

To compute a discrete time approximation one must evaluate numerically the integral (4.4.6) using only values of the process $\log \sigma^{(a)}$ on a discrete partition of [a,t] at points $j/n, j=0,1\ldots, [nt]$.³⁴ A natural way to proceed is to use step functions, generating the following proxy process:

$$\log \hat{\sigma}_t^n = \sum_{j=1}^{[nt]} \left(t - (j-1)/n\right)^{\alpha} / \Gamma(1+\alpha) \Delta \log \sigma_{j/n}^{(\alpha)}$$
(4.4.8)

where $\Delta \log \sigma_{j/n}^{(\alpha)} = \log \sigma_{j/n}^{(\alpha)} - \log \sigma_{(j-1)/n}^{(\alpha)}$. Comte and Renault (1995) show that $\log \sigma_{nt}$ converges to the $\log \sigma_t$ process for $n \to \infty$ uniformly on compact sets. Moreover, by rearranging (4.4.8) one obtains:

$$\log \hat{\sigma}_{j/n}^{n} = \left[\sum_{i=0}^{j-1} \left(\left| (i+1)^{\alpha} - i^{\alpha} \right| / n^{\alpha} \Gamma \left(1 + \alpha \right) \right) L_{n}^{i} \right] \log \sigma_{j/n}^{(\alpha)} \tag{4.4.9}$$

where L_n is the lag operator corresponding to the sampling scheme j/n, i.e. $L_n Z_{j/n} = Z_{(j-1)/n}$. With this sampling scheme $\log \sigma^{(n)}$ is a discrete time AR(1) deduced from the continuous time process with the following representation:

$$(1 - \rho_n L_n) \log \sigma_{j/n}^{(\alpha)} = u_{j/n} \tag{4.4.10}$$

 $^{34}[z]$ is the integer k such that $k \le z < k+1$.

²³With usual GARCH or SV models, it vanishes at an exponential rate (see Drost and Nijman (1993) and Drost and Werker (1994) for these issues in the short memory case).

where $\rho_n = \exp(-k/n)$ and $u_{j/n}$ is the associated innovations process. Since the process is stationary we are allowed to write (assuming $\log \sigma_{j/n}^{(a)} = u_{j/n} = 0$ for $j \le 0$):

$$\log \hat{\sigma}_{j/n}^{(n)} = \left[\sum_{i=0}^{j-1} \frac{(i+1)^{\alpha} - i^{\alpha}}{n^{\alpha} \Gamma(1+\alpha)} L_n^i \right] (1 - \rho_n L_n)^{-1} u_{j/n}$$
(4.4.11)

which gives a parametrization of the volatility dynamics in two parts: (1) a long memory part which corresponds to the filter $\sum_{i=0}^{\infty} a_i \ L_n^i/n^{\alpha}$ with $a_i = [(i+1)^{\alpha} - i^{\alpha}]/\Gamma(1+\alpha)$ and (2) a short memory part which is characterized by the AR(1) process: $(1-\rho_n L_n)^{-1} u_{j/n}$. Indeed, one can show that the long memory filter is "long-term equivalent" to the usual discrete time long memory filters $(1-L)^{-\alpha}$ in the sense that there is a long term relationship (a cointegration relation) between the two types of processes. However, this long-term equivalence between the long-memory filter and the usual discrete time one $(1-L)^{-\alpha}$ does no imply that the standard parametrization FARIMA(1, α , 0) is well-suited in our framework. Indeed, one can show that the usual discrete time filter $(1-L)^{-\alpha}$ introduces some mixing between long and short term characteristics whereas the parsimonious continuous time model doesn't. This feature clearly puts the continuous time FSV at an advantage with regard to the discrete time SV and GARCH long-memory models.

5. Statistical Inference

Evaluating the likelihood function of ARCH models is a relatively straightforward task. In sharp contrast for SV models it is impossible to obtain explicit expressions for the likelihood function. This is a generic feature common to almost all nonlinear latent variable models. The lack of estimation procedures for SV models made them for a long time an unattractive class of models in comparison to ARCH. In recent years, however, remarkable progress has been made regarding the estimation of nonlinear latent variable models in general and SV models in particular. A flurry of methods are now available and are up and running on computers with ever increasing CPU performance. The early attempts to estimate SV models used a GMM procedure. A prominent example is Melino and Turnbull (1990). Section 5.1 is devoted to GMM estimation in the context of SV models. Obviously, GMM is not designed to handle continuous time diffusions as it requires discrete time processes satisfying certain regularity conditions. A continuous time GMM approach, developed by Hansen and Scheinkman (1994), involves moment conditions directly drawn from the continuous time representation of the process. This approach is discussed in Section 5.3. In between, namely in section 5.2, we discuss the QML approach suggested by Harvey, Ruiz and Shephard (1994) and Nelson (1988). It relies on the fact that the nonlinear (Gaussian) SV model can be transformed into a linear non-Gaussian state space model as in Section 3, and from this a Gaussian quasi-likelihood can be computed. None of the methods covered in Sections 5.1 through 5.3 involve simulation. However, increased computer power has made simulation-based estimation techniques increasingly popular. The simulated method of moments, or simulation-based GMM approach proposed by Duffie and Singleton (1993), is a first example which is covered in Section 5.4. Next we discuss the indirect inference approach of Gouriéroux, Monfort and Renault (1993) and the moment matching methods of Gallant and Tauchen (1994) in Section 5.5. Finally, Section 5.6 covers a very large class of estimators using computer intensive Markov Chain Monte Carlo methods applied in the context of SV models by Jacquier, Polson and Rossi

³⁶Namely, $(1-L_n)^{\alpha} \log \partial_{i/n}^n$ is not an AR(1) process.

(1994) and Kim and Shephard (1994), and simulation based ML estimation proposed in Danielsson (1994) and Danielsson and Richard (1993).

In each section we will only try to limit our focus to the use of estimation procedures in the context of SV models and avoid details regarding econometric theory. Some useful references to complement the material which will be covered are (1) Hansen (1992), Gallant and Whits (1988), Hall (1993) and Ogaki (1993) for GMM estimation, (2) Gouriéroux and Monfort (1993b) and Wooldridge (1994) for QMLE, (3) Gouriéroux and Monfort (1995) and Tauchen (1995) for simulation based econometric methods including indirect inference and moment matching, and finally (4) Geweke (1995) and Shephard (1995) for Markov Chain Monte Carlo methods.

5.1. Generalized Method of Moments

Let us consider the simple version of the discrete time SV as presented in equations (3.1.2) and (3.1.3) with the additional assumption of normality for the probability distribution of the innovation process (ε_t, η_t) . This log-normal SV model has been the subject of at least two extensive Monte Carlo studies on GMM estimation of SV models. They were conducted by Andersen and Sørensen (1993) and Jacquier, Polson and Rossi (1994). The main idea is to exploit the stationary and ergodic properties of the SV model which yield the convergence of sample moments to their unconditional expectations. For instance, the second and fourth moments are simple expressions of σ^2 and σ_h^2 , namely $\sigma^2 \exp(\sigma_h^2/2)$ and $3\sigma^4 \exp(2\sigma_h^2)$ respectively. If these moments are computed in the sample, σ_h^2 can be estimated directly from the sample kurtosis, $\bar{\kappa}$, which is the ratio of the fourth moment to the second moment equared. The expression is just $\bar{\sigma}_h^2 = \log{(\bar{\kappa}/3)}$. The parameter σ^2 can then be estimated from the second moment by substituting in this estimate of σ_h^2 . We might also compute the first-order autocovariance of y_t^2 , or simply the sample mean of $y_t^2y_{t-1}^2$ which has expectation $\sigma^4 \exp(\{1+\phi\}\sigma_h^2\})$ and from which, given the estimate of σ^2 and σ_h^2 , it is straightforward to get an estimate of ϕ .

The above procedure is an example of the application of the method of moments. In general terms, m moments are computed. For a sample of size T, let $g_T(\beta)$ denotes the $m \times 1$ vector of differences between each sample moment and its theoretical expression in terms of the model parameters β . The generalized method of moments (GMM) estimator is constructed by minimizing the criterion function

$$\hat{\beta}_T = \underset{\beta}{\operatorname{Arg\,min}} \ g_T(\beta)' W_T g_T(\beta)$$

where W_T is an $m \times m$ weighting matrix reflecting the importance given to matching each of the moments. When ϵ_t and η_t are mutually independent, Jacquier, Poison and Rossi (1994) suggest using 24 moments. The first four are given by (3.2.2) for c = 1, 2, 3, 4, while the analytic expression for the others is:

$$E[|\ y_t^*y_{t-\tau}^c\ |] = \left\{\sigma^{2c}2^c[\Gamma\left(\tfrac{c}{2}+\tfrac{1}{2}\right)]^2/\pi\right\}\exp(\tfrac{c^2}{4}\sigma_k^2[1+\phi^\tau]), c = 1, 2, \tau = 1, 2, ..., 10.^{36}$$

In the more general case when ε_t and η_t are correlated, Melino and Turnbull (1990) included estimates of: $E[\parallel y_t \parallel y_{t-1} \mid , \tau = 0, \pm 1, \pm 2, ..., 10$. They presented an explicit expression in the case of $\tau = 1$ and show that its sign is entirely determined by ρ .

The GMM method may also be extended to handle a non-normal distribution for ε_i . The required analytic expressions can be obtained as in section 3.2. On the other hand, the analytic expression of

²⁶A simple way to derive these moment conditions is via a two-step approach similar in spirit to (2.4.8) and (2.4.9) or (3.2.3).

unconditional moments presented in section 2.4 for the general SARV model may provide the basis of GMM estimation in more general settings (see Andersen (1994)).

From the very start we expect the GMM estimator not to be efficient. The question is how much inefficiency should be tolerated in exchange for its relative simplicity. The generic setup of GMM leaves unspecified the number of moment conditions, except for the minimal number required for identification, as well as the explicit choice of moments. Moreover, the computations of the weighting matrix is also an issue since many options exist in practice. The extensive Monte Carlo studies of Andersen and Sørensen (1993) and Jacquier, Polson and Rossi (1994) attempted to answer these many outstanding questions. In general they find that GMM is a fairly inefficient procedure primarily stemming from the stylized fact, noted in section 2.2, that ϕ in equation (3.1.3) is quite close to unity in most empirical findings because volatility is highly persistent. For parameter values of ϕ close to unity convergence to unconditional moments is extremely slow suggesting that only large samples can rescue the situation. The Monte Carlo study of Andersen and Sørensen (1993) provides some guidance on how to control the extent of the inefficiency, notably by keeping the number of moment conditions small. They also provide specific recommendations for the choice of weighting matrix estimators with data-dependent bandwidth using the Bartlett kernel.

5.2. Quasi Maximum Likelihood Estimation

5.2.1. The Basic Model

Consider the linear state space model described in sub-section 3.4.1, in which (3.2.8) is the measurement equation and (3.1.3) is the transition equation. The QML estimators of the parameters ϕ , σ_{η}^2 and the variance of ξ_t , σ_{ξ}^2 , are obtained by treating ξ_t and η_t as though they were normal and maximizing the prediction error decomposition form of the likelihood obtained via the Kalman filter. As noted in Harvey, Ruiz and Shephard (1994), the quasi maximum likelihood (QML) estimators are asymptotically normal with covariance matrix given by applying the theory in Dunsmuir (1979, p. 502). This assumes that η_t and ξ_t have finite fourth moments and that the parameters are not on the boundary of the parameter space.

The parameter ω can be estimated at the same time as the other parameters. Alternatively, it can be estimated as the mean of the log y_i^2 's, since this is asymptotically equivalent when ϕ is less than one in absolute value.

Application of the QML method does not require the assumption of a specific distribution for ε_t . We will refer to this as unrestricted QML. However, if a distribution is assumed, it is no longer necessary to estimate σ_{ξ}^2 , as it is known, and an estimate of the scale factor, σ^2 , can be obtained from the estimate of ω . Alternatively, it can be obtained as suggested in sub-section 3.4.1.

If unrestricted QML estimation is carried out, a value of the parameter determining a particular distribution within a class may be inferred from the estimated variance of ξ_t . For example in the case of the Student's t, ν may be determined from the knowledge that the theoretical value of the variance of ξ_t is $4.93 + \psi(\nu/2)$ (where $\Psi(\cdot)$ is the digamma function introduced in section 3.2.2).

5.2.2. Asymmetric Model

In an asymmetric model, QML may be based on the modified state space form in (3.4.3). The parameters σ_{ξ}^2 , σ_{η}^2 , ϕ , μ^* , and γ^* can be estimated via the Kalman filter without any distributional assumptions, apart from the existence of fourth moments of η_t and ξ_t and the joint symmetry of ξ_t and η_t . However, if an estimate of ρ is wanted it is necessary to make distributional assumptions about the disturbances, leading to formulae like (3.4.4) and (3.4.5). These formulae can be used to set up an optimization with respect to

the original parameters σ^2 , σ_n^2 , ϕ and ρ . This has the advantage that the constraint $|\rho| < 1$ can be imposed. Note that any t-distribution gives the same relationship between the parameters, so within this class it is not necessary to specify the degrees of freedom.

Using the QML method with both the original disturbances assumed to be Gaussian, Harvey and Shephard (1993) estimate a model for the CRSP daily returns on a value weighted US market index for 3rd July 1962 to 31st December 1987. These data were used in the paper by Nelson (1991) to illustrate his EGARCH model. The empirical results indicate a very high negative correlation.

5.2.3. QML in the Frequency Domain

For a long memory SV model, QML estimation in the time domain becomes relatively less attractive because the state space form (SSF) can only be used by expressing h_t as an autoregressive or moving average process and truncating at a suitably high lag. Thus the approach is cumbersome, though the initial state covariance matrix is easily constructed, and the truncation does not affect the asymptotic properties of the estimators. If the autoregressive approximation, and therefore the SSF, is not used, time domain QML requires the repeated construction and inversion of the $T \times T$ covariance matrix of the $\log y_t^{D}$; see Sowell (1992). On the other hand, QML estimation in the frequency domain is no more difficult than it is in the AR(1) case. Cheung and Diebold (1994) present simulation evidence which suggests that although time domain estimation is more efficient in small samples, the difference is less marked when a mean has to be estimated.

The frequency domain (quasi) log-likelihood function is, neglecting constants,

$$\log L = -\frac{1}{2} \sum_{i=1}^{T-1} \log g_i - \pi \sum_{i=1}^{T-1} I(\lambda_i)/g_i$$
 (5.2.1)

where $I(\lambda_j)$ is the sample spectrum of the $\log y_i^p$ s and g_j is the spectral generating function (SGF), which for (3.5.1) is

$$g_j = \sigma_n^2 [2(1 - \cos \lambda_j)]^{-d} + \sigma_\ell^2$$

Note that the summation in (5.2.1) is from j=1 rather than j=0. This is because g_0 cannot be evaluated for positive d. However, the omission of the zero frequency does remove the mean. The unknown parameters are σ_{η}^2 , σ_{ξ}^2 and d, but σ_{ξ}^2 may be concentrated out of the likelihood function by a reparameterisation in which σ_{η}^2 is replaced by the signal-noise ratio $q=\sigma_{\eta}^2/\sigma_{\xi}^2$. On the other hand if a distribution is assumed for ε_{ℓ} , then σ_{ξ}^2 is known. Breidt, Crato and de Lima (1993) show the consistency of the QML estimator.

When d lies between 0.5 and one, h_i is nonstationary, but differencing the $\log y_i^{2s}$ yields a zero mean stationary process, the SGF of which is

$$g_j = \sigma_{\eta}^2 [2(1 - \cos \lambda_j)]^{1-d} + 2(1 - \cos \lambda_j)\sigma_{\xi}^2$$

One of the attractions of long memory models is that inference is not affected by the kind of unit root issues which arise with autoregressions. Thus a likelihood based test of the hypotheses that d=1 against the alternative that it is less than one can be constructed using standard theory; see Robinson (1993).

5.2.4. Comparison of GMM and QML

Simulation evidence on the finite sample performance of GMM and QML can be found in Andersen and Sørensen (1993), Ruiz (1994), Jacquier, Polson and Rossi (1994), Breidt and Carriquiry (1995), Andersen

and Sørensen (1996) and Harvey and Shephard (1996). The general conclusion seems to be that QML gives estimates with a smaller MSE when the volatility is relatively strong as reflected in a high coefficient of variation. This is because the normally distributed volatility component in the measurement equation, (3.2.8), is large relative to the non-normal error term. With a lower coefficient of variation, GMM dominates. However, in this case Jacquier, Polson and Rosei (1994, p. 383) observe that "...the performance of both the QML and GMM estimators deteriorates rapidly." In other words the case for one of the more computer intensive methods outlined in Section 5.6 becomes stronger.

Other things being equal, an AR coefficient, ϕ , close to one tends to favor QML because the autocorrelations are slow to die out and are hence captured less well by the moments used in GMM. For the same

reason, GMM is likely to be rather poor in estimating a long memory model.

The attraction of QML is that it is very easy to implement and it extends easily to more general models, for example nonstationary and multivariate ones. At the same time, it provides filtered and smoothed estimates of the state, and predictions. The one-step ahead prediction errors can also be used to construct diagnostics, such as the Box-Ljung statistic, though in evaluating such tests it must be remembered that the observations are non-normal. Thus even if the hyperparameters are eventually estimated by another method, QML may have a valuable role to play in finding a suitable model specification.

5.3. Continuous Time GMM

Hansen and Scheinkman (1995) propose to estimate continuous time diffusions using a GMM procedure specifically tailored for such processes. In section 5.1 we discussed estimation of SV models which are either explicitly formulated as discrete time processes or else are discretizations of the continuous time diffusions. In both cases inference is based on minimizing the difference between unconditional moments and their sample equivalent. For continuous time processes Hansen and Scheinkman (1995) draw directly upon the diffusion rather than its discretization to formulate moment conditions. To describe the generic setup of the method they proposed let us consider the following (multivariate) system of n diffusion equations:

$$dy_t = \mu(y_t; \theta)dt + \sigma(y_t; \theta)dW_t \qquad (5.3.1)$$

A comparison with the notation in section 2 immediately draws attention to certain limitations of the setup. First, the functions $\mu_{\theta}\left(\cdot\right) \equiv \mu\left(\cdot;\theta\right)$ and $\sigma_{\theta}\left(\cdot\right) \equiv \sigma\left(\cdot;\theta\right)$ are parameterized by y_t only which restricts the state variable process U_t in section 2 to contemporaneous values of y_t . The diffusion in (5.3.1) involves a general vector process y_t , hence y_t could include a volatility process to accommodate SV models. Yet, the y_t vector is assumed observable. For the moment we will leave these issues aside, but return to them at the end of the section. Hansen and Scheinkman (1995) consider the infinitesimal operator A defined for a class of square integrable functions $\varphi \in \mathbb{R}^n \to \mathbb{R}$ as follows:

$$A_{\theta}\varphi\left(y\right) = \frac{d\varphi\left(y\right)}{dy'}\mu_{\theta}\left(y\right) + \frac{1}{2}Tr\left(\sigma_{\theta}\left(y\right)\sigma_{\theta}'\left(y\right)\frac{d^{2}\varphi\left(y\right)}{dydy'}\right). \tag{5.3.2}$$

Because the operator is defined as a limit, namely :

$$A_{\theta}\varphi\left(y\right)=\lim_{t\to0}t^{-1}[\mathbb{E}\left(\varphi(y_{t})\right|y_{o}=y)-y],$$

it does not necessarily exist for all square integrable functions φ but only for a restricted domain D. A set of moment conditions can now be obtained for this class of functions $\varphi \in D$. Indeed, as shown for instance by Revuz and Yor (1991), the following equalities hold:

$$EA_{\theta}\varphi\left(y_{\theta}\right)=0,\tag{5.3.3}$$

$$E\left[A_{\theta}\varphi\left(y_{i+1}\right)\tilde{\varphi}\left(y_{i}\right)-\varphi\left(y_{i+1}\right)A_{\theta}^{*}\tilde{\varphi}\left(y_{i}\right)\right]=0,\tag{5.3.4}$$

where A_x^a is the adjoint infinitesimal operator of A_θ for the scalar product associated with the invariant measure of the process y.²⁷ By choosing an appropriate set of functions, Hansen and Scheinkman exploit moment conditions (5.3.3) and (5.3.4) to construct a GMM estimator of θ .

The choice of the function $\varphi \in D$ and $\tilde{\varphi} \in D^*$ determines what moments of the data are used to estimate the parameters. This obviously raises questions regarding the choice of functions to enhance efficiency of the estimator but first and foremost also the identification of θ via the conditions (5.3.3) and (5.3.4). It was noted in the beginning of the section that the multivariate process y_t , in order to cover SV models, must somehow include the latent conditional variance process. Gouriéroux and Monfort (1994, 1995) point out that since the moment conditions based on φ and $\bar{\varphi}$ cannot include any latent process it will often (but not always) be impossible to attain identification of all the parameters, particularly those governing the latent volatility process. A possible remedy is to augment the model with observations indirectly related to the latent volatility process, in a sense making it observable. One possible candidate would be to include in y_t both the security price and the Black-Scholes implied volatilities obtained through option market quotations for the underlying asset. This approach is in fact suggested by Pastorello, Renault and Touzi (1993) although not in the context of continuous time GMM but instead using indirect inference methods which will be discussed in section 5.5.38 Another possibility is to rely on the time deformation representation of SV models as discussed in the context of continuous time GMM by Conjev et al. (1995).

5.4. Simulated Method of Moments

The estimation procedures discussed so far do not involve any simulation techniques. From now on we cover methods combining simulation and estimation beginning with the simulated method of moments (SMM) estimator, which is covered by Duffie and Singleton (1993) for time series processes. In section 5.1 we noted that GMM estimation of SV models is based on minimizing the distance between a set of chosen sample moments and unconditional population moments expressed as analytical functions of the model parameters. Suppose now that such analytical expressions are hard to obtain. This is particularly the case when such expressions involve marginalizations with respect to a latent process such a stochastic volatility process. Could we then simulate data from the model for a particular value of the parameters and match moments from the simulated data with sample moments as a substitute? This strategy is precisely what SMM is all about. Indeed, quite often it is fairly straightforward to simulate processes and therefore take advantage of the SMM procedure. Let us consider again as point of reference and illustration the (multivariate) diffusion of the previous section (equation (5.3.1)) and conduct H simulations i=1,...,H using a discretization:

$$\Delta \tilde{y}_{t}^{i}(\theta) = \mu(\tilde{y}_{t}^{i}(\theta); \theta) + \sigma(\tilde{y}_{t}^{i}(\theta); \theta) \varepsilon_{t} \text{ and } i = 1, ..., H \text{ and } t = 1, ..., T$$

³⁷Please note that A_d^* is again associated with a domain D^* so that $\varphi \in D$ and $\tilde{\varphi} \in D^*$ in (5.3.4).

³⁸It was noted in section 2.1.3 that implied volatilities are biased. The indirect inference procedures used by Pastorello, Renault and Touzi (1993) can cope with such biases, as will be explained in section 5.5. The use of option price data is further discussed in section 5.7.

²⁰SMM was originally proposed for cross-section applications, see Pakes and Pollard (1989) and McFadden (1989). See also Gouriéroux and Monfort (1993a).

where $\hat{y}_t(\theta)$ are simulated given a parameter θ and ϵ_t is i.i.d. Gaussian.⁴⁰ Subject to identification and other regularity conditions one then considers

$$\widehat{\theta}_{T}^{H} = \underset{\boldsymbol{\theta}}{Argmin} \parallel f\left(y_{1},...y_{T}\right) - \frac{1}{H} \sum_{i=1}^{H} f\left(\widehat{y}_{i}\left(\boldsymbol{\theta}\right),...,\widehat{y}_{T}^{i}\left(\boldsymbol{\theta}\right)\right) \parallel$$

with a suitable choice of norm, i.e. weighting matrix for the quadratic form as in GMM, and function f of the data, i.e. moment conditions. The asymptotic distribution theory is quite similar to that of GMM, except that simulation introduces an extra source of random error affecting the efficiency of the SMM estimator in comparison to its GMM counterpart. The efficiency loss can be controlled by the choice of $H.^{41}$

5.5. Indirect Inference and Moment Matching

The key insight of the indirect inference approach of Gouriéroux, Monfort and Renault (1993) and the moment matching approach of Gallant and Tauchen (1994) is the introduction of an auxiliary model parameterized by a vector, say β , in order to estimate the model of interest. In our case the latter is the SV model.⁴² In the first subsection we will describe the general principle while a second one will focus exclusively on estimating diffusions.

5.5.1. The Principle

We noted at the beginning of section 5 that ARCH type models are relatively easy to estimate in comparison to SV models. For this reason an ARCH type model may be a possible candidate as an auxiliary model. An alternative strategy would be to try to summarize the features of the data via a SNP density as developed by Gallant and Tauchen (1989). This empirical SNP density, or more specifically its score, could also fulfill the role of auxiliary model. Other possibilities could be considered as well. The idea is then to use the auxiliary model to estimate β , so that:

$$\widehat{\beta}_{T} = \underset{\beta}{\operatorname{Argmax}} \sum_{t=1}^{T} \log f^{*}(y_{t} \mid y_{t-1}, \beta)$$
(5.5.1)

where we restrict our attention here to a simple dynamic model with one lag for the purpose of illustration. The objective function f^* in (5.5.1) can be a pseudo-likelihood function when the auxiliary model is deliberately misspecified to facilitate estimation. As an alternative f^* can be taken from the class of SNP densities. On Gouriéroux, Monfort and Renault then propose to estimate the same parameter vector $\boldsymbol{\beta}$ not using the actual sample data but instead using samples $\{\tilde{y}_i^t(\theta)\}_{i=1}^T$ simulated i=1,...H times drawn from the model of interest given θ . This yields a new estimator of $\boldsymbol{\beta}$, namely:

⁴⁰We discuss in detail the simulation techniques in the next section. Indeed, to control for the discretization bias, one has to simulate with a finer sampling interval.

⁴³The asymptotic variance of the SMM estimator depends on H through a factor(1 + H^{-1}), see e.g. Gouriéroux and Monfort (1995).

⁴² It is worth noting that the simulation based inference methods we will describe here are applicable to many other types of models for cross-sectional, time series and panel data.

⁴³The discussion should not leave the impression that the auxiliary model can only be estimated via ML-type estimators. Any root T consistent saymptotically normal estimations procedure may be used.

$$\widehat{\beta}_{HT}(\theta) = \underset{\beta}{\operatorname{Argmax}} \left(1/H \right) \sum_{i=1}^{H} \sum_{t=1}^{T} \log f^{*} \left(\vec{y}_{t}^{i}(\theta) \mid \vec{y}_{t-1}^{i}(\theta), \beta \right). \tag{5.5.2}$$

The next step is to minimize a quadratic distance using a weighting matrix W_T to choose an indirect estimator of θ based on H simulation replications and a sample of T observations, namely:

$$\hat{\theta}_{HT} = Argmin \left(\hat{\beta}_{T} - \hat{\beta}_{HT}(\theta) \right)' W_{T} \left(\hat{\beta}_{T} - \hat{\beta}_{HT}(\theta) \right)$$
(5.5.3)

The approach of Gallant and Tauchen (1994) avoids the step of estimating $\hat{\beta}_{HT}(\theta)$ by computing the score function of f^* and minimizing a quadratic distance similar to (5.5.3) but involving the score function evaluated at $\hat{\beta}_T$ and replacing the sample data by simulated series generated by the model of interest. Under suitable regularity conditions the estimator θ_{HT} is root T consistent and asymptotically normal. As with GMM and SMM there is again an optimal weighting matrix. The resulting asymptotic covariance matrix depends on the number of simulations in the same way the SMM estimator depends on H.

Gouriéroux, Monfort and Renault (1993) illustrated the use of indirect inference estimator with a simple example that we would like to briefly discuss here. Typically AR models are easy to estimate while MA models require more elaborate procedures. Suppose the model of interest is a moving average model of order one with parameter θ . Instead of estimating the MA parameter directly from the data they propose to estimate an AR(p.) model involving the parameter vector β . The next step then consists of simulating data using the MA model and proceed further as described above. They found that the indirect inference estimator for $\hat{\theta}_{HT}$ appeared to have better finite sample properties than the more traditional maximum likelihood estimators for the MA parameter. In fact the indirect inference estimator exhibited features similar to the median unbiased estimator proposed by Andrews (1993). These properties were confirmed and clarified by Gouriéroux, Renault and Touzi (1994) who studied the second order asymptotic expansion of indirect inference estimators and their ability to reduce finite sample bias.

5.5.2. Estimating Diffusions

Let us consider the same diffusion equation as in section 5.3 which dealt with continuous time GMM, namely:

$$dy_t = \mu(y_t; \theta) dt + \sigma(y_t; \theta) dW_t \qquad (5.5.4)$$

In section 5.3 we noted that the above equation holds under certain restrictions such as the functions μ and σ being restricted to y_t as arguments. While these restrictions were binding for the setup of section 5.3 this will not be the case for the estimation procedures discussed here. Indeed, equation (5.5.4) is only used as an illustrative example. The diffusion is then simulated either via exact discretizations or some type of approximate discretization (e.g. Euler or Mil'shtein, see Pardoux and Talay (1985) or Kloeden and Platten (1992) for further details). More precisely we define the process $y_t^{(0)}$ such that:

$$y_{(k+1)k}^{(\delta)} = y_{k\delta}^{(\delta)} + \mu \left(y_{k\delta}^{(\delta)}; \theta\right) \delta + \sigma \left(y_{k\delta}^{(\delta)}; \theta\right) \delta^{1/2} \varepsilon_{(k+1)k}^{(\delta)}$$
 (5.5.5)

[&]quot;Again one could use a score principle here, following Gallant and Tanchen (1994). In fact in a linear Gaussian setting the SNP approach to fit data generated by a MA (1) model would be to estimate on AR(p) model. Glysels, Khalaf and Vodounou (1994) provide a more detailed discussion of score-based and indirect inference estimators of MA models as well as their relation with more standard estimators.

Under suitable regularity conditions (see for instance Strook and Varadhan (1979)) we know that the diffusion admits a unique solution (in distribution) and the process $y_t^{(\delta)}$ converges to y_t as δ goes to zero. Therefore one can expect to simulate y_t quite accurately for δ sufficiently small. The auxiliary model may be a discretization of (5.5.4) choosing $\delta = 1$. Hence, one formulates a ML estimator based on the nonlinear AR model appearing in (5.5.5) setting $\delta = 1$. To control for the discretization bias one can simulate the underlying diffusion with $\delta = 1/10$ or 1/20, for instance, and aggregate the simulated data to correspond with the sampling frequency of the DGP. Broze, Scaillet and Zakoïan (1994) discuss of the effect of the simulation step size on the asymptotic distribution.

The use of simulation-based inference methods becomes particularly appropriate and attractive when diffusions involve latent processes, such as is the case with SV models. Gouriéroux and Monfort (1994, 1995) discuss several examples and study their performance via Monte Carlo simulation. It should be noted that estimating the diffusion at a coarser discretization is not the only possible choice of auxiliary model. Indeed, Pastorello, Renault and Touzi (1993), Engle and Lee (1994) and Gallant and Tauchen (1994) suggest the use of ARCH-type models.

There have been several successful applications of these methods to financial time series. They include Broze et al. (1995), Engle and Lee (1994), Gallant, Hsieh and Tauchen (1994), Gallant and Tauchen (1994, 1995), Ghysels, Gouriéroux and Jasiak (1995b), Ghysels and Jasiak (1994a and b), Pastorello et al. (1993), among others.

5.6. Likelihood-based and Bayesian Methods

In a Gaussian linear state space model the likelihood function is constructed from the one step ahead prediction errors. This prediction error decomposition form of the likelihood is used as the criterion function in QML, but of course it is not the exact likelihood in this case. The exact filter proposed by Watanabe (1993) will, in principle, yield the exact likelihood. However, as was noted in section 3.4.2, because this filter uses numerical integration, it takes a long time to compute and if numerical optimization is to be carried out with respect to the hyperparameters it becomes impractical.

Kim and Shephard (1994) work with the linear state space form used in QML but approximate the $\log(\chi^2)$ distribution of the measurement error by a mixture of normals. For each of these normals, a prediction error decomposition likelihood function can be computed. A simulated EM algorithm is used to find the best mixture and hence calculate approximate ML estimates of the hyperparameters.

The exact likelihood function can also be constructed as a mixture of distributions for the observations conditional on the volatilities, that is

$$L(y;\phi,\sigma_{\eta}^{2},\sigma^{2})=\int p(y|h)p(h)dh$$

where y and h contain the T elements of y_t and h_t respectively. This expression can be written in terms of the $\sigma_t^{p}s$, rather than their logarithms, the $h_t's$, but it makes little difference to what follows. Of course the problem is that the above likelihood has no closed form, so it must be calculated by some kind of simulation method. Excellent discussions can be found in Shephard (1995) and in Jacquier, Polson and Rossi (1994), including the comments. Conceptually, the simplest approach is to use Monte Carlo integration by drawing from the unconditional distribution of h for given values of the parameters, $(\phi, \sigma_\eta^2, \sigma^2)$, and estimating the likelihood as the average of the p(y|h)'s. This is then repeated, searching over ϕ, σ_η^2 until the maximum of the simulated likelihood is found. As it stands this procedure is not very satisfactory, but it may be improved by using ideas of importance sampling. This has been implemented for ML estimation of SV

models by Danielsson and Richard (1993) and Danielsson (1994). However, the method becomes more difficult as the sample size increases.

A more promising way of attacking likelihood estimation by simulation techniques is to use Markov Chain Monte Carlo (MCMC) to draw from the distribution of volatilities conditional on the observations. Ways in which this can be done were outlined in sub-section 3.4.2 on nonlinear filters and smoothers. Kim and Shephard (1994) suggest a method of computing ML estimators by putting their multimove algorithm within a simulated EM algorithm. Jacquier, Polson and Rossi (1994) adopt a Bayesian approach in which the specification of the model has a hierarchical structure in which a prior distribution for the hyperparameters, $\varphi = (\sigma_n, \phi, \sigma)$, joins the conditional distributions, yih and hig. (Actually the σ 's are used rather than the h_i 's). The joint posterior of h and φ is proportional to the product of these three distributions, that is $p(h,\varphi|y) \propto p(y|h)p(h|\varphi)p(\varphi)$. The introduction of h makes the statistical treatment tractable and is an example of what is called data augmentation; see Tanner and Wong (1987). From the joint posterior, $p(h,\varphi|y)$, the marginal p(h|y) solves the smoothing problem for the unobserved volatilities, taking account of the sampling variability in the hyperparameters. Conditional on h, the posterior of φ , $p(\varphi|h,y)$ is simple to compute from standard Bayesian treatment of linear models. If it were also possible to sample directly from $p(h|\varphi,y)$ at low cost, it would be straightforward to construct a Markov chain by alternating back and forth drawing from $p(\varphi|h,y)$ and $p(h|\varphi,y)$. This would produce a cyclic chain, a special case of which is the Gibbs sampler. However, as was noted in sub-section 3.4.2, Jacquier, Polson and Rosei (1994) show that it is much better to decompose $p(h|\varphi, y)$ into a set of univariate distributions in which each h_t , or rather σ_t , is conditioned on all the others.

The prior distribution for ω , the parameters of the volatility process in JPR (1994), is the standard conjugate prior for the linear model, a (truncated) Normal-Gamma. The priors can be made extremely diffuse while remaining proper. JPR conduct an extensive sampling experiment to document the performance of this and more traditional approaches. Simulating stochastic volatility series, they compare the sampling performances of the posterior mean with that of the QML and GMM point estimates. The MCMC posterior mean exhibit root mean squared errors anywhere between half and a quarter of the size of the GMM and QML point estimates. Even more striking are the volatility smoothing performance results. The root mean squared error of the posterior mean of h_i produced by the Bayesian filter is 10% smaller than the point estimate produced by an approximate Kalman filter supplied with the true parameters.

Shephard and Kim in their comment of JPR (1994) point out that for very high ϕ and small σ_η , the rate of convergence of the JPR algorithm will slow down. More draws will then be required to obtain the same amount of information. They propose to approximate the volatility disturbance with a discrete mixture of normals. The benefit of the method is that a draw of the vector h is then possible, faster than T draws from each h_t . However this is at the cost that the draws navigate in a much higher dimensional space due to the discretization effected. Also, the convergence of chains based upon discrete mixtures is sensitive to the number of components and their assigned probability weights. Mahieu and Schotman (1994) add some generality to the Shephard and Kim idea by letting the data produce estimates of the characteristics of the discretized state space (probabilities, mean and variance).

The original implementation of the JPR algorithm was limited to a very basic model of stochastic volatility, AR(1) with uncorrelated mean and volatility disturbances. In a univariate setup, correlated disturbances are likely to be important for stock returns, i.e., the so called leverage effect. The evidence in Gallant, Rossi, and Tauchen (1994) also points at non normal conditional errors with both skewness and kurtosis. Jacquier, Polson, and Rossi (1995a) show how the hierarchical framework allows the convenient extension of the MCMC algorithm to more general models. Namely, they estimate univariate stochastic volatility models with correlated disturbances, and skewed and fat-tailed variance disturbance, as well as

multivariate models. Alternatively, the MCMC algorithm can be extended to a factor structure. The factors exhibit stochastic volatility and can be observable or non observable.

5.7. Inference and Option Price Data

Some of the continuous time SV models currently found in the literature were developed to answer questions regarding derivative security pricing. Given this rather explicit link between derivates and SV diffusions it is perhaps somewhat surprising that relatively little attention has been paid to the use of option price data to estimate continuous time diffusions. Melino (1994) in his survey in fact notes: "Clearly, information about the stochastic properties of an asset's price is contained both in the history of the asset's price and the price of any options written on it. Current strategies for combining these two sources of information, including implicit estimation, are uncomfortably ad hoc. Statistically speaking, we need to model the source of the prediction errors in option pricing and to relate the distribution of these errors to the stock price process. For example implicit estimation, like computation of BS implied volatilities, is certainly uncomfortably ad hoc from a statistical point of view. In general, each observed option price introduces one source of prediction error when compared to a pricing model. The challenge is to model the joint nondegenerate probability distribution of options and asset prices via a number of unobserved state variables. This approach has been pursued in a number of recent papers, including Christensen (1992), Renault and Touzi (1992), Pastorello et al. (1993), Duan (1994) and Renault (1995).

Christensen (1992) considers a pricing model for n assets as a function of a state vector x_t which is (l+n) dimensional and divided in a l-dimensional observed (z_t) and n-dimensional unobserved (ω_t) component. Let p_t be the price vector of the n assets, then:

$$p_t = m\left(z_t, \omega_t, \theta\right) \tag{5.7.1}$$

Equation (5.7.1) provides a one-to-one relationship between the n latent state variables ω_t and the n observed prices p_t , for given z_t and θ . From a financial viewpoint, it implies that the n assets are appropriate instruments to complete the markets if we assume that the observed state variables z_t are already mimicked by the price dynamics of other (primitive) assets. Moreover, from a statistical viewpoint it allows full structural maximum likelihood estimation provided the log-likelihood function for observed prices can be deduced easily from a statistical model for x_t . For instance, in a Markovian setting where, conditionally on x_0 , the joint distribution of $x_1^T = (x_t)_{1 \le t \le T}$ is given by the density:

$$f_{x}\left(x_{1}^{T}|x_{0},\theta\right) = \prod_{t=1}^{T} f\left(z_{t},\omega_{t}|z_{t-1},\omega_{t-1},\theta\right)$$
(5.7.2)

the conditional distribution of data $D_1^T = (p_t, z_t)_{1 \le t \le T}$ given $D_0 = (p_0, z_0)$ is obtained by the usual Jacobian formula:

$$f_{D}\left(D_{1}^{T}|D_{0},\theta\right) = \prod_{t=1}^{T} f\left[z_{t}, m_{\theta}^{-1}(z_{t}, p_{t}) \middle| z_{t-1}, m_{\theta}^{-1}(z_{t-1}, p_{t-1}), \theta\right] x \\ \left|\nabla_{\omega} m\left(z_{t}, m_{\theta}^{-1}(z_{t}, p_{t}), \theta\right)\right|^{-1}$$
(5.7.3)

where $m_{\theta}^{-1}(z,.)$ is the ω -inverse of $m(z,.,\theta)$ defined formally by $m_{\theta}^{-1}(z,m(z,\omega,\theta)) = \omega$ while $\nabla_{\omega}m(\cdot)$ represents the columns corresponding to ω of the Jacobian matrix. This MLE using price data of derivatives was proposed independently by Christensen (1992) and Duan (1994). Renault and Touzi (1992) were instead more specifically interested in the Hull and White option pricing formula with: $z_t = S_t$ observed

underlying asset price, and $\omega_t = \sigma_t$ unobserved stochastic volatility process. Then with the joint process $x_t = (S_t, \sigma_t)$ being Markovian we have a call price of the form:

$$C_t = m(x_t, \theta, K)$$

where $\theta = (\alpha', \gamma')$ involves two types of parameters: (1) the vector α of parameters describing the dynamics of the joint process $x_t = (S_t, \sigma_t)$ which under the equivalent martingale measure allows to compute the expectation with respect to the (risk-neutral) conditional probability distribution of γ^2 (t, t + h) given σ_t ; and (2) the vector γ of parameters which characterize the risk premia determining the relation between the risk neutral probability distribution of the x process and the Data Generating Process.

Structural MLE is often difficult to implement. This motivated Renault and Touzi (1992) and Pastorello, Renault and Touzi (1993) to consider less efficient but simpler and more robust procedures involving some proxies of the structural likelihood (5.7.3).

To illustrate these procedures let us consider the standard log-normal SV model in continuous time:

$$d\log\sigma_t = k\left(a - \log\sigma_t\right)dt + cdW_t^{\sigma}. \tag{5.7.4}$$

Standard option pricing arguments allow us to ignore misspecifications of the drift of the underlying asset price process. Hence, a first step towards simplicity and robustness is to isolate from the likelihood function the volatility dynamics, namely:

$$\prod_{i=1}^{n} \left(2\pi c^{2} \right)^{-\frac{1}{2}} \exp \left[-\left(2c^{2} \right)^{-1} \left(\log \sigma_{t_{i}} - e^{-k\Delta t} \log \sigma_{t_{i-1}} - a \left(1 - e^{-k\Delta t} \right) \right) \right]^{2}$$
 (5.7.5)

associated with a sample σ_{t_1} , $i=1,\ldots,n$ and $t_i-t_{i-1}=\Delta t$. To approximate this expression one can consider a direct method, as in Renault and Touzi (1992) or an indirect method, as in Pastorello et al. (1993). The former involves calculating implied volatilities from the Hull and White model to create pseudo samples σ_{t_1} parameterized by k, a and c and computing the maximum of (5.7.5) with respect to those three parameters. Pastorello et al. (1993) proposed several indirect inference methods, described in section 5.5, in the context of (5.7.5). For instance, they propose to use an indirect inference strategy involving GARCH(1,1) volatility estimates obtained form the underlying asset (also independently suggested by Engle and Lee (1994)). This produces asymptotically unbiased but rather inefficient estimates. Pastorello et al. indeed find that an indirect inference simplification of the Renault and Touzi direct procedure involving option prices is far more efficient. It is a clear illustration of the intuition that the use of option price data paired with suitable statistical methods should largely improve the accuracy of estimating volatility diffusion parameters.

5.8. Regression Models with Stochastic Volatility

A single equation regression model with stochastic volatility in the disturbance term may be written

$$y_t = x_t'\beta + u_t, \quad t = 1, ..., T,$$
 (5.8.1)

where y_t denotes the t-th observation, x_t is a $k \times 1$ vector of explanatory variables, β is a $k \times 1$ vector of coefficients and $u_t = \sigma \varepsilon_t \exp{(0.5h_t)}$ as discussed in section 3. As a special case, the observations may simply have a non-zero mean so that $x_t'\beta = \mu \forall t$.

⁴⁵The direct maximization of (5.7.5) using BS implied volatilities has also been proposed, see e.g. Heynes, Kenna and Vorst (1994). Obviously the use of BS implied volatility induces a misspecification bias due to the BS model assumptions.

Since u_t is stationary, an OLS regression of y_t on x_t yields a consistent estimator of β . However it is not efficient.

For given values of the SV parameters, ϕ and σ_{η}^2 , a smoothed estimator of h_t , $h_{t|T}$, can be computed using one of the methods outlined in section 3.4. Multiplying (5.8.1) through by $\exp(-..5h_{t|T})$ gives

$$\tilde{y}_t = \tilde{\mathbf{x}}_t' \boldsymbol{\beta} + \tilde{\mathbf{u}}_t, t = 1, ..., T$$
 (5.8.2)

where the $\widetilde{u_i}$'s can be thought of as heteroskedasticity corrected disturbances. Harvey and Shephard (1993) show that these disturbances have zero mean, constant variance and are serially uncorrelated and hence suggest the construction of a feasible GLS estimator

$$\tilde{\beta} = \left[\sum_{t=1}^{T} e^{-h_{t|T}} x_t x_t' \right]^{-1} \sum_{t=1}^{T} e^{-h_{t|T}} x_t y_t$$
 (5.8.3)

In the classical heteroskedastic regression model h_t is deterministic and depends on a fixed number of unknown parameters. Because these parameters can be estimated consistently, the feasible GLS estimator has the same asymptotic distribution as the GLS estimator. Here h_t is stochastic and the MSE of its estimator is of O(1). The situation is therefore somewhat different. Harvey and Shephard (1993) show that, under standard regularity conditions on the sequence of x_t , $\tilde{\beta}$ is asymptotically normal with mean β and a covariance matrix which can be consistently estimated by

$$av\widetilde{ar}(\widetilde{\beta}) = \left[\sum_{t=1}^{T} e^{-h_{t|T}} \mathbf{x}_t \mathbf{x}_t'\right]^{-1} \sum_{t=1}^{T} (y_t - \mathbf{x}_t' \widetilde{\beta})^2 e^{-2h_{t|T}} \mathbf{x}_t \mathbf{x}_t' \left[\sum_{t=1}^{T} e^{-h_{t|T}} \mathbf{x}_t \mathbf{x}_t'\right]^{-1}$$
(5.8.4)

When $h_{t|T}$ is the smoothed estimate given by the linear state space form, the analysis in Harvey and Shephard (1993) suggests that, asymptotically, the feasible GLS estimator is almost as efficient as the GLS estimator and considerably more efficient than the OLS estimator. It would be possible to replace $\exp(h_{t|T})$ by a better estimate computed from one of the methods described in section 3.4 but this may not have much effect on the efficiency of the resulting feasible GLS estimator of β .

When h_t is nonstationary, or nearly nonstationary, Hansen (1995) shows that it is possible to construct a feasible adaptive least squares estimator which is asymptotically equivalent to GLS.

6. Conclusions

No survey is ever complete. There are two particular areas we expect to flourish in the years to come but which we were not able to cover. The first is the area of market microstructures which is well surveyed in a recent review paper by Goodhart and O'Hara (1995). With the ever increasing availability of high frequency data series, we anticipate more work involving game theoretic models. These can now be estimated because of recent advances in econometric methods, similar to those enabling us to estimate diffusions. Another area where we expect interesting research to emerge is that involving nonparametric procedures to estimate SV continuous time and derivative securities models. Recent papers include Ait-Sahalia (1994), Ait-Sahalia et al. (1994), Bossaerts, Hafner and Härdle (1995), Broadie et al. (1995), Conley et al. (1995), Elsheimer et al. (1995), Gouriéroux, Monfort and Tenreiro (1994), Gouriéroux and Scaillet (1995), Hutchinson, Lo and Poggio (1994), Lezan et al. (1995), Lo (1995), Pagan and Schwert (1992).

Research into the econometrics of Stochastic Volatility models is relatively new. As our survey has shown, there has been a burst of activity in recent years drawing on the latest statistical technology. As

regards the relationship with ARCH, our view is that SV and ARCH are not necessarily direct competitors, but rather complement each other in certain respects. Recent advances such as the use of ARCH models as filters, the weakening of GARCH and temporal aggregation and the introduction of nonparametric methods to fit conditional variances, illustrate that a unified strategy for modelling volatility needs to draw on both ARCH and SV.

BIBLIOGRAPHY

- Abramowitz, M. and N.C. Stegun (1970), Handbook of Mathematical Functions, (Dover Publications Inc., New York).
- [2] Ait-Sahalia, Y. (1994), "Nonparametric Pricing of Interest Rate Derivative Securities". Discussion Paper, Graduate School of Business, University of Chicago.
- [3] Ait-Sahalia, Y., S.J. Bickel and T.M. Stoker (1994), "Goodness-of-Fit Tests for Regression Using Kernel Methods", Discussion Paper, University of Chicago.
- [4] Amin, K.L. and V. Ng (1993) "Equilibrium Option Valuation with Systematic Stochastic Volatility", Journal of Finance 48, 881-910.
- [5] Andersen, T.G. (1992), "Volatility", Discussion paper, Northwestern University.
- [6] Andersen, T.G. (1994). "Stochastic Autoregressive Volatility: A Framework for Volatility Modeling", Mathematical Finance 4, 75-102.
- [7] Andersen, T.G. (1996), "Return Volatility and Trading Volume: An Information Flow Interpretation of Stochastic Volatility", Journal of Finance (forthcoming).
- [8] Andersen, T.G. and T. Bollerslev (1995), "Intraday Seasonality and Volatility Persistence in Financial Markets", Journal of Empirical Finance (forthcoming).
- [9] Andersen, T.G. and B. Sørensen (1993), "GMM Estimation of a Stochastic Volatility Model: A Monte Carlo Study", Journal of Business and Economic Statistics (forthcoming).
- [10] Andersen, T.G. and B. Sørensen (1996), "GMM and QML Asumptotic Standard Deviations in Stochastic Volatility Models: A Response to Ruiz (1994)", Journal of Econometrics (forthcoming).
- [11] Andrews, D.W.K. (1993), "Exactly Median-Unbiased Estimation of First Order Autoregressive Unit Root Models", Econometrica, 61, 139-165.
- [12] Bachelier, L. (1900): "Théorie de la spéculation", Ann. Sci. Ecole Norm. Sup. 17, 21-86, [On the Random Character of Stock Market Prices (Paul H. Cootner, ed.) The MIT Press, Cambridge, Mass. 1964].
- [13] Baillie, R.T. and T. Bollerslev (1989), "The Message in Daily Exchange Rates: A Conditional Variance Tale", Journal of Business and Economic Statistics 7, 297-305.
- [14] Baillie, R.T. and T. Bollerslev (1991), "Intra Day and Inter Day Volatility in Foreign Exchange Rates", Review of Economic Studies 58, 565-585.
- [15] Baillie, R.T., T. Bollerslev and H. O. Mikkelsen (1993), "Fractionally integrated generalized autore-gressive conditional heteroekedasticity", Journal of Econometrics (forthcoming).

- [16] Bajeux, I. and J.C. Rochet (1992), "Dynamic Spanning: Are Options an Appropriate Instrument?", Mathematical Finance, (forthcoming).
- [17] Bates, D.S. (1995a), "Testing Option Pricing Models", in G.S. Maddala (ed.), Handbook of Statistics, Vol. 14, Statistical Methods in Finance (North Holland, Amsterdam), (forthcoming).
- [18] Bates, D.S. (1995b), "Jumps and stochastic volatility: exchange rate processes implicit in PHLX Deutschemark options", Review of Financial Studies (forthcoming).
- [19] Beckers, S. (1981), "Standard deviations implied in option prices as predictors of future stock price variability", Journal of Banking and Finance 5, 363-381.
- [20] Bera, A.K. and M.L. Higgins (1995), "On ARCH models: properties, estimation and testing", In L. Exley, D.A.R. George, C.J. Roberts and S. Sawyer (eds.), Surveys in Econometrics. (Basil Blackwell: Oxford) Reprinted from Journal of Economic Surveys.
- [21] Black, F. (1976), "Studies in Stock Price Volatility Changes", Proceedings of the 1976 Business Meeting of the Business and Economic Statistics Section, American Statistical Association, 177-181.
- [22] Black, F. and M. Scholes (1973), "The Pricing of Options and Corporate Liabilities", Journal of Political Economy 81, 637-654.
- [23] Bollerslev, T. (1986), "Generalized Autoregressive Conditional Heteroskedasticity", Journal of Econometrics 31, 307-327.
- [24] Bollerslev, T., Y.C. Chou and K. Kroner (1992), "ARCH Modelling in Finance: A Selective Review of the Theory and Empirical Evidence", Journal of Econometrics 52, 201-224.
- [25] Bollerslev, T. and R. Engle (1993), "Common Persistence in Conditional Variances", Econometrica 61, 166-187.
- [26] Bollerslev, T., R. Engle and D. Nelson (1994), "ARCH Models", in R.F. Engle and D. McFadden (eds.), Handbook of Econometrics, Volume IV, (North-Holland, Amsterdam).
- [27] Bollerslev, T., R. Engle and J. Wooldridge (1988), "A Capital Asset Pricing Model with Time Varying Covariances", Journal of Political Economy 96, 116-131.
- [28] Bollerslev, T. and E. Ghysels (1994), "On Periodic Autoregression Conditional Heteroskedasticity", Journal of Business and Economic Statistics (forthcoming).
- [29] Bollerslev, T. and H. O. Mikkelsen (1995), "Modeling and Pricing Long-Memory in Stock Market volatility", Journal of Econometrics (forthcoming).
- [30] Bossaerts, P., C. Hafner and W. Härdle (1995), "Foreign Exchange Rates Have Surprising Volatility", Discussion Paper, CentER, University of Tilburg.
- [31] Bossaerts, P. and P. Hillion (1995), "Local Parametric Analysis of Hedging in Discrete Time", Journal of Econometrics (forthcoming).

- [32] Breidt, F. J., N. Crato and P. de Lima (1993), "Modeling Long-Memory Stochastic Volatility", Discussion paper, lowa State University.
- [33] Breidt, F. J. and A.L. Carriquiry (1995), "Improved Quasi-Maximum Likelihood Estimation for Stochastic Volatility Models". Mimeo, Department of Statistics, University of Iowa.
- [34] Broadie, M., J. Detemple, E. Ghysels and O. Torrès (1995), "American Options with Stochastic Volatility: A Nonparametric Approach", Discussion Paper, CIRANO.
- [35] Broze, L., O. Scaillet and J.M. Zakoian (1994), "Quasi Indirect Inference for Diffusion Processes", Discussion Paper CORE.
- [36] Broze, L., O. Scaillet and J.M. Zakoian (1995), "Testing for continuous time models of the short term interest rate", Journal of Empirical Finance, 199-223.
- [37] Campa, J. M. and P.H.K. Chang (1995), "Testing the expectations hypothesis on the term structure of implied volatilities in foreign exchange options", Journal of Finance 50, (forthcoming).
- [38] Campbell, J.Y. and A.S. Kyle (1993), "Smart Money, Noise Trading and Stock Price Behaviour", Review of Economics Studies 60, 1-34.
- [39] Canina, L. and S. Figlewski (1993), "The informational content of implied volatility", Review of Financial Studies 6, 659-682.
- [40] Canova, F. (1992), "Detrending and Business Cycle Facts", Discussion Paper, European University Institute, Florence.
- [41] Chesney, M. and L. Scott (1989), "Pricing European Currency Options: A comparison of the Modified Black-Scholes Model and a Random Variance Model", Journal of Financial and Quantitative Analysis 24, 267-284.
- [42] Cheung, Y.-W. and F.X. Diebold (1994), "On Maximum Likelihood Estimation of the Differenciary Parameter of Fractionally - Integrated Noise with Unknown Mean", Journal of Econometrics 62, 301-316.
- [43] Chiras, D.P. and S. Manaster (1978), "The information content of option prices and a test of market efficiency", Journal of Financial Economics 6, 213-234.
- [44] Christensen, B.J. (1992), "Asset Prices and the Empirical Martingale Model", Discussion Paper, New York University.
- [45] Christie, A. A. (1982), "The Stochastic Behavior of Common Stock Variances: Value, Leverage, and Interest Rate Effects", Journal of Financial Economics 10, 407-432.
- [46] Clark, P.K. (1973), "A Subordinated Stochastic Process Model with Finite Variance for Speculative Prices", Econometrica 41, 135-156.
- [47] Clewlow, L and X. Xu (1993), "The Dynamics of Stochastic Volatility", Discussion Paper, University of Warwick.

- [48] Comte, F. and E. Renault (1993), "Long memory continuous time models", Journal of Econometrics (forthcoming).
- [49] Comte, F. and E. Renault (1995), "Long memory continuous time Stochastic Volatility Models", Paper presented at the HFDF-1 Conference, Zürich.
- [50] Conley, T., L.P. Hansen, E. Luttmer and J. Scheinkman (1995), "Estimating Subordinated Diffusions from discrete Time Data" Discussion paper, University of Chicago.
- [51] Cornell, B. (1978), "Using the Options Pricing Model to Measure the Uncertainty Producing Effect of Major announcements", Financial Management 7, 54-59.
- [52] Cox, J.C. (1975), "Notes on Option Pricing I: Constant Elasticity of Variance Diffusions", Discussion Paper, Stanford University.
- [53] Cox, J.C. and S. Ross (1976), "The Valuation of Options for Alternative Stochastic Processes", Journal of Financial Economics 3, 145-166.
- [54] Cox, J.C. and M. Rubinstein (1985), Options Markets, (Englewood Cliffs, Prentice-Hall, New Jersey).
- [55] Dacorogna, M.M., U.A. Müller, R.J. Nagler, R.B. Olsen and O.V. Pictet (1993), "A Geographical Model for the Daily and Weekly Seasonal Volatility in the Foreign Exchange Market", Journal of International Money and Finance 12, 413-438.
- [56] Danielsson, J. (1994), "Stochastic Volatility in asset Prices: Estimation with Simulated Maximum Likelihood", Journal of Econometrics 61, 375-400.
- [57] Danielsson, J. and J.F. Richard (1993), "Accelerated Gaussian Importance Sampler with Application to Dynamic Latent Variable Models", Journal of Applied Econometrics 3, S153-S174.
- [58] Dassios, A. (1995), "Asymptotic expressions for approximations to stochastic variance models". mimeo, London School of Economics.
- [59] Day, T.E. and C.M. Lewis (1988), "The behavior of the volatility implicit in the prices of stock index options", Journal of Financial Economics 22, 103-122.
- [60] Day, T.E. and C.M. Lewis (1992), "Stock market volatility and the information content of stock index options", Journal of Econometrics 52, 267-287.
- [61] Diebold, F.X. (1988), Empirical Modeling of Exchange Rate Dynamics, (Springer Verlag, New York).
- [62] Diebold, F.X. and J.A. Lopez (1995), "Modeling Volatility Dynamics", in K. Hoover (ed), Macroe-conomics: Developments, Tensions and Prospects.
- [63] Diebold, F.X. and M. Nerlove (1989), "The Dynamics of Exchange Rate Volatility: A Multivariate Latent Factor ARCH Model", Journal of Applied Econometrics 4, 1-22.
- [64] Ding, Z., C.W.J. Granger and R.F. Engle (1993), "A Long Memory Property of Stock Market Returns and a New Model", Journal of Empirical Finance 1, 83-108.

- [65] Diz, F. and T.J. Finucane (1993), "Do the options markets really overreact?", Journal of Futures Markets 13, 298-312.
- [66] Drost, F.C. and T. E. Nijman (1993), "Temporal Aggregation of GARCH Processes", Econometrica 61, 909-927.
- [67] Drost, F.C. and B.J.M. Werker (1994), "Closing the GARCH Gap: Continuous Time GARCH Modelling", Discussion Paper CentER, University of Tilburg.
- [68] Duan, J.C. (1994), "Maximum Likelihood Estimation Using Price Data of the Derivative Contract", Mathematical Finance 4, 155-167.
- [69] Duan, J.C. (1995), "The GARCH Option Pricing Model", Mathematical Finance 5, 13-32.
- [70] Duffie, D. (1989), Futures Markets, (Prentice-Hall International Editions).
- [71] Duffie, D. (1992), Dynamic Asset Pricing Theory, (Princeton University Press).
- [72] Duffie, D. and K. J. Singleton (1993), "Simulated Moments Estimation of Markov Models of Asset Prices", Econometrica 61, 929-952.
- [73] Dunsmuir, W. (1979), "A central limit theorem for parameter estimation in stationary vector time series and its applications to models for a signal observed with noise", Annuals of Statistics 7, 490-506.
- [74] Easley, D. and M. O'Hara (1992), "Time and the Process of Security Price Adjustment", Journal of Finance, 47, 577-605.
- [75] Ederington, L.H. and J.H. Lee (1993), "How markets process information: news releases and volatility", Journal of Finance 48, 1161-1192.
- [76] Elsheimer, B., M. Fisher, D. Nychka and D. Zirvos (1995), "Smoothing Splines Estimates of the Discount Function based on US Bond Prices", Discussion Paper Federal Reserve, Washington, D.C.
- [77] Engle, R.F. (1982), "Autoregressive Conditional Heteroskedasticity with Estimates of the Variance of United Kingdom Inflation", Econometrica 50, 987-1007.
- [78] Engle, R.F. and C.W.J. Granger (1987), "Co-Integration and Error Correction: Representation, Estimation and Testing", Econometrica 55, 251-576.
- [79] Engle, R.F. and S. Kozicki (1993), "Testing for Common Features", Journal of Business and Economic Statistics 11, 369-379.
- [80] Engle, R.F. and G.G.J Lee (1994), "Estimating diffusion models of stochastic volatility", Discussion Paper, University of California at San Diego.
- [81] Engle, R.F. and C. Mustafa (1992), "Implied ARCH models from option prices", Journal of Econometrics 52, 289-311.
- [82] Engle, R.F. and V.K. Ng (1993), "Measuring and Testing the Impact of News on Volatility", Journal of Finance 48, 1749-1801.

- [83] Fama, E.F. (1963), "Mandelbrot and The Stable Paretian Distribution", Journal of Business 36, 420-429.
- [84] Fama, E.F. (1965), "The Behavior of Stock Market Prices", Journal of Business 38, 34-105.
- [85] Foster, D. and S. Viswanathan (1993a), "The Effect of Public Information and Competition on Trading Volume and Price Volatility", Review of Financial Studies 6, 23-56.
- [86] Foster, D. and S. Viswanathan (1993b), "Can Speculative Trading Explain the Volume Volatility Relation", Discussion paper, Fuqua School of Business, Duke University.
- [87] French, K. and R. Roll (1986), "Stock Return Variances: The Arrival of Information and the Reaction of Traders", Journal of Financial Economics 17, 5-26.
- [88] Gallant, A.R., D.A. Hsieh and G. Tauchen (1994), "Estimation of Stochastic Volatility Models with Suggestive Diagnostics", Discussion paper, Duke University.
- [89] Gallant, A.R., P.E. Rossi and G. Tauchen (1992), "Stock Prices and Volume", Review of Financial Studies, 5, 199-242.
- [90] Gallant, A.R., P.E. Rossi and G. Tauchen (1993), "Nonlinear Dynamic Structures", Econometrica 61, 871-907.
- [91] Gallant, A.R. and G. Tauchen (1989), "Semiparametric Estimation of Conditionally Constrained Heterogeneous Processes: Asset Pricing applications", Econometrica 57, 1091-1120.
- [92] Gallant, A.R. and G. Tauchen (1992), "A Nonparametric Approach to Nonlinear Time Series Analysis: Estimation and Simulation", in E. Parzen, D. Brillinger, M. Rosenblatt, M. Taqqu, J. Geweke and P. Caines (eds.), New Dimensions in Time Series Analysis, Springer-Verlag, New York.
- [93] Gallant, A.R. and G. Tauchen (1994), "Which Moments to Match", Econometric Theory (forthcoming).
- [94] Gallant, A.R. and G. Tauchen (1995), "Estimation of Continuous Time Models for Stock Returns and Interest Rates", Discussion Paper, Duke University.
- [95] Gallant, A.R. and H. White (1988), A Unified Theory of Estimation and Inference for Nonlinear Dynamic Models, (Basil Blackwell, Oxford).
- [96] Garcia, R. and E. Renault (1995), "Risk Aversion, Intertemporal Substitution and Option Pricing", Discussion Paper CIRANO.
- [97] Geweke, J. (1994), "Comment on Jacquier, Polson and Rossi", Journal of Business and Economics Statistics 12, 397-399.
- [98] Geweke, J. (1995), "Monte Carlo Simulation and Numerical Integration", in H. Amman, D. Kendrick and J. Rust (ed.) Handbook of Computational Economics (North Holland).
- [99] Ghysels, E., C. Gouriéroux and J. Jasiak (1995a), "Market Time and Asset Price Movements: Theory and Estimation", Discussion paper CIRANO and C.R.D.E., Université de Montréal.

- [100] Ghysels, E., C. Gouriéroux and J. Jasiak (1995b), "Trading Patterns, Time Deformation and Stochastic Volatility in Foreign Exchange Markets", Paper presented at the HFDF Conference, Zürich.
- [101] Ghysels, E. and J. Jasiak (1994a), "Comments on Bayesian Analysis of Stochastic Volatility Models", Journal of Business and Economic Statistics 12, 399-401.
- [102] Ghysels, E. and J. Jasiak (1994b), "Stochastic Volatility and Time Deformation An Application of Trading Volume and Leverage Effects", Paper presented at the Western Finance Association Meetings, Santa Fe.
- [103] Ghysels, E., L. Khalaf and C. Vodounou (1994), "Simulation Based Inference in Moving Average Models", Discussion Paper, CIRANO and C.R.D.E.
- [104] Ghysels, E., H.S. Lee and P. Siklos (1993), "On the (Mis)Specification of Seasonality and its Consequences: An Empirical Investigation with U.S. Data", Empirical Economics 18, 747-760.
- [105] Goodhart, C.A.E. and M. O'Hara (1995), "High Frequency Data in Financial Markets: Issues and Applications", Paper presented at HFDF Conference, Zürich.
- [106] Gouriéroux, C. and A. Monfort (1993a), "Simulation Based Inference: A Survey with Special Reference to Panel Data Models", Journal of Econometrics 59, 5-33.
- [107] Gouriéroux, C. and A. Monfort (1993b), "Pseudo-Likelihood Methods" in Maddala et al. (ed.) Handbook of Statistics Vol. 11, (North Holland, Amsterdam).
- [108] Gouriéroux, C. and A. Monfort (1994), "Indirect Inference for Stochastic Differential Equations", Discussion Paper CREST, Paris.
- [109] Gouriéroux, C. and A. Monfort (1995), Simulation-Based Econometric Methods, (CORE Lecture Series, Louvain-la-Neuve).
- [110] Gouriéroux, C., A. Monfort and E. Renault (1993), "Indirect Inference", Journal of Applied Econometrics 8, S85-S118.
- [111] Gouriéroux, C., A. Monfort and C. Tenreiro (1994), "Kernel M-Estimators: Nonparametric Diagnostics for Structural Models", Discussion Paper, CEPREMAP.
- [112] Gouriéroux, C., A. Monfort and C. Tenreiro (1995), "Kernel M-Estimators and Functional Residual Plots", Discussion Paper CREST - ENSAE, Paris.
- [113] Gouriéroux, C., E. Renault and N. Touzi (1994), "Calibration by Simulation for Small Sample Bias Correction", Discussion Paper CREST.
- [114] Gouriéroux, C. and O. Scaillet (1994), "Estimation of the Term Structure from Bond Data", Journal of Empirical Finance, forthcoming.
- [115] Granger, C.W.J. and Z. Ding (1994), "Stylized Facts on the Temporal and distributional Properties of Daily Data for Speculative Markets", Discussion Paper, University of California, San Diego.

- [116] Hall, A.R. (1993), "Some Aspects of Generalized Method of Moments Estimation" in Maddala et al. (ed.) Handbook of Statistics Vol. 11, (North Holland, Amsterdam).
- [117] Hamao, Y., R.W. Masulis and V.K. Ng (1990), "Correlations in Price Changes and Volatility Across International Stock Markets", Review of Financial Studies 3, 281-307.
- [118] Hansen, B.E. (1995), "Regression with Nonstationary Volatility", Econometrica 63, 1113-1132.
- [119] Hansen, L.P. (1982), "Large Sample Properties of Generalized Method of Momerata Estimators", Econometrica 50, 1029-1054.
- [120] Hansen, L.P. and J.A. Scheinkman (1995), "Back to the Future: Generating Moment Implications for Continuous-Time Markov Processes", Econometrica 63, 767-804.
- [121] Harris, L. (1986), "A Transaction Data Study of Weekly and Intradaily Patterns in Stock Returns", Journal of Financial Economics 16, 99-117.
- [122] Harrison, M. and D. Kreps (1979), "Martingale and Arbitrage in Multiperiod Securities Markets", Journal of Economic Theory 20, 381-408.
- [123] Harrison, J.M. and S. Pliska (1981), "Martingales and Stochastic Ingegrals in the Theory of Continuous Trading", Stochastic Processes and Their Applications 11, 215-260.
- [124] Harrison, P.J. and C.F. Stevens (1976), "Bayesian Forecasting" (with discussion), J. Royal Statistical Society, Ser. B, 38, 205-247.
- [125] Harvey, A.C. (1989), Forecasting, Structural Time Series Models and the Kalman Filter, (Cambridge University Press).
- [126] Harvey, A.C. and A. Jaeger (1993), "Detrending Stylized Facts and the Business Cycle", Journal of Applied Econometrics 8, 231-247.
- [127] Harvey, A.C. (1993), "Long Memory in Stochastic Volatility", Discussion Paper, London School of Economics.
- [128] Harvey, A.C. and S.J. Koopman (1993), "Forecasting Hourly Electricity Demand using Time-Varying Splines", Journal of the American Statistical Association 88, 1228-1236.
- [129] Harvey, A.C., E. Ruiz and E. Sentana (1992), "Unobserved Component Time Series Models with ARCH Disturbances", Journal of Econometrics 52, 129-158.
- [130] Harvey, A.C., E. Ruiz and N. Shephard (1994), "Multivariate Stochastic Variance Models", Review of Economic Studies 61, 247-264.
- [131] Harvey, A.C. and N. Shephard (1993), "Estimation of Testing of Stochastic Variance Models", STICERD Econometrics, Discussion paper, EM93/268, London School of Economics.
- [132] Harvey, A.C. and N. Shephard (1996), "Estimation of an Asymmetric Stochastic Volatility Model for Asset Returns", Journal of Business and Economic Statistics (forthcoming).

- [133] Harvey, C.R. and R.D. Huang (1991), "Volatility in the Foreign Currency Futures Market", Review of Financial Studies 4, 543-569.
- [134] Harvey, C.R. and R.D. Huang (1992), "Information Trading and fixed Income Volatility, Discussion Paper, Duke University.
- [135] Harvey, C.R. and R.E. Whaley (1992), "Market volatility prediction and the efficiency of the S&P 100 index option market", Journal of Financial Economics 31, 43-74.
- [136] Hausman, J.A. and A. W. Lo (1991), "An Ordered Probit Analysis of Transaction Stock Prices", Discussion paper, Wharton School, University of Pennsylvania.
- [137] He, H. (1993), "Option Prices with Stochastic Volatilities: An Equilibrium Analysis", Discussion Paper, University of California, Berkeley.
- [138] Heston, S.L. (1993), "A Closed-Form Solution for Options with Stochastic Volatility with Applications to Bond and Currency Options", Review of Financial Studies 6, 327-343.
- [139] Heynen, R., A. Kernna and T. Vorst (1994), "Analysis of the term structure of implied volatility, Journal of Financial Quantitative Analysis.
- [140] Hull, J. (1993), Options, futures and other derivative securities 2nd ed., (Prentice-Hall International Editions, New Jersey).
- [141] Hull, J. (1995), Introduction to Futures and Options Markets, 2nd ed., (Prentice-Hall, Englewood Cliffs, New Jersey).
- [142] Hull, J. and A. White (1987), "The Pricing of Options on Assets with Stochastic Volatilities", Journal of Finance 42, 281-300.
- [143] Huffman, G.W. (1987), "A Dynamic Equilibrium Model of Asset Prices and Transactions Volume", Journal of Political Economy 95, 138-159.
- [144] Hutchinson, J.M., A.W. Lo and T. Poggio (1994), "A Nonparametric Approach to Pricing and Hedging Derivative Securities via Learning Networks", Journal of Finance 49, 851-890.
- [145] Jacquier, E., N.G. Poison and P.E. Rossi (1994), "Bayesian Analysis of Stochastic Volatility Models" (with discussion), Journal of Business and Economic Statistics 12, 371-417.
- [146] Jacquier, E., N.G. Polson and P.E. Rossi (1995a), "Multivariate and Prior Distributions for Stochastic Volatility Models", Discussion Paper CIRANO.
- [147] Jacquier, E., N.G. Polson and P.E. Rossi (1995b), "Stochastic Volatility: Univariate and Multivariate Extensions", Rodney White Center for Financial Research Working Paper 19-95, The Wharton School, University of Pennsylvania.
- [148] Jacquier, E., N.G. Polson and P.E. Rossi (1995c), "Efficient Option Pricing under Stochastic Volatility", Manuscript, The Wharton School, University of Pennsylvania.
- [149] Jarrow, R. and Rudd (1983), Option Pricing, (Irwin, Homewood III).

- [150] Johnson, H. and D. Shanno (1987), "Option Pricing when the Variance is Changing", Journal of Financial and Quantitative Analysis 22, 143-152.
- [151] Jorion, P. (1995), "Predicting volatility in the foreign exchange market", Journal of Finance 50, (forthcoming).
- [152] Karatzas, I. and S.E. Shreve (1988), Brownian Motion and Stochastic Calculus. Springer-Verlag. New York, NY.
- [153] Karpoff, J. (1987), "The Relation between Price Changes and Trading Volume: A Survey", Journal of Financial and Quantitative Analysis 22, 109-126.
- [154] Kim, S. and N. Shephard (1994), "Stochastic Volatility: Optimal Likelihood Inference and Comparison with ARCH Model", Discussion Paper, Nuffield College, Oxford.
- [155] King, M., E. Sentana and S. Wadhwani (1994), "Volatility and Links Between National Stock Markets", Econometrica 62, 901-934.
- [156] Kitagawa, G. (1987), "Non-Gaussian State Space Modeling of Nonstationary Time Series" (with discussion), Journal of the American Statistical Association 79, 378-389.
- [157] Kloeden, P.E. and E. Platten (1992), Numerical Solutions of Stochastic Differential Equations (Springer-Verlag, Heidelberg).
- [158] Lamoureux, C. and W. Lastrapes (1990), "Heteroskedasticity in Stock Return Data: Volume versus GARCH Effect", Journal of Finance 45, 221-229.
- [159] Lamoureux, C. and W. Lastrapes (1993), "Forecasting stock-return variance: towards an understanding of stochastic implied volatilities", Review of Financial Studies 6, 293-326.
- [160] Latane, H. and R. Jr. Rendleman (1976), "Standard Deviations of Stock Price Ratios Implied in Option Prices", Journal of Finance 31, 369-381.
- [161] Lezan, G., E. Renault and T. deVitry (1995) 'Forecasting Foreign Exchange Risk", Paper presented at 7th World Congres of the Econometric Society, Tokyo.
- [162] Lin, W.L., R.F. Engle and T. Ito (1994), "Do Bulls and Bears Move Across Borders? International Transmission of Stock Returns and Volatility as the World Turns", Review of Financial Studies, forthcoming.
- [163] Lo, A.W. (1995), "Statistical Inference for Technical Analysis Via Nonparametric Estimation", Discussion Paper, MIT.
- [164] Mahieu, R. and P. Schotman (1994a), "Stochastic volatility and the distribution of exchange rate news", Discussion Paper, University of Limburg.
- [165] Mahieu, R. and P. Schotman (1994b), "Neglected Common Factors in Exchange Rate Volatility". Journal of Empirical Finance 1, 279-311.

- [166] Mandelbrot, B.B. (1963), "The Variation of Certain Speculative Prices", Journal of Business 36, 394-416.
- [167] Mandelbrot, B. and H. Taylor (1967), "On the Distribution of Stock Prices Differences", Operations Research 15, 1057-1062.
- [168] Mandelbrot, B.B. and J.W. Van Ness (1968), "Fractal Brownian Motions, Fractional Noises and Applications", SIAM Review 10, 422-437.
- [169] McFadden, D. (1989), "A Method of Simulated Moments for Estimation of Discrete Response Models Without Numerical Integration", Econometrica 57, 1027-1057.
- [170] Meddahi, N. and E. Renault (1995), "Aggregations and Marginalisations of GARCH and Stochastic Volatility Models", Discussion Paper, GREMAQ.
- [171] Melino, A. and M. Turnbull (1990), "Pricing Foreign Currency Options with Stochastic Volatility", Journal of Econometrics 45, 239-265.
- [172] Melino, A. (1994), "Estimation of Continuous Time Models in Finance", in C.A. Sims (eds.) Advances in Econometrics (Cambridge University Press).
- [173] Merton, R.C. (1973), "Rational Theory of Option Pricing", Bell Journal of Economics and Management Science 4, 141-183.
- [174] Merton, R.C. (1976), "Option Pricing when Underlying Stock Returns are Discontinuous", Journal of Financial Economics 3, 125-144.
- [175] Merton, R.C. (1990), Continuous Time Finance, (Basil Blackwell, Oxford).
- [176] Merville, L.J. and D.R. Pieptea (1989), "Stock-price volatility, mean-reverting diffusion, and noise", Journal of Financial Economics 242, 193-214.
- [177] Metropolis, N.,A.W. Rosenbluth, M.N. Rosenbluth, A.H. Teller and E. Teller (1954), "Equation of State Calculations by Fast Computing Machines", The Journal of Chemical Physics 21, 1087-1092.
- [178] Müller, U.A., M.M. Dacorogna, R.B. Olsen, W.V.Pictet, M. Schwarz and C. Morgenegg (1990), "Statistical Study of Foreign Exchange Rates. Empirical Evidence of a Price change Scaling Law and Intraday analysis", Journal of Banking and Finance 14, 1189-1208.
- [179] Nelson, D.B. (1988), "Time Series Behavior of Stock Market Volatility and Returns", Ph.D. dissertation, MIT.
- [180] Nelson, D.B. (1990), "ARCH Models as Diffusion Approximations", Journal of Econometrics 45, 7-39.
- [181] Nelson, D.B. (1991), "Conditional Heteroekedasticity in Asset returns: A New Approach", Econometrica 59, 347-370.
- [182] Nelson, D.B. (1992), "Filtering and Forecasting with Misspecified ARCH Models I: Getting the Right Variance with the Wrong Model", Journal of Econometrics 25, 61-90.

- [183] Nelson, D.B. (1994), "Comment on Jacquier, Polson and Rossi", Journal of Business and Economic Statistics 12, 403-406.
- [184] Nelson, D.B. (1995a), "Asymptotic Smoothing Theory for ARCH Models", Economictrica (forthcoming).
- [185] Nelson, D.B. (1995b), "Asymptotic Filtering Theory for Multivariate ARCH Models", Journal of Econometrics (forthcoming).
- [186] Nelson, D.B. and D.P. Foster (1994), "Asymptotic Filtering Theory for Univariate ARCH Models", Econometrica 62, 1-41.
- [187] Nelson, D.B. and D.P. Foster (1995), "Filtering and Forecasting with Misspecified ARCH Models II: Making the Right Forecast with the Wrong Model", Journal of Econometrics (forthcoming).
- [188] Noh, J., R.F. Engle and A. Kane (1994), "Forecasting volatility and option pricing of the S&P 500 index", Journal of Derivatives, 17-30.
- [189] Ogaki, M. (1993), "Generalized Method of Moments: Econometric Applications", in Maddala et al. (ed.) Handbook of Statistics Vol. 11, (North Holland, Amsterdam).
- [190] Pagan, A.R. and G.W. Schwert (1990), "Alternative Models for Conditional Stock Volatility", Journal of Econometrics 45, 267-290.
- [191] Pakes, A. and D. Pollard (1989), "Simulation and the Asymptotics of Optimization Estimators", Econometrica 57, 995-1026.
- [192] Pardoux, E. and D. Talay (1985), "Discretization and Simulation of Stochastic Differential Equations", Acta Applicandae Mathematica 3, 23-47.
- [193] Pastorello, S., E. Renault and N. Touzi (1993), "Statistical Inference for Random Variance Option Pricing", Discussion Paper, CREST.
- [194] Patell, J.M. and M.A. Wolfson (1981), "The Ex-Ante and Ex-Post Price Effects of quarterly Earnings Announcement Reflected in Option and Stock Price", Journal of Accounting Research 19, 434-458.
- [195] Patell, J.M. and M.A. Wolfson (1979), "Anticipated Information Releases Reflected in Call Option Prices", Journal of Accounting and Economics 1, 117-140.
- [196] Pham, H. and N. Touzi (1993), "Intertemporal Equilibrium Risk Premia in a Stochastic Volatility Model", Mathematical Finance (forthcoming).
- [197] Platten, E. and Schweizer (1995), "On Smile and Skewness", Discussion Paper, Australian National University, Canberra.
- [198] Poterba, J. and L. Summers (1986), "The persistence of volatility and stock market fluctuations", American Economic Review 76, 1142-1151.
- [199] Renault, E. (1995), "Econometric Models of Option Pricing Errors", Invited Lecture presented at 7th W.C.E.S., Tokyo, August.

- [200] Renault, E. and N. Touzi (1992), "Option Hedging and Implicit Volatility", Mathematical Finance (forthcoming).
- [201] Revuz, A. and M. Yor (1991), Continuous Martingales and Brownian Motion (Springer Verlag, Berlin).
- [202] Robinson, P. (1993), "Efficient tests of nonstationary hypotheses", mimeo, London School of Economics.
- [203] Rogers, L.C.G. (1995), "Arbitrage with Fractional Brownian Motion", University of Bath. Discussion paper.
- [204] Rubinstein, M. (1985), "Nonparametric Tests of Alternative Option Pricing Models Using all Reported Trades and Quotes on the 30 Most Active CBOE Option Classes from August 23, 1976 through August 31, 1978", Journal of Finance 40, 455-480.
- [205] Ruiz, E. (1994), "Quasi-maximum Likelihood Estimation of Stochastic Volatility Models", Journal of Econometrics 63, 289-306.
- [206] Schwert, G.W. (1989), "Business Cycles, Financial Crises, and Stock Volatility", Carnegie-Rochester Conference Series on Public Policy 39, 83-126.
- [207] Scott, L.O. (1987), "Option Pricing when the Variance Changes Randomly: Theory. Estimation and an Application", Journal of Financial and Quantitative Analysis 22, 419-438.
- [208] Scott, L. (1991), "Random Variance Option Pricing", Advances in Futures and Options Research, Vol. 5, 113-135.
- [209] Sheikh, A.M. (1993), "The behavior of volatility expectations and their effects on expected returns", Journal of Business 66, 93-116.
- [210] Shephard, N. (1995), "Statistical Aspect of ARCH and Stochastic volatility", Discussion Paper 1994, Nuffield College, Oxford University.
- [211] Sims, A. (1984), "Martingale-Like Behavior of Prices", University of Minnesota.
- [212] Sowell, F. (1992), "Maximum likelihood estimation of stationary univariate fractionally integrated time series models", Journal of Econometrics 53, 165-188.
- [213] Stein, J. (1989): "Overreactions in the Options Market", The Journal of Finance 44, 1011-1023.
- [214] Stein, E.M. and J. Stein (1991), "Stock Price distributions with Stochastic Volatility: An Analytic Approach", Review of Financial Studies 4, 727-752.
- [215] Stock, J.H. (1988), "Estimating Continuous Time Processes Subject to Time Deformation", Journal of the American Statistical Association 83, 77-84.
- [216] Strook, D.W. and S.R.S. Varadhan (1979), Multi-dimensional Diffusion Processes, (Springer Verlag, Heidelberg).

- [217] Tanner, T. and W. Wong (1987), "The Calculation of Posterior Distributions by Data Augmentation", Journal of the American Statistical Association 82, 528-549.
- [218] Tauchen, G. (1995), "New Minimum Chi-Square Methods in Empirical Finance", Invited Paper presented at the 7th World Congress of the Econometric Society, Tokyo.
- [219] Tauchen, G. and M. Pitts (1983), "The Price Variability-Volume Relationship on Speculative Markets", Econometrica 51, 485-505.
- [220] Taylor, S.J. (1986), Modeling Financial Time Series, (John Wiley: Chichester).
- [221] Taylor, S.J. (1994), "Modeling Stochastic Volatility: A Review and Comparative Study", Mathematical Finance 4, 183-204.
- [222] Taylor, S.J. and X. Xu (1994), "The Term structure of volatility implied by foreign exchange options", Journal of Pinancial and Quantitative Analysis 29, 57-74.
- [223] Taylor, S.J. and X. Xu (1993), "The Magnitude of Implied Volatility Smiles: Theory and Empirical Evidence for Exchange Rates", Discussion Paper, University of Warwick.
- [224] Von Furstenberg, G.M. and B. Nam Jeon (1989), "International Stock Price Movements: Links and Messages", Brookings Papers on Economic Activity I, 125-180.
- [225] Wang, J. (1993), "A Model of Competitive Stock Trading Volume", Discussion Paper, MIT.
- [226] Watanabe, T. (1993), "The Time Series Properties of Returns, Volatility and Trading Volume in Financial Markets", Ph.D. Thesis, Department of Economics, Yale University.
- [227] West, M. and J. Harrison (1990), "Bayesian Forecasting and Dynamic Models", (Springer Verlag, Berlin).
- [228] Whaley, R.E. (1982), "Valuation of American call options on dividend-paying stocks. Journal of Financial Economics 10, 29-58.
- [229] Wiggins, J.B. (1987), "Option Values under Stochastic Volatility: Theory and Empirical Estimates", Journal of Financial Economics 19, 351-372.
- [230] Wood, R.T. McInish and J.K. Ord (1985), "An Investigation of Transaction Data for NYSE Stocks", Journal of Finance 40, 723-739.
- [231] Wooldridge, J.M. (1994), "Estimation and Inference for Dependent Processes" in R.F. Engle and D. McFadden (ed.) Handbook of Econometrics Vol. 4, (North Holland, Amsterdam).

Université de Montréal Département de sciences économiques Centre de documentation C.P. 6128, succursale Centre-ville Montréal (Québec) H3C 3J7

Cahiers de recherche (Discussion papers) 1995 à aujourd'hui (1995 to date)

Si vous désirez obtenir un exemplaire, vous n'avez qu'à faire parvenir votre demande et votre paiement (5 \$ l'unité) à l'adresse ci-haut mentionnée. I To obtain a copy (\$ 5 each), please send your request and prepayment to the above-mentioned address.

- 9501: Boyer, Marcel et Jean-Jacques Laffont, "Environmental Risks and Bank Liability", janvier 1995, 46 pages.
- 9502: Margolia, David. N., "Firm Heterogeneity and Worker Self-Selection Bias Estimated Returns to Seniority", décembre 1994, 29 pages.
- 9503: Abowd, John M., Francis Kramarz et David N. Margolis, "High-Wage Workers and High-Wage Firms", janvier 1995, 73 pages
- 9504: Cardia, Emanuela et Steve Ambler, "Indexation Lags and Heterodox Stabilization Programs", janvier 1995, 29 pages.
- 9505 : Garcia, René et Huntley Schaller, "Are the Effects of Monetary Policy Asymmetric?", février 1995, 42 pages.
- 9506 : Parent, Daniel, "Survol des contributions théoriques et empiriques liées au capital humain", février 1995, 70 pages.
- 9507: Parent, Daniel, "Wages and Mobility: The Impact of Employer-Provided Training", février 1995, 34 pages.
- 9508: Parent, Daniel, "Industry-Specific Capital and the Wage Profile: Evidence from the NLSY and the PSID", février 1995, 21 pages.
- 9509: Parent, Daniel, "Matching, Human Capital, and the Covariance Structure of Earnings", février 1995, 54 pages.
- 9510: Garcia, René, "Asymptotic Null Distribution of the Likelihood Ratio Test in Markov Switching Models", mars 1995, 50 pages.
- 9511: Garcia, René, Annamaria Lusardi et Serena Ng, "Excess Sensitivity and Asymmetries in Consumption: An Empirical Investigation", mars 1995, 26 pages.
- 9512 : Sprumont, Yves, "An Axiomatization of the Pazner-Schmeidler Rules in Large Fair Division Problems", mars 1995, 26 pages.
- 9513: Ghysels, Eric, Lynda Khalaf et Cosmé Vodounou, "Simulation Based Inference in Moving Average Models", mars 1995, 10 pages.
- 9514: Ng, Serena, "Looking for Evidence of Speculative Stockholding in Commodity Markets", mars 1995, 25 pages.
- 9515: Ng, Screna et Huntley Schaller, "The Risky Spread, Investment, and Monetary Policy Transmission: Evidence on the Role of Asymmetric Information", mars 1995, 26 pages.

- 9516: Ng, Serena, "Testing for Homogeneity in Demand Systems when the Regressors are Non-Stationary", mars 1995, 26 pages.
- 9517: Ghysels, Eric, Clive W.J. Granger et Pierre L. Siklos, "Is Seasonal Adjustment a Linear or Nonlinear Data Filtering Process?", mars 1995, 34 pages.
- 9518: Ghysels, Eric, Alastair Hall et Hahn S. Lee, "On Periodic Structures and Testing for Seasonal Unit Roots", mars 1995, 45 pages.
- 9519: Sprumont, Yves, "On the Game-Theoretic Structure of Public-Good Economies", mars 1995, 21 pages.
- 9520: Charles, Sandra, François Vaillancourt et Nicolas Marceau, "The Impact of Decentralization on Growth and Democracy: A Note", mars 1995, 13 pages.
- 9521: Sprumont, Yves, "Balanced Egalitarian Redistribution of Income", mars 1995, 17 pages.
- 9522 Bronsard, Camille, Lise Salvas-Bronsard et Alain Trognon, "On the Residual Dynamics Implied by the Rational Expectations Hypothesis", mars 1995, 18 pages.
 9523 Campbell, Bryan et Eric Ghysels, "An Empirical Analysis of the Canadian Budget
- 9523: Campbell, Bryan et Eric Ghysels, "An Empirical Analysis of the Canadian Budget Process", mars 1995, 30 pages.
 9524: Ghysels, Eric, Alain Guay et Alastair Hall, "Predictive Tests for Structural Change with
- 9524: Ghysels, Eric, Alain Guay et Alastair Hall, "Predictive Tests for Structural Change with Unknown Breakpoint", avril 1995, 29 pages.
- 9525: Ghysels, Eric, "On Stable Factor Structures in the Pricing of Risk", avril 1995, 37 pages.
- 9526: Kollmann, Robert, "Mark Up Fluctuations in U.S. Manufacturing and Trade: Empirical Evidence Based on A Model of Optimal Storage", avril 1995, 18 pages.
- 9527: Laberge-Nadeau, Claire, Georges Dionne, Urs Maag, Denise Desjardins, Charles Vanasse et J.-M. Ékoé, "Medical Conditions and the Severity of Commercial Motor Vehicle (CMV) Drivers' Road Accidents", mai 1995, 26 pages.
- 9528: Dionne, Georges, Manuel Artis et Montserrat Guillén, "On the Repayment of Personal Loans Under Asymmetrical Information: A Count Data Model Approach", mai 1995, 32 pages.
- 9529: Ruge-Murcia, Francisco J., "Government Expenditure and the Dynamics of High Inflation", juin 1995, 45 pages.
- 9530 Pesaran, M. Hashem et Francisco J. Ruge-Murcia, "A Discrete-Time Version of Target Zone Models with Jumps", juin 1995, 50 pages.
- 9531: Dionne, Georges et N. Fombaron, "Non-Convexities and Efficiency of Equilibria in Insurance Markets with Asymmetric Information", juin 1995, 22 pages.
- 9532: Bronsard, C., P. Michel et L. Salvas-Bronsard, "Singular Demand Systems as an Instrument of Scientific Discovery" willet 1995, 17 pages
- Instrument of Scientific Discovery", juillet 1995, 17 pages.

 9533: Dionne, G. et C. Fluet, "Incentives in Multi-Period Regulation and Procurement: A Graphical Analysis", juillet 1995, 46 pages.
- 9534: Ng. Serena et Pierre Perron, "Estimation and Inference in Nearly Unbalanced, Nearly Cointegrated Systems", août 1995, 38 pages.
- 9535: Ng, Serena et Pierre Perron, "The Exact Error in Estimating the Spectral Density at the Origin", août 1995, 33 pages.
- 9536: Ghysels, Eric, Christian Gouriéroux et Joanna Jasiak, "Market Tirne and Asset Price Movements: Theory and Estimation", septembre 1995, 61 pages.
- 9537: Dionne, Georges, Robert Gagné et Charles Vanasse, "Inferring Technological Parameters from Incomplete Panel Data", septembre 1995, 40 pages.
- 9538: Dufour, Jean-Marie et Eric Renault, "Short-Run and Long-Run Causality in Time Series: Theory", septembre 1995, 40 pages.

- 9539: Dufour, Jean-Marie, "Some Impossibility Theorems in Econometrics with Applications to Instrumental Variables, Dynamic Models and Cointegration", septembre 1995, 42 pages.
- 9540: Mercenier, Jean et Philippe Michel, "Temporal Aggregation in a Multi-Sector Economy with Endogenous Growth", septembre 1995, 17 pages.
- 9541: Loranger, Jean-Guy, "Neoliberalism and the Overwhelming Influence of Financial Markets: a Comparative Analysis between NAFTA Countries and Other G-7
- Countries", séptembre 1995, 34 pages.

 9542 : Proulx, Pierre-Paul, "La décentralisation : facteur de développement ou d'éclatement du Québec", août 1995, 26 pages.
- 9543: Arcand, Jean-Louis, Ghazi Boulila et Christian Tritten, "Intergenerational Contracts, Remittances, and Growth", octobre 1995, 57 pages.
- 9544: Kichian, Maral, René Garcia et Eric Ghysels, "On the Dynamic Specification of International Asset Pricing Models", octobre 1995, 29 pages.
- 9545: Arcand, Jean-Louis, "Land Ownership, Working Capital, and Agricultural Output: Egypt, 1913-1958", octobre 1995, 93 pages.
- 9546: Arcand, Jean-Louis et Marcel G. Dagenais, "The Empirics of Economic Growth in a Cross Section of Countries: Do Errors in Variables Really Not Matter?", octobre 1995, 65 pages.
- 9547: Dufour, Jean-Marie et Jan F. Kiviet, "Exact Inference Methods for First-Order Autoregressive Distributed Lag Models", octobre 1995, 52 pages.
- 9548: Dufour, Jean-Marie et Jan F. Kiviet, "Exact Tests for Structural Change in First-Order Dynamic Models", octobre 1995, 39 pages.
- 9549: Kiviet, Jan F. et Jean-Marie Dufour, "Exact Tests in Single Equation Autoregressive Distributed Lag Models", octobre 1995, 38 pages.
- 9550: Briys, Eric et Pascale Viala, "Optimal Insurance Design under Background Risk", octobre 1995, 14 pages.
- 9551: Ng, Serena et Joris Pinkse, "Nonparametric Two-Step Estimation of Unknown Regression Functions when the Regressors and the Regression Error Are not Independent", octobre 1995, 22 pages.
- 9552: Bai, Jushan et Pierre Perron, "Estimating and Testing Linear Models with Multiple Structural Changes", octobre 1995, 58 pages.
- 9553: de Lemos Grandmont, Renato, "Mexico's Financial Liberalization and Reform, A Critical Overview", octobre 1995, 66 pages.
- 9554 : de Lemos Grandmont, Renato, "Multivariate Cointegration in the Presence of Structural
- Breaks: The Case of Money Demand in Mexico", octobre 1995, 69 pages.

 9555: Boyer, Marcel, Philippe Mahenc et Michel Moreaux, "Entry Blockading Locations",
- octobre 1995, 37 pages.

 9556: Boyer, Marcel et Michel Moreaux, "Capacity Commitment versus Flexibility: The
- Technological Choice Nexus in a Strategic Context", octobre 1995, 37 pages.

 9557: Boyer, Marcel et Jean-Jacques Laffont, "Environmental Protection, Producer Insolvency and Lender Liability", octobre 1995, 38 pages.
- 9558: Montmarquette, Claude, Sophie Mahseredjian et Rachel Houle, "The Determinants of University Dropouts: a Sequential Decision Model with Selectivity Bias", décembre 1995, 17 pages.
- 9559: Margolis, David N., "Cohort Effects and Returns to Seniority in France", décembre 1995, 29 pages.

- 9560: Dionne, Georges et Christian Gollier, "A Model of Comparative Statics for Changes in Stochastic Returns with Dependent Risky Assets", décembre 1995, 14 pages.
- 9601: Deaton, Angus et Serena Ng, "Parametric and Nonparametric Approaches to Price and Tax Reform", janvier 1996, 28 pages.
- 9602: Lévy-Garboua, Louis et Claude Montmarquette, "Cognition in Seemingly riskless Choices and Judgments", janvier 1996, 29 pages.
- 9603: Gonzalo, Jesus et Serena Ng, "A Systematic Framework for Analyzing the Dynamic Effects of Permanent and Transitory Shocks", mars 1996, 42 pages.
- 9604: Boyer, Marcel et Jean-Jacques Laffont, "Toward a Political Theory of Environmental Policy", avril 1996, 42 pages.
- 9605: Ai, Chunrong, Jean-Louis Arcand et François Ethier, "Moral Hazard and Marshallian Inefficiency: Evidence from Tunisia", avril 1996, 38 pages.
- 9606: Mercenier, Jean et Erinç Yeldan, "How Prescribed Policy can Mislead when Data are Defective: A Follow-up to Srinivasan (1994) Using General Equilibrium", avril 1996, 29 pages.
- 9607: Fortin, Nicole M. et Thomas Lemieux, "Rank Regressions, Wage Distributions, and the Gender GAP", avril 1996, 45 pages.
- 9608: Fortin, Nicole M. et Thomas Lemieux, "Labor Market Institutions and Gender Differences in Wage Inequality", avril 1996, 13 pages.
- 9609: S. Hosken, Daniel et David N. Margolis, "The Efficiency of Collective Bargaining in Public Schools", mai 1996, 54 pages.
- 9610: Dionne, Georges et Tahar Mounsif, "Investment Under Demand Uncertainty: the Newsboy Problem Revisited", mai 1996, 18 pages.
- 9611: Perron, Pierre et Serena Ng, "An Autoregressive Spectral Density Estimator at Frequency Zero for Nonstationarity Tests", juin 1996, 44 pages.
- 9612: Ghysels, Eric et Serena Ng, "A Semi-Parametric Factor Model for Interest Rates", juillet 1996, 29 pages.
- 9613 : Ghysels, Eric, Andrew Harvey et Eric Renault, "Stochastic Volatility", juillet 1996, 70 pages.